# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

# FORM 8-K

#### **CURRENT REPORT**

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

# Date of report (Date of earliest event reported): October 18, 2013 GLOBAL EAGLE ENTERTAINMENT INC.

(Exact Name of Registrant as Specified in Charter)

Delaware (State or Other Jurisdiction of Incorporation) 001-35176 (Commission File Number) 27-4757800 (I.R.S. Employer Identification No.)

4353 Park Terrace Drive. Westlake Village, California (Address of Principal Executive Offices) 91361 (Zip Code)

Registrant's telephone number, including area code: (818) 706 -3111

Not Applicable (Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Item 1.01. Entry into a Material Definitive Agreement.

#### Sale and Purchase Agreement

On October 18, 2013, IFES Acquisition Corp. Limited, an English company ("IFES") and an indirect wholly owned subsidiary of Global Eagle Entertainment Inc. (the "Company"), entered into Sale and Purchase Agreements (the "Purchase Agreement") with GCP Capital Partners LLP and certain individuals (each a "Seller" and collectively, the "Sellers") relating to the acquisition by IFES of all of the issued share capital (the "Purchased Shares") of Travel Entertainment Group Equity Limited, a private limited company incorporated in England and Wales ("IFE" and the acquisition, the "IFE Purchase"). IFE provides media content for use by airlines in in-flight entertainment and connectivity systems. The Company completed the IFE Purchase on October 18, 2013.

#### Consideration to the Sellers

The purchase price for the Purchased Shares was approximately \$36 million payable in cash. In connection with the IFE Purchase, the Company: (i) sold 2,453,472 shares of the Company's common stock to Putnam Equity Spectrum Fund ("Putnam"), and (ii) issued a convertible unsecured promissory note (the "Note") to PAR Investment Partners, L.P. ("PAR"), each as further described below.

#### Warranties

Under the Purchase Agreement, certain of the Sellers made customary warranties under English law and practice, including, but not limited to, those relating to: ownership and authority; no liens or encumbrances on the Purchased Shares; no litigation, arbitration, administrative or other legal proceedings; licenses and consents; insurance; confidential information; subsidiaries; financial records; books and records; indebtedness; tax matters; intellectual property; contracts; employees; compliance with laws; insurance; products liability; environmental liability; restrictions on business activities; and disclosure. The Purchase Agreement contains customary indemnification provisions under English law and practice regarding taxation liabilities. Subject to certain exceptions, the rights of IFES with respect to any claim for breach of a warranty (other than certain tax-related warranties) will survive until October 18, 2015. The rights of IFES with respect to certain tax-related warranties will survive until October 18, 2017.

#### Covenants

The Purchase Agreement contains certain customary covenants of the parties under English law and practice, including but not limited to (i) a covenant by each of the Sellers to do all such acts and procure all such things as IFES may reasonably require for transferring to IFES legal and beneficial ownership of the Purchased Shares; (ii) a non-competition covenant applicable to certain Sellers beginning on October 18, 2013 and ending on October 18, 2015; and (iii) a covenant by each of the Purchase Agreement to keep confidential certain information obtained in connection with the negotiations of the Purchase Agreement and certain ancillary agreements related to the Purchase Agreement.

The foregoing description of the Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the text of the Purchase Agreement, which is included as Exhibit 2.1 to this Current Report on Form 8-K and incorporated by reference herein. The Purchase Agreement contains warranties and covenants that the respective parties made to each other as of the date of such agreement or other specific dates. The assertions embodied in those warranties and covenants were made for purposes of the contract among the respective parties and are subject to important qualifications and limitations agreed to by the parties in connection with negotiating each of such agreement. The warranties and covenants in the Purchase Agreement are subject to a contractual standard of materiality different from that generally applicable to stockholders and were used for the purpose of allocating risk among the parties rather than establishing matters as facts.



#### Sale of Convertible Note

#### Note Purchase Agreement

On October 21, 2013, the Company entered into a Convertible Note Purchase Agreement with PAR (the "Note Purchase Agreement"). Pursuant to the Note Purchase Agreement, the Company issued the Note to PAR for a purchase price of \$19,000,000. The Note is convertible into shares of the Company's non-voting common stock. Interest will accrue on the unpaid principal amount of the Note at an annual rate of 12%. The principal amount of the Note, together with any accrued but unpaid interest, is payable in full on the earlier of (i) 60 days from issuance or (ii) the occurrence of an "Event of Default" (as defined in the Note) (the "Maturity Date"). On the Maturity Date, the Company may either (a) pay to PAR cash in the aggregate principal amount of the Note plus all accrued but unpaid interest or (b) issue to PAR non-voting shares of common stock equal to the quotient obtained by dividing (x) the principal amount of the Note to partially finance the IFE Purchase. The Company also paid a commitment fee of \$950,000 in cash to PAR upon execution of the Note Purchase Agreement. PAR currently owns (i) approximately 9.37 million shares of our common stock, which is convertible into shares of our common stock in accordance with our amended and restated certificate of incorporation and the Conversion Agreement discussed below and (iii) warrants to purchase 509,512 shares of our common stock. PAR is a party to the Registration Rights Agreement described below. The Company's Chairman of the Board, Edward L. Shapiro, is a Partner and Vice President of PAR.

The foregoing description of the Note Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the text of the Note Purchase Agreement, which is included as Exhibit 10.1 to this Current Report on Form 8-K and incorporated by reference herein.

#### **Conversion Agreement**

On October 21, 2013, the Company also entered into an Agreement with PAR (the "Conversion Agreement"). Pursuant to Article IV, Section 4.4(i) of the Company's Second Amended and Restated Certificate of Incorporation, any outstanding share of the Company's non-voting common stock shall, without the payment of any additional consideration or any other action on the part of the Company or the holder thereof, convert into one fully paid and non-assessable share of the Company's common stock, par value \$0.0001 per share, provided that the holder of such non-voting common stock delivers written notice of the holder's election to convert such non-voting common stock to common stock to the Company's transfer agent at any time on or after October 31, 2013. Under the Conversion Agreement, PAR has agreed to forbear from electing to convert non-voting common stock into common stock until the earlier of (i) May 1, 2014 or (ii) 30 days after the date on which the shares of capital stock of Advanced Inflight Alliance AG, a German stock corporation and the Company's majority-owned subsidiary, cease to be listed on the Regulated Market (General Standard) of the Frankfurt Stock Exchange (the "Outside Date"). This restriction only applies to non-voting common stock that PAR held at the time that the Conversion Agreement was entered into and to any non-voting common stock that PAR acquires after the date of the Conversion Agreement but before the Outside Date.

The foregoing description of the Conversion Agreement does not purport to be complete and is qualified in its entirety by reference to the text of the Conversion Agreement, which is included as Exhibit 10.2 to this Current Report on Form 8-K and incorporated by reference herein.

#### **Common Stock Purchase Agreement**

On October 18, 2013, the Company entered into a Common Stock Purchase Agreement (the "Stock Purchase Agreement") with Putnam relating to the sale by the Company of 2,453,472 shares (the "Putnam Shares") of the Company's common stock, par value \$0.0001 per share, at a per share purchase price of \$8.5593 for gross proceeds of approximately \$21,000,000. Under the Stock Purchase Agreement, the obligation of Putnam to purchase the Putnam Shares was subject to the fulfillment of certain conditions, including, but not limited to the following: (i) the closing of the Company's acquisition of one or more businesses must occur, or happen concurrently with, the purchase of the Putnam Shares; and (ii) the Company and affiliates of Putnam must have entered into a voting rights waiver agreement. Any of the closing conditions in the Stock Purchase Agreement may be waived by Putnam. As discussed in Item 8.01, the Company issued the Putnam Shares pursuant to the Stock Purchase Agreement on October 21, 2013. The Company used the proceeds of the sale of the Putnam Shares to partially finance the IFE Purchase.

The foregoing description of the Stock Purchase Agreement does not purport to be complete and is qualified in its entirety by reference to the text of the Stock Purchase Agreement, which is included as Exhibit 10.3 to this Current Report on Form 8-K and incorporated by reference herein.

#### Amendment to Amended and Restated Registration Rights Agreement

On October 21, 2013, the Company entered into Amendment No. 1 to the Amended and Restated Registration Rights Agreement (the "Amendment") with Global Eagle Acquisition LLC, PAR, and Putnam Capital Spectrum Fund and Putnam Equity Spectrum Fund (collectively, the "Holders"). The Amendment amends the Amended and Restated Registration Rights Agreement (the "Registration Rights Agreement"), dated January 31, 2013, previously entered into by the Company, the Holders and the other individuals party thereto. The Amendment amends the Registration Rights Agreement to, among other things: (i) include in the definition of "Registrable Securities" the shares of non-voting common stock issuable upon conversion of the Note and the shares of non-voting common stock; and (ii) in the event that the Note is converted into shares of non-voting common stock, require the Company to file, within 30 days of the Maturity Date, either (a) a registration statement pursuant to Rule 415 of the Securities Act of 1933, as amended (the "Securities Act") or (b) a post-effective amendment to the Company's existing shelf registration statement on Form S-3 (File No. 333-188121), in each case registering the resale from time to time of the shares common stock issuable upon the conversion of such shares of non-voting common stock.

The foregoing description of the Amendment does not purport to be complete and is qualified in its entirety by reference to the text of the Amendment, which is included as Exhibit 10.4 to this Current Report on Form 8-K and incorporated by reference herein.

#### Voting Rights Waiver Agreement

In connection with the above-referenced sale of 2,453,472 shares of the Company's common stock to Putnam, on October 21, 2013, the Company entered into a Voting Rights Waiver Agreement (the "Voting Rights Waiver Agreement") with the investment manager for Putnam (Putnam Investment Management, LLC ("PIM")) and certain other entities and individuals affiliated with PIM and Putnam (the "Other Putnam Investors"). Pursuant to the Voting Rights Waiver Agreement, PIM, Putnam and the Other Putnam Investors have agreed to waive all voting rights that they may have in respect of any voting securities issued by the Company that exceed, in the aggregate, 4.99% of the total voting rights exercisable by the Company's outstanding voting securities. The Voting Rights Waiver Agreement provides that any voting rights waived by Putnam, PIM or the Other Putnam Investors will be apportioned among those parties on a *pro rata* basis based upon their relative holdings of the Company's voting securities. The Voting Rights Waiver Agreement will expire at the time that Putnam and all Other Putnam Investors that are investment companies registered under the Investment Company Act of 1940, as amended, no longer own any of the Company's voting securities.

The foregoing description of the Voting Rights Waiver Agreement does not purport to be complete and is qualified in its entirety by reference to the text of the Voting Rights Waiver Agreement, which is included as Exhibit 10.5 to this Current Report on Form 8-K and incorporated by reference herein.

#### Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information set forth in Item 1.01 of this report under the heading "Sale of Convertible Note" is incorporated by reference into this Item 2.03.

#### Item 3.02. Unregistered Sales of Equity Securities.

The information set forth in Item 1.01 of this report under the heading "Sale of Convertible Note" is incorporated by reference into this Item 3.02. The Note was offered and sold to an accredited investor in a private placement transaction made in reliance upon exemptions from registration pursuant to Section 4(2) under the Securities Act and/or Rule 506 of Regulation D promulgated under the Securities Act.

#### Item 7.01. Regulation FD Disclosure.

On October 21, 2013, the Company issued a press release announcing the IFE Purchase, the sale of the Note to PAR and the registered direct offering to Putnam, each as discussed above. A copy of the press release is furnished and attached as Exhibit 99.1 hereto and is incorporated by reference into this Item 7.01.

The information furnished pursuant to this Item 7.01 and the accompanying Exhibit 99.1 shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall it be incorporated by reference into a filing under the Securities Act of 1933, as amended, or the Exchange Act, except as shall be expressly set forth by specific reference in such a filing.

#### Item 8.01. Other Events.

As described in Item 1.01, the Company completed the IFE Purchase on October 18, 2013.

On October 21, 2013, the Company sold an aggregate of 2,453,472 shares of its common stock, \$0.0001 par value per share, in a registered direct offering to Putnam at a purchase price of \$8.5593 per share for gross proceeds, before expenses, of approximately \$21,000,000. The Company sold the shares of common stock directly to Putnam and did not engage an underwriter, placement agent or advisor in connection with the Offering. Consequently, there were no fees or commissions paid to any member of the Financial Industry Regulatory Authority, Inc. or independent broker-dealer in connection with the Offering.

The Company used the proceeds from the Offering to partially finance the IFE Purchase.

The offer and sale of shares of the Company's common stock pursuant to the Offering was registered under the Company's existing shelf registration statement on Form S-3 (Registration No. 333-188121) (the "Registration Statement"), which the Securities and Exchange Commission declared effective on August 22, 2013.

A copy of the legal opinion of McDermott Will & Emery LLP, relating to the validity of the shares issued to Putnam under the Stock Purchase Agreement and pursuant to the Registration Statement, is filed as Exhibit 5.1 to this report and is filed with reference to, and is hereby incorporated by reference into, the Registration Statement

#### Item 9.01. Financial Statements and Exhibits.

#### (d) Exhibits.

- 2.1 Sale and Purchase Agreement by and among IFES and the Sellers, dated October 18, 2013.
- 4.1 Convertible Note (included as exhibit to Note Purchase Agreement filed as Exhibit 10.1).
- 5.1 Opinion of McDermott Will & Emery LLP.
- 10.1 Convertible Note Purchase Agreement between the Company and PAR, dated October 21, 2013.
- 10.2 Agreement between the Company and PAR, dated October 21, 2013.
- 10.3 Common Stock Purchase Agreement between the Company and Putnam, dated October 18, 2013.
- 10.4 Amendment No.1 to the Amended and Restated Registration Rights Agreement among the Company and certain holders party thereto, dated October 21, 2013.
- 10.5 Voting Rights Waiver Agreement between the Company and PIM, dated October 21, 2013.
- 23.1 Consent of McDermott Will & Emery LLP (included in its opinion filed as Exhibit 5.1).
- 99.1 Press Release issued October 21, 2013.

# SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

### GLOBAL EAGLE ENTERTAINMENT INC.

By: /s/ Michael Pigott

Name: Michael Pigott Title: VP Legal

Dated: October 21, 2013

#### EXHIBIT INDEX

- 2.1 Sale and Purchase Agreement by and among IFES and the Sellers, dated October 18, 2013.
- 4.1 Convertible Unsecured Promissory Note (included as exhibit to Note Purchase Agreement filed as Exhibit 10.1).
- 5.1 Opinion of McDermott Will & Emery LLP.
- 10.1 Convertible Note Purchase Agreement between the Company and PAR, dated October 21, 2013.
- 10.2 Agreement between the Company and PAR, dated October 21, 2013.
- 10.3 Common Stock Purchase Agreement between the Company and Putnam, dated October 18, 2013.
- 10.4 Amendment No.1 to the Amended and Restated Registration Rights Agreement among the Company and certain holders party thereto, dated October 21, 2013.
- 10.5 Voting Rights Waiver Agreement between the Company and PIM, dated October 21, 2013.
- 23.1 Consent of McDermott Will & Emery LLP (included in its opinion filed as Exhibit 5.1).
- 99.1 Press Release issued October 21, 2013.

Exhibit 2.1

2013

# THE SELLERS

#### AND

# IFES ACQUISITION CORP LIMITED

## SALE AND PURCHASE AGREEMENT relating to the entire issued share capital of TRAVEL ENTERTAINMENT GROUP EQUITY LIMITED

1

#### 5 New Street Square, London EC4A 3TW Tel +44 (0)20 7300 7000 Fax +44 (0)20 7300 7100 DX 41 London www.taylorwessing.com

# **TaylorWessing**

Index

Clause No.		
		Page No.
1.	Definitions and interpretation	4
2.	Sale and Purchase	11
3.	Consideration	12
4.	Locked Box	12
5.	Warranties	13
6.	Limitations	14
7.	Buyer Warranties	18
8.	Completion	19
9.	Post-Completion	20
10.	Non-competition	21
11.	Confidentiality	22
12.	Notices and other Communications	24
13.	Third party rights	25
14.	Assignment	25
15.	Entire agreement	25
16.	Miscellaneous	25
17.	Governing Law and Jurisdiction	27
SCHEDULE 1		29
The Sellers and the Consideration		29
SCHEDULE 2		30
Details of the Company and the Subsidiary Undertakings		30
Part B		32
Subsidiary Undertakings		32
Direct Subsidiary		32
SCHEDULE 3		45
Intellectual property		45
SCHEDULE 4		46
Warranties		46
SCHEDULE 5		63
Tax Covenant		
SCHEDULE 6		
Permitted Leakage		
SCHEDULE 7		
Property		73



Agreed Form Documents:

- DVD of the Data Room Deed of Termination Resignation Letters
- 1. 2. 3.
- 5. 4. 5. 6. Short Form Sale and Purchase Agreements
- Disclosure Letter
- Employment Letter
- 7. 8.
- Announcement Permitted Leakage schedule

# THIS AGREEMENT is made on 18th October 2013

#### BETWEEN

- (1) THE PERSONS whose names and addresses are set out in schedule 1 (the "Sellers"); and
- (2) **IFES ACQUISITION CORP LIMITED** (company number 08732410) whose registered office is 15 Stukeley Street, London, WC2B 5LT (the "**Buyer**").

# INTRODUCTION

The Sellers have agreed to sell (for the consideration set opposite their names in schedule 1) and the Buyer has (in reliance upon the warranties made in this agreement) agreed to buy all of the issued share capital of Travel Entertainment Group Equity Limited which is a private company limited by shares incorporated in England and Wales, further details of which are set out in part A of schedule 2.

#### AGREED TERMS

#### 1. Definitions and interpretation

1.1 Definitions

In this agreement:

"accounting period" means an accounting period determined in accordance with Chapter 2 of Part 2 of CTA 2009;

"Accounting Standards" means in relation to the Accounts, the accounting requirements of the Companies Act and all applicable reporting standards and UK GAAP, in each case as at the date of the relevant accounts;

"Accounts" means the audited (and, where relevant, consolidated) accounts of each Group Company for the financial year ended on the Accounts Date including the auditors' and directors' reports, the audited balance sheets as at the Accounts Date, the audited profit and loss accounts for that period and the notes to them, copies of which have been supplied to the Buyer and are attached to the Disclosure Letter;

"Accounts Date" means 31 December 2012;

"Advisers" in relation to a person means professional advisers advising that person, including (unless the context requires otherwise) partners or members in or directors of (as the case may be) such advisers and employees of such advisers;

"Agreed Percentage" means, in relation to each of the Warrantors, the percentage set out in the table contained in clause 6.6 against that Warrantor's name;

"Announcement" means the announcements in the agreed form;

"**Applicable Law**" means any law (including common law or other binding law), statute, regulation, code, ordinance, rule, judgment, order, decree or directive or any determination by or requirement of a Competent Authority or interpretation or administration of any of the foregoing by a Competent Authority;

"Associate" means in relation to a person:

- (a) a person who is his associate, determined in accordance with section 435 of the Insolvency Act 1986; and (whether or not an associate as so defined)
- (b) any Group Undertaking (as defined in section 1161 of the Companies Act) of that person;

"Associated Person" means, in relation to a company, a person (including an employee, agent or Subsidiary Undertaking) who performs or has performed services for or on behalf of that company;

"Associated Undertaking" means a company or any other body corporate (whether or not registered in the United Kingdom) in which any of the Group Companies holds shares conferring the right to exercise 20 per cent or more of the votes which could be cast on a poll at a general meeting of such company and which is not a subsidiary of any of the Group Companies, details of which are set out in part B of schedule 2;

"Business Day" means a day other than a Saturday, Sunday or public holiday in England;

"Buyer's Solicitors" means SJ Berwin LLP of 10 Queen Street Place, London EC4R 1BE (Ref: 423/G28137.3);

"Claim" includes a claim, action, proceeding or demand;

"Companies Act" means the Companies Act 2006;

"Company" means Travel Entertainment Group Equity Limited, details of which are set out in part A of schedule 2;

"**Competent Authority**" means any national, state or local governmental authority, any governmental, quasi-governmental, judicial, public or administrative agency, authority or body, any court of competent jurisdiction, any operator or regulator of a Recognised Investment Exchange, the Panel on Takeovers and Mergers, and any local, national or supranational agency, inspectorate, minister, ministry, official or public or statutory person (whether autonomous or not) acting within their powers and having jurisdiction over this agreement or any of the parties;

"Completion" means completion of the sale and purchase of the Sale Shares in accordance with clause 8;

"Completion Date" means the date this agreement is entered into;

"Computer Data" means the computer-readable information or data owned or used by any Group Company and stored in electronic form;

"**Computer Hardware**" means the computer hardware, firmware, equipment and ancillary equipment (other than Computer Software and Computer Data) owned or used by any Group Company and all related manuals and documentation;

"Computer Software" means the computer programs owned or used by any Group Company and all related manuals and documentation;

"Computer System" means the Computer Hardware, Computer Data and Computer Software;

"Confidential Information" has the meaning given to it in clause 11;

"**Consultant**" means a consultant, independent contractor or other individual engaged by a Group Company at the date of this agreement under a contract for service not being an Employee;

"Connected" has, in relation to a person, the meaning given in Section 1122 CTA 2010;

"CTA 2009" means the Corporation Tax Act 2009;

"CTA 2010" means the Corporation Tax Act 2010;

"**Customer**" means a customer, client or prospective customer or client of a Group Company with whom any Group Company has either dealt in connection with the Prohibited Business or with whom any Group Company has been negotiating in respect of the supply of any goods or services by a Group Company at any time during the 24 month period ending on the Completion Date;

"D Minority Shareholders" means Howard Clark, Simon Howard, Rouba Korfali, Adrian Lambert, Flavia Nogueira, Zoe Tierney, Michael White, Pamela McEwan, Michelle Gosling, Alexandra White, Sarah Howard, Matthew Bunker and Alison Clark;

"D Minority Shares" means the D ordinary shares in the Company held by the D Minority Shareholders;

"**Data Protection Legislation**" means any data protection and privacy legislation applicable to a Group Company's business including, in the United Kingdom, the Data Protection Act 1998 and the Privacy and Electronic Communications (EC Directive) Regulations 2003;

"**Data Room**" means the online data room made available by the Sellers to the Buyer, the contents of which is set out in the agreed form DVD of the Data Room;

"Data Room Documents" means the documents in the Data Room;

"**Deed of Termination**" means the deed of termination in the agreed form terminating the shareholders' agreement dated 12 July 2011 which was made between, the Company, GCP Capital Partners LLP, the Executives (as defined therein) and the Greenhill Funds (as defined therein);

"Director" means a director of a Group Company;

"**Disclosed**" means fairly disclosed to the Buyer in the Disclosure Letter in such a manner and detail so as to enable the Buyer to make a reasonably informed assessment of the fact, matter or information;

"**Disclosure Letter**" means the letter of the same date as this agreement from the Warrantors to the Buyer executed and delivered immediately before the signing of this agreement;

"Employee" means an individual who has entered into or works under a current contract of employment with a Group Company;

"Employment Letter" means the employment letter for Andrew McEwan in the agreed form;

"Encumbrance" includes a mortgage, charge, lien, pledge, right of pre-emption, option, covenant, restriction, lease, trust, order, decree, title defect or any other security interest or conflicting claim of ownership or right to use or any other third party right;

"financial year" shall be construed in accordance with section 390 of the Companies Act;

"Group Companies" means the Company and the Subsidiary Undertakings and Associated Undertakings and "Group Company" means any of them and "Group" means the Group Companies together;

"Group Intellectual Property" means all Intellectual Property owned or used by any Group Company;

"Group Member" means at any relevant time, in relation to any undertaking, a "group undertaking" (as defined in section 1161 of the Companies Act) of that undertaking and "Member of its Group", in relation to any undertaking, means any group undertaking as so defined of that undertaking;

"IHTA 1984" means the Inheritance Tax Act 1984;

"Institutional Sellers" means Greenhill Capital Partners Europe LP, Greenhill Capital Partners Europe (Employees) LP and Parallel Private Equity Greenhill LP, in each case acting through GCP Capital Partners LLP in its capacity as their manager;

"Insurance Policy" means the warranty and indemnity insurance policy in agreed form to be acquired by the Buyer on Completion;

"Intellectual Property" means all patents, utility models, trade and service marks, logos, design rights, trade names, business names, domain names, copyright rights, rights in the nature of copyright, moral rights, database rights, know-how, rights in trade secrets and confidential information, rights protecting reputation and goodwill, rights in unfair competition and all other intellectual property rights and analogous rights as may exist anywhere in the world for the full term of the rights concerned together with all reversions, revivals, extensions and renewals of such rights (whether registered or not); all registrations and pending registrations relating to any such rights; and all rights of action, powers or benefits belonging to or accrued in relation to such rights (including the right to sue for and recover damages for past infringements);

"Licence Agreements" means the agreement set out in documents 21.1.7.1, 21.1.7.2 and 21.1.7.3 of the Data Room (licence agreements with film studios);

"Locked Box Date" means 30 September 2013;

"Loss" means any loss, damage, liability, fine, penalty, charge and any other cost and expense including any reasonable costs of recovery;

"Management Accounts" means the management accounts of the Group Companies comprising the balance sheet as at the Locked Box Date and the profit and loss account for the months commencing on the day immediately following the Accounts Date and ending on the Locked Box Date, copies of which are attached to the Disclosure Letter;

"Managers' Representative" means Steven Gosling being one of the Sellers or any person appointed by Steven Gosling or his successors;

"Permitted Leakage" means a payment set out in schedule 6 of this agreement and "Permitted Leakages" means any or all of those payments;

"**Prohibited Area**" means anywhere in the world in which the Group Companies are currently, or have been in the 12 months prior to the Completion Date, carrying on business to a material extent;

"**Prohibited Business**" means the businesses of the provision of "in-flight" entertainment services to airlines and international cruise lines carried on by the Company at the Completion Date and including, inter alia, doing any of the following in relation to such businesses:

- (a) the design and recommendation of programming schedules;
- (b) the procurement of film television, audio and other media content from film studios, televisions studios and other distributors;
- (c) the in-house production of original audio and visual media content, including information, safety and destination films;
- (d) all ancillary services relating to the above including:
  - (i) dubbing media content in relation to local language requirements;
  - (ii) data media copying, encoding and reformatting;
  - (iii) graphic user interface design;
  - (iv) the development and application of integration software; and
  - (v) publishing of hard copy material,

and any other business conducted by any Group Company in the 12 calendar month period immediately prior to the Completion Date;

"Property" means all the properties short particulars of which are set out in schedule 7;

"**Recognised Investment Exchange**" has the meaning given in section 285 of the Financial Services and Markets Act 2000 (such exchanges being at the date of this agreement, London Stock Exchange Plc (including, without limitation, in its capacity as operator and regulator of AIM and Turquoise Derivatives);

"Sale Shares" means the issued share capital of the Company referred to in schedule 1;

"Sellers' Solicitors" means Taylor Wessing of 5 New Street Square, London EC4A 3TW;

"Sellers' Solicitors Account" means National Westminster Bank Plc, Fleet Street, PO Box 281, 156 Fleet Street, London, EC4A 2DX, sort code 60-80-08, account number 03666328;

"Short Form Sale and Purchase Agreements" means the sale and purchase agreements in the agreed form to be entered into by the D Minority Shareholders and the Buyer on the date of this agreement;



"Subsidiary Undertakings" means the subsidiary undertakings (as defined in section 1162 of the Companies Act) of the Company details of which are set out in part B of schedule 2;

"**Supplier**" means a supplier to (other than utilities in respect of the supply of services in the ordinary and normal course of their business to their general body of customers), or sub-contractor of any Group Company in connection with the Prohibited Business with whom any Group Company has traded during the 24 months immediately ending on the Completion Date;

"Tax" or "Taxation" has the meaning given to it in schedule 5;

"Tax Authority" has the meaning given to it in schedule 5;

"Tax Claim" means a claim for breach of a Tax Warranty or a Claim under the Tax Covenant;

"Tax Covenant" means the covenant set out in schedule 5;

"Taxes Act" means the Income and Corporation Taxes Act 1988;

"Tax Warranties" means the warranties set out in part 2 of schedule 5;

"Transaction Document" means any of this agreement and any other document entered into on or about the date of this agreement in connection with it;

"UK GAAP" means accounting principles, standards and practices generally accepted from time to time in the United Kingdom and approved by the United Kingdom Accounting Standards Board;

"VAT" means Value Added Tax chargeable under the VATA 1994 or under any legislation replacing it or under any legislation which the VATA replaced and further means Value Added Tax at the rate in force when the relevant supply is made and any tax of a similar nature which is introduced in substitution for or as an addition to such tax from time to time and any penalties or fines in relation to them and shall include any similar tax or sales tax in any jurisdiction outside the UK;

"VATA 1994" means the Value Added Tax Act 1994;

"Warranties" means the warranties set out in part 1 of schedule 4 and the Tax Warranties; and

"Warrantors" means Steven Gosling, Andrew McEwan and Estibaliz Asiain Sancho.

#### 1.2 Interpretation

In this agreement:

- (a) reference to:
  - (i) any statute or statutory provision includes a reference:
    - (A) to that statute or statutory provision as from time to time consolidated, modified, re-enacted (with or without modification) or replaced by any statute or statutory provision; and
    - (B) any subordinate legislation made under the relevant statutory provision,



except to the extent that the effect of referring to any such consolidation, modification or re-enactment coming into force after the date of this agreement would be to increase or extend the liability of a party under this agreement;

- (ii) the singular includes the plural and vice versa and any gender includes other genders;
- (iii) the "introduction" or to a "clause" or "schedule" is a reference to the Introduction or the relevant clause or schedule of or to this agreement;
- (iv) a person includes all forms of legal entity including an individual, company, body corporate (wherever incorporated or carrying on business), unincorporated association, governmental entity and a partnership and, in relation to a party who is an individual, his legal personal representative(s);
- (v) a document or DVD "in agreed form" is to a document or DVD in the form agreed by and initialled by or on behalf of each party for the purposes of identification or executed simultaneously with this agreement;
- (vi) a party or the parties means a party or the parties to this agreement and includes his successors and permitted assigns and for this purpose "**permitted assigns**" includes:
  - in relation to a right of a party any person to whom that right may have been assigned except to the extent that the assignment of that right would be in breach of the provisions of this or any other agreement or deed or prohibited by law; and
  - in relation to an obligation of a party any person to whom that obligation may have been assigned with the written agreement of the party to whom the obligation is owed,

provided that notwithstanding any succession, assignment or transfer, no party shall be relieved from any obligation arising under this agreement except:

- by operation of law;
- as expressly provided in this agreement; or
- with the written agreement of the party to whom the obligation is owed;
- (vii) "writing" and "written" includes typing, printing, lithography, photography and other modes of representing or reproducing words in a legible and non transitory form but excludes electronic mail; and
- (viii) "this agreement" includes this agreement as amended or supplemented from time to time;
- (b) the words "**include**", "**including**" and "**in particular**" are to be construed as being by way of illustration or emphasis only and are not to be construed so as to limit the generality of any words preceding them;

- (c) the words "other" and "otherwise" are not to be construed as being limited by any words preceding them;
- (d) the word "**property**" includes choses in action and other intangible property;
- (e) the table of contents and the headings to clauses and schedules are to be ignored in construing this agreement;
- (f) the words, "**parent undertaking**", "**subsidiary undertaking**" and "**undertaking**" have the meanings given to them in sections 1162 and 1161 of the Companies Act;
- (g) if a period of time is specified and dates from a given day or the day of an act or event, it shall (unless otherwise stated in clause 12) be calculated excluding that day and a reference to a time of day is unless otherwise specifically stated a reference to the time in England;
- (h) if a party must do something on a given day (other than service of a communication in accordance with clause 12), they must do it by 5:30 p.m. (in the United Kingdom) on that day (unless this agreement expressly states otherwise). If they do the thing after 5:30 p.m. on a day, they are treated as not having done it until the next Business Day; and
- (i) a reference to any:
  - (i) English legal term for any action, remedy, procedure, judicial proceeding, legal document, legal status, or legal concept; and/or
  - (ii) English legislation,

is in respect of any jurisdiction other than England and Wales, deemed to include what most nearly approximates in that jurisdiction to the English legal term or legislation, as applicable.

1.3 The schedules form part of this agreement as if set out in full in this agreement and a reference to "this agreement" includes a reference to the schedules.

#### 2. Sale and Purchase

- 2.1 Each Seller shall sell with full title guarantee and free from all Encumbrances the number of Sale Shares set out opposite their names in schedule 1 and the Buyer shall, subject to clause 2.3, buy the Sale Shares in each case on the terms of this agreement.
- 2.2 The Sale Shares shall be sold with all rights to dividends and other distributions declared or paid after the date of this agreement in respect of the Sale Shares and all other rights and advantages belonging to or accruing on the Sale Shares on or after that date.
- 2.3 If any Seller or D Minority Shareholder fails to comply with his obligation (whether under this agreement or under a Short Form Sale and Purchase Agreement) to transfer his Sale Shares or D Minority Shares on Completion, the Buyer shall not be obliged to complete the purchase of the other Sale Shares.
- 2.4 Each Seller irrevocably waives all pre-emption rights which he may have under the Company's articles of association or any other agreement relating to the Sale Shares or otherwise so as to enable the sale of the Sale Shares to the Buyer to proceed free of pre-emption rights.

#### 3. Consideration

- 3.1 The total consideration for the sale of the Sale Shares and the D Minority Shares is £22,363,000, payable in cash at Completion in accordance with the terms of this agreement and the Short Form Sale and Purchase Agreements (the "**Consideration**").
- 3.2 The portion of the Consideration to which each Seller is entitled is the amount set out opposite his name in schedule 1.
- 3.3 The Consideration in respect of the amounts due to the Sellers shall be satisfied in cash on Completion by transfer to the Sellers' Solicitors Account.

#### 4. Locked Box

- 4.1 Each Seller undertakes (in respect of payments made to him or benefit conferred on him and persons Connected with him only and not in respect of any payments made to or benefit conferred on any other Seller or persons Connected with them) to the Buyer that from (and excluding) the Locked Box Date to (and including) the Completion Date, other than any Permitted Leakage:
  - (a) there has been no payment of any management, service or other fees or compensation or transaction bonuses from any Group Company to that Seller or to persons Connected with him;
  - (b) no share or loan capital of any Group Company (or any interest thereon) has been issued to that Seller or to any person Connected with that Seller or redeemed, purchased or repaid by such Group Company for the benefit of that Seller or for the benefit of any person Connected with that Seller;
  - (c) the only payments received by that Seller or by a person Connected with that Seller from any Group Company have been payments in respect of Permitted Leakages;
  - (d) no dividend or distribution of profits, capital or assets, or any bonus or other payment of any nature has been paid or declared or made by any Group Company to or in favour of that Seller;
  - (e) no Group Company has waived, deferred or released or agreed to waive, defer or release (whether conditional or not) any amount. liability or obligation owed to that Group Company by that Seller or by a person Connected with that Seller;
  - (f) no liabilities have been assumed or incurred (or any indemnity or guarantee given in respect thereof) by any Group Company in favour of or on behalf of that Seller or by a person Connected with that Seller;
  - (g) no Encumbrance has been created over the assets of any Group Company in favour of or on behalf of that Seller or by a person Connected with that Seller;
  - (h) no costs or expenses of that Seller or any person Connected with that Seller which relate either to the sale of the Sale Shares (to any person) or to the other transactions contemplated by this agreement (and any document required to be executed or delivered at Completion) have been paid or incurred (whether by way of repayment of any debt or otherwise) or have been agreed to be paid or incurred, by any Group Company;

- (i) no assets have been transferred to or liabilities assumed, indemnified or incurred by any Group Member in favour or for the benefit of that Seller or any person Connected with that Seller; and
- (j) the relevant Seller has made or entered into any agreement or arrangement relating to any matters referred to in this clause 4.1.
- 4.2 Each Seller undertakes (in respect of himself and persons Connected with him only and not in respect of any other Seller or persons Connected with them) to the Buyer to notify the Buyer in writing promptly after becoming aware of any receipt by that Seller or by a person Connected with that Seller of any payments or benefits conferred or circumstances which could constitute a breach by him of the undertakings in clause 4.1
- 4.3 In the event of any breach of clause 4.1, the relevant Seller shall, on demand by the Buyer at any time between Completion and the date falling 9 months following Completion, pay to the Buyer (the Buyer acting for itself and as agent and trustee for each Group Company) an amount equal (on a pound for pound basis) to the payment, benefit or other item received by the Seller or by the person Connected with that Seller in breach of clause 4.1. Any such payments shall, to the extent possible, be deemed to give rise to a corresponding reduction in the consideration received by that Seller for the Sale Shares sold by that Seller. The Buyer shall have no remedy for a breach of any of the undertakings set out in clause 4.1 save for this clause 4.3 including without limitation in respect of any demand served after a 9 month period following Completion (save in the event of fraud, wilful misstatement, wilful non-disclosure or dishonesty by the relevant Seller or any person Connected with him).
- 4.4 The liability of each Seller under this clause 4 shall not be limited or qualified in any respect by the provisions of clause 6.
- 4.5 Notwithstanding any provision to the contrary contained in this agreement, the Sellers shall not be deemed to be Connected with any Group Companies or any other Seller (with the exception, in the case of the Warrantors, of their spouses) for the purpose of the Locked Box arrangements set out in this clause 4.

## 5. Warranties

- 5.1 Each Seller, in respect of himself only, warrants to the Buyer that he is entitled to sell and transfer the full legal and beneficial ownership in the Sale Shares stated to be held by him in schedule 1 on the terms of this agreement.
- 5.2 Each Seller, in respect of himself only, warrants on a several basis to the Buyer that there is no Encumbrance on, over or affecting the Sale Shares held by him in schedule 1 and there is no agreement or commitment to give or create any Encumbrance and no claim has been made by any person to be entitled to an Encumbrance in relation thereto.
- 5.3 Each Seller, in respect of himself only, warrants to the Buyer that he has full power and authority to enter into and perform this agreement and that this agreement and each of the documents in the agreed form to be executed by him constitute valid and binding obligations on that Seller which are enforceable in accordance with their terms. Each Institutional Seller warrants to the Buyer in respect of itself only that it has taken all corporate and other action necessary to enable it to enter into and perform this agreement and each of the documents in the agreed form to be executed by it and has obtained all approvals and consents required by it for the performance by it of the transactions contemplated by this agreement, provided always that no warranty is given in this clause 5.3 in respect of the need for any competition or anti-trust approval, filing, clearance or consent.

- 5.4 Each Warrantor warrants to the Buyer in the terms of the Warranties which are given by the Warrantors on the date of this agreement to the Buyer on a joint and several basis subject to the limitations contained in this agreement.
- 5.5 Each Warranty shall be construed as an independent warranty and (except as otherwise provided, including the terms of the other provisions of this clause 5 and clause 6) shall not be limited by reference to or inference from any other term of any Transaction Document or any other Warranty.
- 5.6 Any payments made by a Warrantor to the Buyer in connection with this agreement shall, to the extent possible, be treated by the Warrantor and the Buyer as a reduction in the Consideration for the Sale Shares sold by that Warrantor.
- 5.7 Where a Warranty is qualified by the expression "so far as the Warrantors are aware" or a similar expression, each Warrantor shall be deemed to have such awareness which he would have had had he made all reasonable enquiries of each other Warrantor, or Seller, and the D Minority Shareholders, at the date of this agreement.
- 5.8 Each Warrantor undertakes to the Buyer and to each Group Company that he will waive any right which he may have and not make any Claim in respect of any misrepresentation, inaccuracy or omission in or from any information or advice supplied by a Group Company or its officers, Employees or Consultants in connection with the entering into of this agreement, the giving of the Warranties and the preparation of the Disclosure Letter, save that this clause 5.8 will not preclude any Warrantor from claiming against any other Warrantor under any right of contribution (whether statutory, contractual or otherwise) or indemnity to which such Warrantor may be entitled.
- 5.9 The rights and remedies of the Buyer in respect of any breach of this agreement shall not be affected by Completion, by any investigation made by or on behalf of the Buyer into the affairs of the Group Companies or by any other event or matter whatsoever which otherwise might have affected such rights and remedies except a specific and duly authorised written waiver or release. Subject to clause 6.19, no information relating to the Group Companies of which the Buyer or its Advisers has knowledge (actual, imputed or constructive) other than by reason of its being disclosed in accordance with clause 6.9 shall prejudice any Claim which the Buyer shall be entitled to bring or shall operate to reduce any amount recoverable by the Buyer under this agreement.
- 5.10 Only the Tax Warranties and the Warranties at paragraphs 3 and 4 of schedule 4 are given in respect of Taxation Affairs of the Group Companies.

#### 6. Limitations

Limitations not to apply

- 6.1 The limitations set out in this clause 6 shall not apply:
  - (a) to a Claim for breach of Warranty or to a Tax Claim against the Warrantors which is the consequence of fraud, wilful misstatement, wilful non-disclosure or dishonesty on the part of a Warrantor; or
  - (b) to a Claim for breach of the Warranties set out in paragraphs 2.1 and 2.3 to 2.7 inclusive of schedule 4.

#### Time limits

- 6.2 Subject to clause 6.11, the rights of the Buyer in respect of any Claim for breach of a Warranty (other than a Tax Warranty) shall only be enforceable if the Buyer gives written notice to the Warrantors on or before the second anniversary of Completion.
- 6.3 Subject to clause 6.11, the rights of the Buyer in respect of any Tax Claim shall only be enforceable if the Buyer gives written notice to the Warrantors on or before the fourth anniversary of Completion.

#### Threshold

- 6.4 The Buyer shall not be entitled to recover in respect of a Claim for breach of Warranty unless in respect of any such individual Claim (or series of related such Claims arising from the same subject matter) the amount of the liability exceeds £30,000.
- 6.5 The Warrantors shall not be liable in respect of any Claim for a breach of the Warranties unless the total cumulative liability of the Warrantors in respect of all such Claims exceeds £200,000 (in which event the Warrantors shall be liable for the whole of such liability and not merely for the excess).

#### Maximum claims

6.6 The maximum aggregate individual liability of each of the Warrantors in respect of Claims for breaches of the Warranties set out at paragraph 1 and paragraphs 3 to 29 of Part 1 schedule 4 and Tax Claims for which they are liable shall not exceed either the amount set out opposite that Warrantor's name below or the amount equivalent to the Agreed Percentage of any such Claim as set out below:

Amount (£)	Agreed Percentage
325,000	50
162,500	25
162,500	25
	325,000 162,500

Double claims

6.7 The Buyer shall not be entitled to recover from the Warrantors under the Warranties and the Tax Covenant more than once in respect of the same damage suffered, and accordingly the Warrantors shall not be liable in respect of any breach of the Warranties to the extent that the loss attributable to that breach is or has been included in a Claim under the Tax Covenant to the extent that it has been satisfied, nor shall the Warrantors be liable in respect of a Claim under the Tax Covenant to the extent that the loss is or has been included in a Claim for breach of the Warranties to the extent that it has been satisfied.

#### Contingent liabilities

6.8 The time limits in clauses 6.2 and 6.3 shall not limit any Claim for any breach of the Warranties in respect of a liability which is contingent or unascertained where written notice of the Claim is given to the Warrantors as soon as reasonably practicable after the Buyer becomes aware that the relevant matter gives rise to a right to make a Claim (giving so far as practicable a reasonable estimate of the quantum and details of the Claim) before the expiry of the relevant periods specified in those clauses.

Disclosure Letter

6.9 The Warrantors shall be under no liability for a Claim for breach of the Warranties in respect of any matter to the extent that the matter or circumstance giving rise to such liability was Disclosed.

Commencement of proceedings

6.10 Subject to clause 6.12, the Warrantors shall not be liable in respect of any Claim for breach of the Warranties or for Tax Warranties (if not previously satisfied, settled or withdrawn) unless legal proceedings have been validly issued and served on the Warrantors on or before the date falling 6 months after the date on which notice of that claim was served under clause 6.2 or 6.3 or (in the case of a liability referred to in clause 6.8) 6 months after the date on which the liability becomes actual or (as the case may be) capable of being quantified.

#### Recovery from third party

- 6.11 If the Buyer or any other Group Company recovers any sum from any person (including, without limitation, its insurers) in respect of the same subject matter of any Claim under the Warranties (other than Tax Warranties in respect of which the provisions of schedule 5 shall apply) in respect of which the Warrantors have made any payment to the Buyer, the Buyer shall pay to the Warrantors a sum equal to the lesser of:
  - (a) any amount recovered by the Buyer (net of all reasonable costs and expenses incurred by it in pursuing the claim against the other person and also less any Tax paid or payable as a result of such recovery (or which would have been paid or payable but for the availability of any Buyer's Relief (as defined in schedule 5))); and
  - (b) the amount paid by the Warrantors in respect of such claim less any part of such amount previously repaid to the Warrantors under any provision of this agreement.

in each case, reduced (where applicable) by an amount equal to the actual loss less the aggregate net amount recovered by the Buyer.

6.12 Where the Buyer and/or the Company is/are at any time entitled to recover from some other person any sum in respect of the same loss which would otherwise have been the subject of a Claim under the Warranties (other than the Tax Warranties, in respect of which the provisions of schedule 5 shall apply), the Buyer shall promptly notify the Warrantors, provide such information as the Warrantors may reasonably require and shall, if so requested by the Warrantors and: (i) without prejudice to its ability to enforce such Claim against the Warrantors; and (ii) subject to the Buyer and the Company being first indemnified against all contemplated liabilities, damages, costs, charges, out of pocket costs and expenses reasonably incurred or to be incurred by the Buyer and/or the Company, procure that the Company shall take all reasonable steps, having reasonable regard to the goodwill of the Buyer's Group (including the Group Companies) to enforce such recovery and in the event that the Buyer or the Company shall recover any amount from such other person, the amount of the claim against the Warrantors shall be reduced by the amount recovered.

#### Exclusions on liability

6.13 No Claim shall be made by the Buyer against the Warrantors under the Warranties (other than the Tax Warranties, in respect of which the provisions of schedule 5 shall apply):

- (a) in respect of any liability, matter or thing which would not have arisen but for a voluntary act or transaction by, or at the request of, or with the express written approval of the Buyer after Completion otherwise than as required by law or applicable regulation or other than in the ordinary course of conduct of the business of the Group as carried on prior to Completion or pursuant to a binding obligation of any Group Company in existence at Completion and where the Buyer knew or ought reasonably to have known that the same would give rise to a Claim under the Warranties;
- (b) in respect of any liability, matter or thing to the extent it occurs as a result of any legislation not in force at the date hereof or any change of law having retrospective effect coming into force after the date hereof;
- (c) in respect of a contingent liability unless and until such liability becomes actual and is due and payable (save that this shall not avoid any claim notified with respect to such contingent liability within the time limit referred to in clauses 6.2 or 6.3);
- (d) which arises as a result of any act, event, default, matter or circumstance which occurred or arose prior to 1 January 2010 and of which the Warrantors were not aware at any time prior to the date of this agreement;
- (e) if the facts or matter giving rise to such a Claim result from a change in the accounting or taxation policies or practices of the Buyer or the Company introduced or having effect after Completion (other than to the extent necessary to comply with law or UK GAAP applying or in force on or prior to the Completion Date); or
- (f) where the loss or liability resulting from such Claim has been or is made good or otherwise compensated for at no expense to the Buyer or the Company; or
- (g) to the extent that the subject matter of the claim is a matter specifically provided for or reserved for in the Accounts or Management Accounts.

#### Duty to mitigate

6.14 The Buyer acknowledges to the Warrantors notwithstanding any provision to the contrary contained in this agreement its common law duty to mitigate its loss in respect of any breach of the Warranties in accordance with normal English law principles.

#### Notify Warrantors

6.15 If the Buyer or any Group Company becomes actually aware of any matter which gives or might reasonably give rise to a successful Claim under the Warranties (other than Tax Warranties in respect of which the provisions of schedule 5 shall apply), the Buyer shall, on a "without prejudice" basis, give written notice to the Warrantors as soon as reasonably practicable after the date on which it becomes aware of that matter, specifying the matter in reasonable detail (according to the knowledge of the Buyer in relation to the claim concerned at such time), the Warranties which have or which are likely to have been breached and a reasonable estimate of the amount of the Claim under the Warranties, provided that such notice shall not be a condition precedent to the liability of the Warrantors in respect of the Warranties save that the Warrantors shall not be liable for any increase in loss or damage attributable to the Buyer's or the Group Companies failure to give or delay in giving such notice.

#### Conduct of third party claims

- 6.16 If the Buyer or any Group Company becomes aware of any Claim by a third party which might reasonably be expected to give rise to a successful Claim under the Warranties (other than Tax Warranties in respect of which the provisions of schedule 5 shall apply) subject to clause 6.17 and the remainder of the provisions of this clause, the Buyer shall (and shall procure that the relevant Group Company concerned shall):
  - (a) take such commercial action and give such information and assistance, having reasonable regard to the goodwill of the Buyer's Group (including the Group Companies), as the Managers' Representative may from time to time reasonably request to dispute, resist, avoid, compromise or defend the matter in question;
  - (b) consult insofar as reasonably possible with the Warrantors with regard to any developments relating to the matter in question giving rise to a Claim under the Warranties and provide the Warrantors (at their cost) with copies of all relevant correspondence and documents in relation to that matter; and
  - (c) not admit liability in respect of or settle or compromise the matter in question giving rise to a Claim under the Warranties without the prior written consent of the Managers' Representative, such consent not to be unreasonably withheld or delayed.
- 6.17 The Warrantors shall indemnify the Buyer and any relevant Group Company to their reasonable satisfaction against all losses, damages and expenses incurred by them in complying with their obligations under clause 6.16

Rescission

- 6.18 The parties irrevocably and unconditionally waives any right they may have to rescind this agreement for any misrepresentation (save for any misrepresentation which constitutes fraud) contained in this agreement.
- 6.19 The Buyer shall not be entitled to make a Claim for breach of any Warranty if and to the extent that the Buyer is actually aware at the date of the agreement that it has a right to make such a Claim.
- 6.20 Notwithstanding the fact that this agreement is executed as a deed, no liability of the Sellers shall absent fraud, except as specifically contained in this agreement, continue beyond a six year period from Completion.

#### 7. Buyer Warranties

The Buyer warrants to each of the Sellers that:

- (a) it has the full power and authority to enter into and perform this agreement which constitutes valid and binding obligations on the Buyer which are enforceable in accordance with its terms; and
- (b) it has taken all corporate and other action necessary to enable it to enter into and perform this agreement and has obtained all approvals and consents required by it for the performance by it of the transactions contemplated by this agreement, provided always that no warranty is given in this clause 7 in respect of the need for any competition or anti-trust approval, filing, clearance or consent.



#### 8. Completion

- 8.1 The sale and purchase of the Sale Shares shall be completed at the offices of the Sellers' Solicitors on the Completion Date (or at any other place agreed by the parties in writing).
- 8.2 On Completion the Sellers shall deliver to the Buyer:
  - (a) completed and signed transfers of the Sale Shares to the Buyer and the related share certificates (or an indemnity in a form reasonably acceptable to the Buyer in respect of lost share certificates);
  - (b) the statutory books of each Group Company complete and accurate up to immediately prior to Completion (including the share certificates in relation to each Group Company (or an indemnity in a form reasonably acceptable to the Buyer in respect of lost share certificates)) and at the premises of the Company, any company seal(s), certificates of incorporation, certificates of incorporation on change of name and all unused share certificates of each Group Company and all cheque books and credit cards of each Group Company;
  - (c) the Disclosure Letter in the agreed form executed by the Warrantors together with the DVD of the Data Room in the agreed form;
  - (d) the title deeds and documents relating to the Property at the premises of the Company;
  - (e) the Deed of Termination duly executed by each party to it;
  - (f) copies (which may be by way of electronic screen prints) of bank statements in respect of each account of each Group Company as at 17 October 2013;
  - (g) Short Form Sale and Purchase Agreements duly executed by each of the D Minority Shareholders;
  - (h) letters of resignation in agreed form from Adam Maidment and Brian Phillips; and
  - (i) the Employment Letter duly executed by Andrew McEwan.
- 8.3 On Completion the Sellers shall procure the holding of a meeting of the directors of the Company to do such of the following things as are applicable to it;
  - (a) approve (subject to stamping in respect of the Sale Shares only) the transfers referred to in clause 8.2(a) and the transfers contemplated by the Short Form Sale and Purchase Agreements;
  - (b) appoint the persons nominated by the Buyer as directors (if any) and note the resignations referred to in clause 8.2(h); and
  - (c) resolve to add Mike Zemetra to the bank mandate of the Company's bank account with HSBC.
- 8.4 On Completion the Buyer shall:
  - (a) pay the Consideration by electronic funds transfer in the manner specified in clause 3.1 which payment shall constitute a good discharge for the Buyer of its obligations to pay that amount (and the Buyer shall not be concerned with the application of any such amount between all or any of the Sellers); and



- (b) deliver to the Sellers or to the Sellers' Solicitors (whose receipt shall be a sufficient discharge):
  - (i) a counterpart of the Disclosure Letter signed by the Buyer acknowledging its receipt;
  - (ii) a certified copy of an extract of the minutes of the board of directors of the Buyer authorising the execution and performance by the Buyer of its obligations under this agreement;
  - (iii) a copy of the Insurance Policy.

#### 9. Post-Completion

- 9.1 Each of the Sellers shall and shall procure that any other necessary party shall execute and do all such documents acts and things as the Buyer may reasonably require for transferring to the Buyer the legal and beneficial ownership of the respective Sale Shares held by each of them (it being recognised that the Buyer shall be liable to pay stamp duty in respect of such transfers).
- 9.2 Each Seller:
  - (a) to secure the Buyer's interests under this agreement, irrevocably appoints, for a three month period following Completion, the Buyer to be his attorney pending the Buyer's registration as a member of the Company with power on that Seller's behalf to execute and deliver all deeds and documents and to do all acts and things and exercise all rights (including voting rights) which the Buyer would be entitled to execute, deliver and do and exercise if the Buyer was registered as the holder of the Sale Shares with power to sub-delegate this power and power to appoint a substitute attorney in addition to the Buyer; and
  - (b) agrees that for so long as any Sale Shares remain registered in his name he will:
    - (i) not exercise any of his rights as a member of the Company or appoint any other person to exercise such rights;
    - (ii) hold on trust for and pay or deliver to the Buyer any distributions or notices, documents or other communications which may be received after the date of this agreement by that Seller in his capacity as a member of the Company from the Company or any third party; and
    - (iii) on request by the Buyer ratify all documents executed and acts done by the Buyer as his attorney.
- 9.3 The Buyer shall indemnify each Seller against all losses suffered by the relevant Seller in connection with the exercise of the Buyer's powers under clause 9.2(a).
- 9.4 For a period of 24 months following Completion the Buyer undertakes to procure (or, in the case of any working papers of the auditors of any member of the Group, to use its reasonable endeavours to procure) that each Seller and their respective Advisers are afforded such reasonable access to the books, accounts, working papers and records and other financial information relating to any period prior to Completion of each Group Member as it or they may reasonably require to enable the relevant Institutional Seller and Warrantor):

- (a) to prepare their respective statutory or management accounts; or
- (b) to enable such person to comply with any other Applicable Law.

For the avoidance of doubt, the provisions of clause 11 shall apply to information provided or obtained pursuant to this clause 9.3

#### 10. Non-competition

#### Restrictions

- 10.1 Each Warrantor undertakes with the Buyer and each Group Company that after Completion he will not either himself or by an agent and either on his own account or by or in association with or for the benefit of any other person directly or indirectly for a period of two years following Completion:
  - (a) take up or hold or seek to take up or hold any:
    - (i) office in or with any business which is engaged in the Prohibited Business within the Prohibited Area;
    - (ii) post or position which enables that Warrantor to exercise, whether personally or by an agent and whether on his own account or in association with or for the benefit of any other person, a controlling influence over any business which is engaged in the Prohibited Business within the Prohibited Area; or
    - (iii) employment, engagement or consultancy with any person which is engaged in the Prohibited Business within the Prohibited Area,

which results or would result in that Warrantor being engaged in business activities which are in competition with the Prohibited Business;

- (b) within the Prohibited Area either personally or by an agent and either on his own account or by or in association with any other person or otherwise directly or indirectly engage or seek to engage in any capacity in the Prohibited Business, except that he may hold as an investment not more than 3% of the issued share capital of a company listed or quoted on a Recognised Investment Exchange;
- (c) deal with any Customer or Supplier but this shall not preclude him from dealing with such a Customer or Supplier so long as that Warrantor does not either directly or indirectly deal with that Customer or Supplier in respect of the Prohibited Business;
- (d) canvass, solicit, approach or seek out or cause to be canvassed, solicited, approached or sought out in all cases in respect of the Prohibited Business or by any other means endeavour to entice away from any Group Company, any Customer or Supplier;
- (e) solicit or seek to entice away from any Group Company, or aid or assist any other person or persons in employing or otherwise retaining the services of anyone who is, at Completion, employed by any Group Company or who is a Consultant at Completion otherwise than in a junior administrative or secretarial capacity and provided further that this clause 10.1(e) shall not prohibit the placing of any bona fide job advertisement which is not specifically targeted at the persons referred to herein;



- (f) employ or otherwise retain the services of any of such person as is mentioned in clause 10.1(e) save for persons employed in a junior administrative or secretarial capacity; or
- (g) hold himself out as being interested in or any way connected (other than as a matter of historic fact or as an ongoing employee (if such is the case)) with the Group.
- 10.2 Each Warrantor and each Seller undertakes that he will not whilst any Group Company uses, or has any right to use, any of the names, domain names or marks set out below, directly or indirectly, use in connection with any trade or business the name or any name resembling them or capable of confusion with them: "Travel Entertainment" or "IFE" or any other name or domain name listed in part A or B of schedule 3 or any logo which is similar to or capable of confusion with any existing logo used by any Group Company. For the avoidance of doubt, the provisions of this clause shall not apply to any past, present or future limited partner or general partner of the Institutional Sellers (provided that the Warrantors shall not be excluded pursuant to this provision) and any person holding any form of interest in any such entities whether directly or indirectly (and for the avoidance of doubt any entity in which such entity has invested).
- 10.3 Each Warrantor (and, in the case of clause 10.2, each Seller) acknowledges that the undertakings in clauses 10.1 and 10.2, are reasonable; are integral to the terms on which the Buyer has agreed to purchase the Sale Shares and necessary for the implementation of the purchase; and that each of them is to be construed and take effect independently of the others.
- 10.4 If a breach of clauses 10.1 or 10.2 occurs, the Warrantors (and the Sellers in the case of clause 10.2) and the Buyer agree that damages alone may not to be sufficient compensation and that injunctive relief is reasonable and may be essential to safeguard the interests of the Buyer and of any Group Company and that injunctive relief (in addition to any other equitable remedies) may (subject to the discretion of the courts) be obtained.
- 10.5 No Warrantor shall be treated as committing a breach or violation of the provisions of clause 10 solely when properly and within the terms of his appointment acting as a Director, Consultant or Employee of any Group Company (which for the avoidance of doubt shall include any Group Company following the Completion Date).
- 10.6 If any restrictions in this clause 10, shall be found to be invalid or unenforceable but would be valid if part of the wording of the restriction were deleted or the period for which it applies were reduced or the range of activities or area dealt with by it were reduced in scope, the restriction concerned shall apply with such modifications as may be necessary to make it valid and enforceable.

#### 11. Confidentiality

- 11.1 All obligations of confidence owed by the Buyer to the Sellers relating to the confidential information of the Company shall (without prejudice to any previously accrued rights) cease on Completion and the remaining provisions of this clause 11 shall apply from Completion.
- 11.2 Subject to clauses 11.1 and 11.3, each party undertakes to and shall keep confidential any information which is obtained by it or any of Member of its Group which:



- (a) relates to the negotiation of this agreement or any document referred to in this agreement;
- (b) relates to the provisions or the subject matter of this agreement or of any document referred to in this agreement;
- (c) in the case of a Seller, relates to the Buyer;
- (d) in the case of a Seller, is confidential information which he has acquired about the Company and/or any other Group Company; and
- (e) in the case of the Buyer, relates to the Sellers,

#### (collectively, "Confidential Information").

- 11.3 Clause 11.1 does not apply to information to the extent that:
  - (a) either party (or its Advisers or Associates) is required to disclose it by any Applicable Law or by a Competent Authority;
  - (b) that information is contained in a communication which is consistent with the Announcement;
  - (c) it enters the public domain other than as a result of the unauthorised disclosure by a party or any of its Associates or its or their Advisers;
  - (d) it is in the possession of either party or of any of its Associates or its or their Advisers free from any restriction as to its use or disclosure having been obtained otherwise than from the other party for the purposes of this agreement;
  - (e) a party has disclosed it to any of its Associates or its or their Advisers who need to know such information for the purposes of advising in relation to or furthering the provisions of this agreement and who are aware of the obligations of confidentiality and agree to keep the information confidential and not to use any Confidential Information for any purpose other than the purpose for which it was disclosed;
  - (f) it is required or expedient to vest the full benefit of this agreement in that party or to enforce any of that party's rights under this agreement;
  - (g) it is required or expedient for the Buyer to disclose such information to its current and future debt and equity providers;
  - (h) an Institutional Seller discloses it to the investors in such Institutional Seller or any funds managed by such Institutional Investor or an Associate of the Institutional Seller, provided that the disclosure is made for the purposes of reporting to those investors and is disclosed subject to duties of confidentiality;
  - (i) it is disclosed by one Group Company to another Group Company and/or to directors of any such Group Company following Completion solely for the purposes and to the extent to enable such Group Company to comply with any Applicable Law.
- 11.4 The restrictions contained in clause 11.2 shall continue to apply after Completion without limit in time.

#### 12. Notices and other Communications

- 12.1 Where this agreement provides for the giving of notice or the making of any other communication, such notice or communication shall not (unless otherwise expressly provided) be effective unless given or made in writing in English in accordance with the following provisions of this clause 12.
- 12.2 Any notice or communication to be given or made under or in connection with this agreement shall be delivered or sent by post where notice is to be given to the person in column 1 below to the address set out in column 2 below to:

1	2	
The Buyer	the address set out at the beginning of this agreement, for the attention of Robert Haxton	
	with a copy to Jay Itzkowitz at Global Eagle Entertainment Inc., 4553 Glencoe Avenue, Marina Del Rey, CA 90292 and a copy to the Buyer's Solicitors, for the attention of Ylan Steiner	
The Institutional Sellers	the address as set out in schedule 1	
Estibaliz Asiain Sancho	the address as set out in schedule 1	
Steven Gosling	the address as set out in schedule 1	
Andrew McEwan	the address as set out in schedule 1	

(such addresses being referred to below as the "Postal Address" of the relevant party).

- 12.3 Any notice or other communication so delivered or sent shall be deemed to have been served at the time when it arrives at the address to which it is delivered or sent except that:
  - (a) if such notice or other communication is sent by post in the manner described above it shall, subject to evidencing it has been despatched, be deemed to have been received on the second Business Day after posting the same (or fifth Business Day if sent to or from an address outside the United Kingdom); and
  - (b) if the time at which it is received or deemed received is between 5.30 p.m. on a Relevant Day and 9.00 a.m. on the next Relevant Day it shall be deemed to have been served at 9.00 a.m. on the second of such Relevant Days.
- 12.4 Where any party has given notice to the others of any different address or number to be used for the purposes of this clause 12 then such different address or number shall be substituted for that shown above.

For the purposes of this clause 12:

- (a) "**Relevant Day**" means any day other than a Saturday, Sunday or a day which is a public holiday at the Postal Address of the receiving party; and
- (b) any reference to a time is to the time at the Postal Address of the receiving party.



#### 13. Third party rights

- 13.1 The obligations of confidentiality in paragraphs (a), (b) and (d) of clause 11.2 are assumed for the benefit of each Group Company. Each Group Company may rely on and enforce the obligations of confidentiality accepted by the Sellers and/or Buyer in such clauses.
- 13.2 Any person to whom the benefit of the Warranties or any other rights of the Buyer under this agreement are assigned under clause 14 may rely on and enforce the Warranties and any such rights.
- 13.3 The Sellers and the Buyer may by agreement in writing rescind, terminate or vary the provisions of this agreement without the consent of any third party and, accordingly, section 2(1) of the Contracts (Rights of Third Parties) Act 1999 shall not apply.
- 13.4 The provisions of clauses 5.8,10, 13.1 and 13.2 to the extent expressed to be made in favour of or capable of applying for the benefit of a member of the Group or any officer or employee of such member, or of the Buyer or permitted assignee or other holder of the Sale Shares subject to any Applicable Law, such person shall be entitled, with the prior written consent of the Buyer be enforceable by such persons in accordance with the Contracts (Rights of Third Parties) Act 1999.
- 13.5 Except as provided in clauses 5.8, 10, 13.1, 13.2 and 13.4, no term of this agreement is enforceable under the Contracts (Rights of Third Parties) Act 1999 by a person who is not a party.

#### 14. Assignment

- 14.1 Subject to clause 14.2 or as otherwise provided, all rights and benefits under this agreement are personal to the parties and may not be assigned at law or in equity without the prior written consent of each other party.
- 14.2 The Buyer may assign (in whole or in part) the benefit of this agreement to any other member of the Buyer's Group provided that, if such assignee ceases to be a member of the Buyer's Group, the Buyer will procure that all the benefits relating to this agreement that have been assigned to such assignee are re-assigned to the Buyer immediately before such cessation provided always, that the liability of the Sellers under this agreement or any other relevant Transaction Document shall be no greater in the event of an assignment (as permitted by this clause 14.2) than it would have been in the absence of such assignment.
- 14.3 The Buyer may also assign or charge its rights under this agreement (by way of security only) to any banks and/or financial institutions lending money or making other banking facilities available to the Buyer (or any Member of its Group) in connection with the acquisition of the Sale Shares (or on any re-financing of such debt) but so that, notwithstanding any such assignment in security, the Sellers may, unless they receive written notice of enforcement of the relevant security interest, deal with the Buyer in connection with all matters arising under this agreement and provided always that the liability of the Sellers under this agreement or any other relevant Transaction Document shall be no greater in the event of an assignment (as permitted by this clause 14.3) than it would have been in the absence of such assignment.

#### 15. Entire agreement

15.1 The Transaction Documents constitute the entire agreement between the parties about the subject matter of this agreement and supersede and extinguish all earlier understandings and agreements between any of the parties and all earlier representations by any party about such subject matter.



- 15.2 The parties have not entered into the Transaction Documents in reliance upon any representation, warranty or promise that is not set out in any such Transaction Document and no such representation or warranty or any other term is to be implied in them whether by virtue of any usage or course of dealing or otherwise except as expressly set out in them.
- 15.3 If a party has made or given any representation, warranty or promise, or otherwise made any innocent or negligent misrepresentation, then (except to the extent that it has been set out in the Transaction Documents) the party to whom it is given or made waives any rights or remedies which it may have in respect of it and agrees that the other party shall have no liability in respect of it.
- 15.4 Clause 15.3 shall not exclude the liability of a party for fraud or fraudulent misrepresentation or concealment or any resulting right to rescind any of the Transaction Documents.

#### 16. Miscellaneous

- 16.1 Except as required by Applicable Law or by a Competent Authority, all third party communications by any party or any Group Member of any party shall be consistent with the terms of the Announcement.
- 16.2 Except where this agreement provides otherwise, each party shall bear its own costs incurred in relation to the negotiation and preparation of this agreement and matters incidental to this agreement.
- 16.3 Unless expressly provided otherwise, all obligations or representations entered into or made in this agreement by each of the Sellers are entered into in relation to themselves only.
- 16.4 The Buyer shall pay any stamp duty, transfer tax or similar fees in respect of the transfer of the Sale Shares.
- 16.5 This agreement shall so far as it remains to be performed after Completion continue in force notwithstanding Completion and the rights of the Buyer in respect of any Transaction Document shall not be affected by Completion.
- 16.6 No waiver by a party of any requirement of this agreement or any right which he has under it shall be valid unless such waiver is in writing signed by him.
- 16.7 No omission to exercise, or delay by any party in exercising, any right under this agreement shall operate as a waiver of such right nor shall any single or partial exercise of any right preclude the exercise of any other right.
- 16.8 The Buyer may release or compromise the liability of, or institute proceedings or obtain judgment against, a Seller under this agreement, or grant to a Seller time or other indulgence without affecting the liability of any other Seller under this agreement or the Buyer's rights against any other party.
- 16.9 This agreement may consist of any number of duplicates each executed by at least one party, each of which when so executed and delivered shall be an original, but all the duplicates shall together constitute one instrument.
- 16.10 If a term of this agreement shall be held to be illegal, invalid or unenforceable it shall to that extent be deemed not to form part of this agreement, but the enforceability of the remainder of this agreement shall not be affected.



- 16.11 Other than such equitable remedies as may be available, and in the absence of fraud, the sole remedy of the Buyer in relation to the subject matter of this agreement shall lie in contract only.
- 16.12 The rights of the Buyer under this agreement are independent, cumulative and without prejudice to all other rights available to it whether as a matter of common law, statute, custom or otherwise.
- 16.13 If a party fails to pay any sum payable by it on the due date for payment under this agreement, it shall pay interest on the overdue sum for the period from and including the due date of payment up to the date of actual payment (after as well as before judgment) at the rate of three per cent (3 per cent) above the base rate from time to time of Barclays Bank plc.
- 16.14 All payments to be made under this agreement shall be made in full without any set-off or counterclaim and free from any deduction or withholding save as may be required by law in which event such deduction or withholding shall not exceed the minimum amount which it is required by law to deduct or withhold and the payer will simultaneously pay to the payee such additional amounts as will result in the receipt by the payee of a net amount equal to the full amount which would otherwise have been receivable had no such deduction or withholding been required.
- 16.15 If Taxation is payable on any sum paid by a Seller under this agreement, the sum otherwise so payable shall be grossed up by such amount as will ensure that, after payment of any Taxation charged on or in respect of such payment, there shall be left a sum equal to that which would otherwise be payable under this agreement. If a sum is increased by virtue of this clause 16.15 and the Buyer, acting reasonably, determines that a Credit or Relief attributable to such increased payment has been obtained and utilised by it, then the Buyer shall as soon as reasonably practicable account to the Seller for such proportion of the Credit or Relief as will the Buyer in no better or worse position than if no such Taxation had been required to be made in respect of the payment. The Buyer shall use reasonable endeavours to claim and/or obtain and utilise any such Credit or Relief.
- 16.16 This agreement together with the Transaction Documents shall, to the extent that it remains to be performed, continue in full force and effect notwithstanding Completion.

#### 17. Governing Law and Jurisdiction

- 17.1 The governing law of this agreement, and of any claim, dispute or issue arising out of or in connection with this agreement or its subject matter (including non-contractual claims), shall be that of England and Wales.
- 17.2 The courts of England and Wales shall have exclusive jurisdiction to settle any claim, dispute or issue between the parties whether arising out of or in connection with this agreement or its subject matter, or otherwise (including non-contractual claims). The parties to this agreement irrevocably submit to such jurisdiction and waive any objection to it, on the ground of inconvenient forum or otherwise. No party shall oppose the recognition or enforcement of a judgment, order or decision of those courts in respect of any such claim or dispute by the courts of any state which, under the laws and rules applicable in that state, are competent or able to grant such recognition or enforcement.

#### 18. Spanish Subsidiary Undertakings

18.1 The Warrantors shall use their reasonable endeavours at their expense to cause the Spanish located subsidiaries of the Company set out in schedule 2 to become properly registered and will clear the aprovidencia de apremio with the town council of Madrid published on 22 October 2012.



# **SCHEDULE 1**

#### The Sellers and the Consideration

Name	Address	Number and Class of Sale Shares	Consideration
GCP Capital Partners LLP acting as manager for Greenhill Capital Partners	7 <sup>th</sup> Floor Lansdowne House Berkeley Square	23,230,757 A Preference Shares	5,557,334.93
Europe LP	London W1J 6ER	7,724,500 A Ordinary Shares	0
		60,594,509 B Preference Shares	0
GCP Capital Partners LLP acting as manager for Greenhill Capital Partners	7 <sup>th</sup> Floor Lansdowne House Berkeley Square	28,240,125 A Preference Shares	6,755,691.74
Europe (Employees) LP	London W1J 6ER	9,390,200 A Ordinary Shares	0
		73,660,810 B Preference Shares	0
GCP Capital Partners LLP	7 <sup>th</sup> Floor	29,729,118 A Preference Shares	7,111,893.33
acting as manager for Parallel Private	Lansdowne House Berkeley Square		
Equity Greenhill LP	London W1J 6ER	9,885,300 A Ordinary Shares	0
		77,544,681 B Preference Shares	0
Steven Gosling	5 Little Heath Lane Dunham Massey Altrincham Cheshire WA14 4TS	6,015,700 D Ordinary Shares	631,235.99
Andrew McEwan	15 Holly Road North Wilmslow	12,140,700 D Ordinary Shares	1,273,940.99
	Cheshire SK9 1LX		
Estibaliz Asiain Sancho	Calle Monte Umbrio 49 Urbanizacion Ciudalgolf 28707 Cuidalcampo Madrid, Spain	6,125,000 D Ordinary Shares	642,705.00

28

# **SCHEDULE 2**

# Details of the Company and the Subsidiary Undertakings

# Part A

# The Company

Name:	Travel Entertainment Group Equity Limited
Former Names:	TEG Equity Limited
Number:	06646044
Date of Incorporation:	15/07/2008
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA
Issued share capital:	27,000,000 A Ordinary Shares of £0.001 each
	28,000,000 D Ordinary Shares of £0.001 each 81,200,000 A Preference Shares of £0.10 each 211,800,000 B Preference Shares of £0.10 each
Shareholders:	<ol> <li>GCP Capital Partners LLP acting as manager for Greenhill Capital Partners Europe LP</li> <li>GCP Capital Partners LLP acting as manager for Greenhill Capital Partners Europe (Employees) LP</li> <li>GCP Capital Partners LLP acting as manager for Parallel Private Equity Greenhill LP</li> <li>Steven Gosling</li> <li>Andrew McEwan</li> <li>Estibaliz Asiain Sancho</li> <li>Howard Clark</li> <li>Simon Howard</li> <li>Rouba Korfali</li> <li>Adrian Lambert</li> <li>Flavia Nogueira</li> <li>Zoe Tierney</li> <li>Michael White</li> <li>Pamela McEwan</li> <li>Alexandra White</li> <li>Alexandra White</li> <li>Sarah Howard</li> <li>Matthew Bunker</li> <li>Alison Clark</li> </ol>
Directors:	<ul> <li>(1) Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>

	(2)	Estibaliz Asiain Sancho Calle Monte Umbrio 49 Urbanizacion Cuidalgolf 28707 Cuidalcampo Madrid Spain
	(3)	Adam Maidment 18 Pandora Road London NW6 1TT
	(4)	Andrew McEwan 15 Holly Road North Wilmslow Cheshire SK9 1LX
	(5)	George Brian Phillips 6 Abbotswood Drive Weybridge Surrey KT13 0LT
Secretary:		DWF Secretarial Services Limited 1 Scott Place 2 Hardman Street Manchester M3 3AA
Accounting reference date:		31/12
Last accounts:		31/12/2012
Charges:	(1)	Guarantee and Debenture in favour of Barclays Bank Plc on 26/05/2010 secured over all monies due or to become due from the Company and/or all or any of the companies named therein to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.
	(2)	Debenture created in favour of HSBC Bank Plc on 06/12/2012 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.

### Part B

# Subsidiary Undertakings

# **Direct Subsidiary**

Name:	Travel Entertainment Group Acquisitions Limited	
Former Names:	TEG Acquisitions Limited	
Number:	06646009	
Date of Incorporation:	15/07/2008	
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA	
Issued share capital:	290,000 ordinary shares of £1 each	
Shareholder:	Travel Entertainment Group Equity Limited	
Directors	<ul> <li>Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>	
	<ul> <li>(2) Adam Maidment</li> <li>18 Pandora Road</li> <li>London</li> <li>NW6 1TT</li> </ul>	
	<ul> <li>(3) Andrew McEwan</li> <li>15 Holly Road North</li> <li>Wilmslow</li> <li>Cheshire</li> <li>SK9 1LX</li> </ul>	
Secretary:	DWF Secretarial Services Limited 1 Scott Place 2 Hardman Street Manchester M3 3AA	
Accounting reference date: 31/12		
Last accounts:	31/12/2012	
Charges:	(1) Guarantee and Debenture in favour of Barclays Bank Plc on 26/05/2010 secured over all monies due or to become due from the Company and/or all or any of the companies named therein to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.	
	(2) Debenture created in favour of HSBC Bank Plc on 07/12/2012 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.	

### Indirect subsidiaries

Name:	Travel Entertainment Group Limited		
Former Names:	Templeco 651 Limited		
Number:	06024017		
Date of Incorporation:	11/12/2006		
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Issued share capital:	816,649,359 ordinary shares of £0.01 each		
Shareholder:	Travel Entertainment Group Acquisitions Limited		
Directors:	<ul> <li>Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>		
	<ul> <li>(2) Estibaliz Asiain Sancho Calle Monte Umbrio</li> <li>49 Urbanizacion Cuidalgolf</li> <li>28707 Cuidalcampo Madrid</li> <li>Spain</li> </ul>		
	<ul> <li>(3) Andrew McEwan</li> <li>15 Holly Road North</li> <li>Wilmslow</li> <li>Cheshire</li> <li>SK9 1LX</li> </ul>		
Secretary:	DWF Secretarial Services Limited 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Accounting reference date: 31/12			
Last accounts:	31/12/2012		
Charges:	(1) Guarantee and Debenture in favour of Barclays Bank Plc on 26/05/2010 secured over all monies due or to become due from the Company and/or all or any of the companies named therein to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.		
	(2) Debenture created in favour of HSBC Bank Plc on 07/12/2012 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charges over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.		

Name:	IFE Holdings Limited		
Former Names:	Templeco 523 Limited		
Number:	04185773		
Date of Incorporation:	23/03/2001		
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Issued share capital:	33,707 Ordinary Shares of £1 each		
Shareholder:	Travel Entertainment Group Limited		
Directors:	<ul> <li>Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>		
	<ul> <li>(2) Estibaliz Asiain Sancho</li> <li>Calle Monte Umbrio</li> <li>49 Urbanizacion Cuidalgolf</li> <li>28707 Cuidalcampo</li> <li>Madrid</li> <li>Spain</li> </ul>		
	<ul> <li>(3) Andrew McEwan</li> <li>15 Holly Road North</li> <li>Wilmslow</li> <li>Cheshire</li> <li>SK9 1LX</li> </ul>		
Secretary:	DWF Secretarial Services Limited 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Accounting reference date	: 31/12		
Last accounts:	31/12/2012		
Charges:	Debenture created in favour of HSBC Bank Plc on 07/12/2012 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charges over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.		

Name:	IFE Services Limited			
Former Names:	Hadenlink Limited			
Number:	03395338			
Date of Incorporation:	30/06/1997			
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA			
Issued share capital:	3,350 Ordinary Shares of £1 each			
Shareholder:	IFE Holdings Limited			
Directors:	<ul> <li>Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>			
	<ul> <li>(2) Estibaliz Asiain Sancho</li> <li>Calle Monte Umbrio</li> <li>49 Urbanizacion Cuidalgolf</li> <li>28707 Cuidalcampo</li> <li>Madrid</li> <li>Spain</li> </ul>			
	<ul> <li>(3) Andrew McEwan</li> <li>15 Holly Road North</li> <li>Wilmslow</li> <li>Cheshire</li> <li>SK9 1LX</li> </ul>			
	<ul> <li>Howard Clark</li> <li>134 Bare Lane</li> <li>Morecambe</li> <li>Lancashire</li> <li>LA4 6RS</li> </ul>			
Secretary:	DWF Secretarial Services Limited 1 Scott Place			
	2 Hardman Street Manchester M3 3AA			
Accounting reference date	31/12			
Last accounts:	31/12/2012			
Charges:	<ol> <li>Legal Charge created in favour of Barclays Bank Plc on 17/08/2007 over all monies due or to become due from the Company to the chargee on any account whatsoever. F/H Unit 1 Haig Road, Parkgate Industrial Estate, Knutsford, Cheshire, T/NO CH205013.</li> </ol>			
	(2) Guarantee and Debenture in favour of Barclays Bank Plc on 26/05/2010 secured over all monies due or to become due from the Company and/or all or any of the companies named therein to the chargee on any account whatsoever. Fixed and floating charge over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.			
	(3) Debenture created in favour of HSBC Bank Plc on 07/12/2012 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charges over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.			



Name:	IFE S	ervices Germany Limited
Former Names:	PMG	TEG Holdco Limited
Number:	08010	946
Date of Incorporation:	29/03/	/2012
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA	
Issued share capital:	2 ordi	nary shares of £1 each
Shareholder:	IFE S	ervices Limited
Directors:	(1)	Steven Gosling
	(2)	Andrew McEwan
Secretary:	N/A	
Accounting reference date:	31/03	
Last accounts:	Not ye	et filed.
Charges:	N/A	

Name:	IFE Services (USA), Inc		
Former Names:	None		
Number:	EIN: 77-0692120		
Date of Incorporation:	21 <sup>st</sup> June 2007		
Registered Office:	Principal place of business and CA address for Service of Process		
	21062 Bake Parkway, Lake Forest, CA 92630, USA		
Issued share capital:	$1,000 \ge 0.01 \text{ shares} = \$10.$		
Shareholder:	IFE Holdings Limited		
Directors:	(1) Steven Gosling		
	(2) Andrew McEwan		
	Victor Hernandez is listed on the tax return under compensation of officers, but is not on the annual Delaware return as an official "Officer".		
Accounting reference date:	31 <sup>st</sup> December		
Last accounts:	31 <sup>st</sup> December 2012		
Charges:	None		
36			

Name:	In flight Entertainment Spain SL			
Former Names:	N/A			
Number:	Mercantile Registry of Madrid			
	Volume: 20.454			
	Folio: 143			
	Section: 8			
	Leaf: M-361813			
	Inscription: 1st			
Date of Incorporation:	23/09/2004			
Registered Office:	C/ Conde de Peñalver, 38-5ª 28006 Madrid, Spain			
Issued share capital:	3.100 Ordinary Shares of $\notin$ 1 nominal value each = 3.000 $\notin$			
Shareholder:	IFE Holdings Limited			
Directors:	<ul> <li>Steven Gosling</li> <li>5 Little Heath Lane</li> <li>Dunham Massey</li> <li>Altrincham</li> <li>Cheshire</li> <li>WA14 4TS</li> </ul>			
	(2) Andrew McEwan 15 Holly Road North Wilmslow Cheshire SK9 1LX			
Secretary:				
Accounting reference date:	31/12			
Last accounts:	31/12/2012			
Charges:				

Name:	Transı	nedia Publicidad SL	
Former Names:	None		
Number:			
Number.	Mercantile Registry of Madrid		
	Volun	ne: 21.350	
	Folio:	199	
	Sectio	n: 8	
	Leaf:	M-379472	
	Inscrip	ption: 1st	
Date of Incorporation:	06/04/	2005	
Registered Office:		nde de Peñalver, 38-5ª Madrid, Spain	
Issued share capital:	101 O	rdinary Shares of € 30,06 each	
Shareholder:	In flig	ht Entertainment Spain SL	
Directors:	(1)	Steven Gosling 5 Little Heath Lane Dunham Massey Altrincham Cheshire WA14 4TS	
	(2)	Andrew McEwan 15 Holly Road North Wilmslow Cheshire SK9 1LX	
Secretary:			
Accounting reference date:	31/12		
Last accounts:	31/12/	2012	
Charges:			

Name:	IFE Services Asia Pacific PTE Limited			
Former Names:				
Number:	20091	2405N		
Date of Incorporation:				
Registered Office:	2 Kall	2 Kallang Pudding Road,#06-10 Mactech building, Singapore, 349307		
Issued share capital:	1 x SGD 1 share			
Shareholder:	IFE Holdings Limited			
Directors:	(1)	Steven Gosling		
	(2)	Andrew McEwan		
	(3)	Estibaliz Asian		
	(4)	Ha Wenjie		
Secretary:	Ha Wenjie			
Accounting reference date:	te: 31 <sup>st</sup> December			
Last accounts:	31 <sup>st</sup> December 2011			
Charges:	None			

Name:	Oceans TV, Inc
Former Names:	None
Number:	EIN: 20-8597379
Date of Incorporation:	13 <sup>th</sup> February 2007
Registered Office:	21062 Bake Parkway, Lake Forest, CA 92630, USA
Issued share capital:	\$10
Shareholder:	IFE Holdings Limited
Directors:	(1) Steven Gosling
	(2) Andrew McEwan
Accounting reference date:	31 <sup>st</sup> December
Last accounts:	31 <sup>st</sup> December 2012
Charges:	None

Name:	The In-Flight Entertainment Company Limited		
Former Names:	Dotgift Limited		
Number:	02363496		
Date of Incorporation:	21/03/1989		
Registered Office:	c/o DWF LLP 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Issued share capital:	2 ordinary shares		
Shareholder:	IFE Services Limited		
Directors:	<ul> <li>(1) Steven Gosling <sup>5</sup> Little Heath Lane Dunham Massey Altrincham Cheshire WA14 4TS     </li> <li>(2) Andrew McEwan 1 Cranford Drive Knutsford Business Park     </li> </ul>		
	WA16 8ZR		
Secretary:	DWF Secretarial Services Limited 1 Scott Place 2 Hardman Street Manchester M3 3AA		
Accounting reference date: 31/12			
Last accounts:	31/12/2012		
Charges:	(1) Debenture created in favour of Barclays Bank Plc on 30/09/1996 secured over all monies due or to become due from the Company to the chargee on any account whatsoever. Fixed and floating charges over the undertaking and all property and assets present and future, including goodwill, book debts, uncalled capital, buildings, fixtures, fixed plant and machinery.		

Name:	IFE Services SA (PTE) Limited		
Former Names:	IFE Services SA Proprietary Limited		
Number:	2011/008595/07		
Date of Incorporation:	15 <sup>th</sup> April 2011		
Registered Office:	87A, 9 <sup>th</sup> Street, Parkhurst, Johannesburg, 2196		
Issued share capital:	120 x 1 ZAR shares		
Shareholder:	(1)	IFE Services Limited	89 shares
	(2)	Tebogo Patrick Selokane	21 shares
	(3)	Unati Gwija	10 shares
Directors:	(1)	Steven Gosling	
	(2)	Andrew McEwan	
	(3)	Tebogo Patrick Selokane	
	(4)	Unati Gwija	
Secretary:	KPMG Services Proprietary Limited KPMG Forum 1226 Schoeman St Hatfield 0083		
Accounting reference date:	31 <sup>st</sup> December		
Last accounts:	None since incorporation		
Charges:	None		

\_\_\_\_

# 49% Share in Joint Venture

Name:	Donica Inflight Entertainment Services Inc
Former Names:	None
Number:	45-2385828
Date of Incorporation:	20 <sup>th</sup> May 2011
Registered Office:	21062 Bake Parkway, Lake Forest, California, 92630, USA
Issued share capital:	100 x USD 0.01 = \$1
Shareholder:	Donica International, Inc x 51 shares
	IFE Services (USA), Inc x 49 shares
Directors:	(1) Steve Gosling
Secretary:	(1) Guy Kennett
Accounting reference date:	31 <sup>st</sup> December
Last accounts:	None filed
Charges:	None

# **SCHEDULE 3**

# Intellectual property

# Part A

# Trade marks

The Company has no registered trade marks.

# Part B

# Domain Names

Domain	Renewal Date
ifeservices.co.uk	02/04/2014
ifeservices.com	23/07/2014
oceanstv.com	14/02/2014
travelentertainmentgroup.com	15/02/2014
travelentertainmentgroup.es	01/06/2014
travelentertainmentgroup.cn	01/06/2014
travelentertainmentgroup.sg	01/06/2014
ifeservices.es	01/06/2014
ifeservices.cn	01/06/2014
ifeservices.sg	01/06/2014
donicaifes.com	01/06/2014
hollywoodatsea.com	01/06/2014
theinflightentertainment.com	01/06/2014
theinflightentertainment.co.uk	01/06/2014

#### **SCHEDULE 4**

#### Warranties

#### Part 1

#### General

#### 1. DISCLOSURE OF INFORMATION

The factual information set out in schedule 2 is accurate and not misleading.

## 2. CAPACITY AND OWNERSHIP OF SHARES

- 2.1 Each Warrantor has full power and authority and has taken all action necessary to execute and deliver and to exercise his rights and perform his obligations under this agreement and each of the documents in the agreed form to be executed on or before Completion.
- 2.2 None of the Warrantors has any interest, directly or indirectly, in any business other than that now carried on by any Group Company which is competitive with the business of any Group Company.
- 2.3 The Sale Shares and the D Minority Shares constitute the whole of the allotted and issued share capital of the Company, are fully paid and have been properly allotted and issued. The issued shares of each Subsidiary Undertaking specified in schedule 2 constitute the whole of the allotted and issued share capital of the Subsidiary Undertaking concerned, are fully paid and have been properly allotted and issued.
- 2.4 There is no Encumbrance on, over or affecting the shares in the Subsidiary Undertakings or any shares in the capital of the Company and/or any Subsidiary Undertaking and there is no agreement or commitment to give or create any Encumbrance and no claim has been made by any person to be entitled to an Encumbrance in relation to such share capital.
- 2.5 There is no litigation, arbitration, prosecution, administrative or other legal proceedings or dispute in existence or, so far as the Warrantors are aware, threatened against any of the Warrantors in respect of the Sale Shares held by them or any other shares held in the Company by any other shareholder or against any Group Company in respect of the shares in the Subsidiary Undertakings or the entitlement of any shareholder in the Company to dispose of their shares in the Company or any Group Company's ownership of the shares in the Subsidiary Undertakings.
- 2.6 Other than this agreement, there is no agreement, arrangement or obligation requiring the creation, allotment, issue, transfer, redemption or repayment of, or the grant to any person of the right (whether conditional or not) to require the allotment, issue, transfer, redemption or repayment of, any shares in the capital of, or loan capital in, any Group Company (including, without limitation, an option or right of pre-emption or conversion).
- 2.7 The Company has not exercised nor purported to exercise or claimed any lien over the Sale Shares held by the Sellers or so far as the Warrantors are aware any other shares in the Company.

# 3. ACCOUNTS AND MANAGEMENT ACCOUNTS

Accounts



- 3.1 The Accounts have been prepared and audited in accordance with UK GAAP at the time they were audited and give a true and fair view of the financial position and state of affairs of each of the Group Companies to which they relate as at the Accounts Date and of the profits and losses for the period covered by the Accounts..
- 3.2 Except as noted in the Accounts, the Accounts have been prepared and audited on a basis consistent with the accounting policies used in the preparation of the audited accounts of each of the Group Companies for the two preceding financial years.
- 3.3 The audited balance sheets and profit and loss accounts for those Group Companies where such accounts have been audited for the financial years of the Company ended on the Accounts Date complied with the requirements of all relevant laws then in force and with Accounting Standards and with UK GAAP then in force.
- 3.4 Except as stated in the audited balance sheets and profit and loss accounts of the Accounts, no material changes in the policies of accounting have been made therein for any of those financial periods.
- 3.5 The Accounts make full and proper provision in accordance with UK GAAP for (or, if appropriate, disclose by way of note) all assets and liabilities and all capital and financial commitments of the Group Companies to which they relate as at or on the Accounts Date.
- 3.6 The Accounts make adequate provision in accordance with UK GAAP for (or, if appropriate, disclose by way of note):
  - (a) contingent assets;
  - (b) liabilities whether actual, contingent, unquantified or disputed;
  - (c) guarantees;
  - (d) any amounts written off against the value of assets; and
  - (e) bad and doubtful debts.

#### Management Accounts

3.7 The Management Accounts have been carefully prepared in good faith and on a basis consistent with the previous monthly management accounts of the Group over the 12 month period ending on the day preceding the period to which the Management Accounts relate and fairly present the state of affairs of each of the Group Companies as at the Locked Box Date and of the profits and losses for the period ended on such date and are not materially misstated.

### 4. **POSITION SINCE THE ACCOUNTS DATE**

Since the Accounts Date:

- 4.1 the business of each Group Company has been carried on in the ordinary and usual course without material interruption or material alternation in its nature, scope or manner and so as to maintain each Group Company as a going concern;
- 4.2 no Group Company has allotted or issued or agreed to allot or issue any share or loan capital;
- 4.3 there has been no material deterioration in the turnover or gross profit of the Group;



- 4.4 none of the Group Companies has acquired or disposed of or agreed to acquire or dispose of any business or any asset having a value in excess of £25,000 (excluding VAT) or assumed or acquired any material liability, obligation or expense having a value in excess of £25,000 (excluding VAT) (including a contingent liability) otherwise than in the ordinary course of business;
- 4.5 none of the Group's top 10 suppliers by reference to the Group's cost of sales in the Company's last financial year have threatened to commence legal proceedings against any Group Company in relation to the relevant Group Company's payment and timing of payment for any such supplies;
- 4.6 no debtor (the book value of whose debt exceeded £10,000) has been released by any Group Company on terms that he pays less than the book value of his debt (other than agreed discounts for early settlement in the ordinary course of business), and no such debt owing to any Group Company has, so far as the Warrantors are aware, been subordinated or written off or has proved, to any extent, irrecoverable;
- 4.7 no Group Company has changed their policies or practices as to collection of debtors or payment of creditors and there has been no acceleration of payments or accruals thereof;
- 4.8 no Group Company has made payments or transferred assets (other than in the ordinary course of business and on arm's length commercial terms) to, or assumed or indemnified liabilities for the benefit of, a Seller or any person Connected with a Seller;
- 4.9 no Group Company has dismissed or given notice of termination to any Employee earning more than £50,000 per annum (a "Senior Employee") (unless circumstances exist justifying summary dismissal) or made any material change in the employment terms of a Senior Employee;
- 4.10 no Group Company has employed any person as a Senior Employee;
- 4.11 all payments, receipts and invoices of any Group Company have, so far as the Warrantors are aware, been fully and accurately recorded in all material respects in the books of the relevant Group Company in accordance with the usual practice of that Group Company and as required by Applicable Law;
- 4.12 there has not been any capitalisation of reserves of any Group Company and no Group Company has repaid any loan capital in whole or in part nor has it, by reason of any default by it in its obligations, become bound or liable to be called upon to repay prematurely any loan capital;
- 4.13 there has been no resolution of or agreement by the members of any Group Company or any class thereof (except as duly filed at Companies House or as provided in this agreement or with the prior written consent of the Buyer) and in particular there has been no capital reorganisation, purchase of own shares by any Group Company, any redemption or cancellation of shares of any Group Company or other change in the capital structure of any Group Company,
- 4.14 the Company has not declared or paid any dividend or other distribution to its shareholders;
- 4.15 no Group Company has changed its accounting reference period;
- 4.16 no supplier to or customer of any Group Company who accounted for more than 5% of the Group's annual turnover in the last financial year has ceased to trade with the Group or, so far as the Warrantors are aware, notified any Group Company of its intention to do so and the Group's business has not been materially affected by any change in the terms of business with such a supplier or customer and, so far as the Warrantors are aware, no Group Company has been notified of any such proposed change; and



4.17 no Group Company has borrowed or raised any money or taken up any financial lending facilities.

### 5. LICENCES AND CONSENTS

The Group has obtained all material licences, permissions, authorisations and consents required to own and operate its assets and for the proper carrying on of its business as carried on up to and including the date of this agreement. So far as the Warrantors are aware, all such licences, permissions, authorisations and consents are in full force and effect and no Group Company is in breach of any of the terms and conditions attached thereto and, so far as the Warrantors are aware, there are no circumstances which indicate that any of such licences, permissions, authorisations or consents may be revoked or not renewed in the ordinary course of events.

### 6. ASSETS

- 6.1 Each Group Company owns free from Encumbrance all assets necessary to carry on its business in all material respects as carried on at the date of this agreement.
- 6.2 All the material property and assets which are described and included in the Accounts or which are used in connection with the business of the Group are:
  - (a) legally and beneficially owned by the Group;
  - (b) in the possession or under the control of the Group;
  - (c) free from all Encumbrances and there is not any agreement or commitment to give or create, and no claim has been made by any person entitled to any Encumbrance; and
  - (d) situated at the Property or otherwise in the possession or under the control of the Group.
- 6.3 None of the assets referred to in paragraph 6.1 of this schedule 4 are the subject of any assignment, royalty, overriding royalty, factoring arrangement, leasing or hiring agreement, hire purchase agreement for payment on deferred terms or any similar agreement or arrangement.
- 6.4 So far as the Warrantors are aware, no Group Company has ever been a party to any transaction with any third party or parties which, in the event of any such third party going into liquidation or an administration order or a bankruptcy order being made in relation to it or him, would constitute (in whole or in part) a transaction at an undervalue or part of a general assignment of debts, under sections 238 to 245 (inclusive) and 339 to 344 (inclusive) of the Insolvency Act 1986.

# 7. INSURANCE

- 7.1 Details of the extant insurance cover taken out by the Group are set out in the Data Room all of which policies of insurance are, so far as the Warrantors are aware, in effect.
- 7.2 Each Group Company has at all material times been and is at the date of this agreement insured against accident, occurrence, damage, injury (including death therefrom), third party loss (including, but not limited to, product liability), credit risk, loss of profit, wrongful acts, errors and omissions (including, but not limited to, by directors and officers), environmental liability and such insurance is for the full replacement value and contains no provision for deduction or excess.



7.3 So far as the Warrantors are aware, no Group Company has done or omitted to do or suffered anything to be done or not to be done which has or might render any policies of insurance taken out by it or by any other person in relation to any of the Group's assets void or voidable or which would or might result in an increase in the rate of premiums on the said policies and there are no claims outstanding and no circumstances which the Warrantors are aware give rise to any claim under any of such policies of insurance.

#### 8. RECORDS

All the accounts, books, registers, ledgers and financial and other material records of whatsoever kind of each Group Company (including all invoices and other records required for VAT purposes) are up to date, in its possession or under its control and have been properly kept and compiled in all material respects in accordance with all Applicable Laws and so far as the Warrantors are aware there are no material discrepancies or inaccuracies contained or reflected therein.

### 9. CONFIDENTIAL INFORMATION

- 9.1 No Group Company uses any processes and, so far as the Warrantors are aware is engaged in any activities which involve the misuse of any confidential information belonging to any third party.
- 9.2 No Group Company is aware of any actual or alleged misuse by any person of any of its confidential information.
- 9.3 No Group Company has disclosed to any person any of its confidential information except where such disclosure was properly made in the normal course of the Group's business and was made subject to an agreement under which the recipient is obliged to maintain the confidentiality of such confidential information and is restrained from further disclosing or using it other than for the purposes for which it was disclosed by the relevant Group Company.

## **10. INTELLECTUAL PROPERTY**

- 10.1 A Group Company is the sole registered proprietor of all the registered Group Intellectual Property set out in schedule 3.
- 10.2 All registered Group Intellectual Property used by any Group Company is set out in schedule 3 and a Group Company is the registered proprietor of all such domain names.
- 10.3 The particulars as to registration (and applications therefor) of registered Group Intellectual Property owned by any Group Company as set out in schedule 3 are true and accurate and not misleading in any material respect.
- 10.4 The Group Intellectual Property that is material to the business of the Group is subsisting. The Group Intellectual Property registered or subsisting in the United Kingdom that is material to the business of the Group is valid and enforceable and as far as the Warrantors are aware, any Group Intellectual Property registered or subsisting in any other jurisdiction that is material to the business of the Group is valid and enforceable.



- 10.5 The Group Intellectual Property owned by any Group Company has not been and is not the subject of any pending or, so far as the Warrantors are aware, threatened proceedings, actions or claims impugning the title, validity or enforceability of such Group Intellectual Property, and there are, and have been, no proceedings, actions or claims for opposition, cancellation, revocation or rectification of such Group Intellectual Property by any person (including, without limitation, from any employees or former employees of the Group) and there are no facts or matters of which the Warrantors are aware which might give rise to any such proceedings or to any threat to the validity or enforceability of such Group Intellectual Property owned by any Group Company.
- 10.6 So far as the Warrantors are aware, no Group Company is engaged in any activities which, and none of the Group's activities, processes or products or the Group Intellectual Property owned by a Group Company, infringe any Intellectual Property or other rights belonging to or vested in any third party, nor have any such rights been so infringed in the three year period preceding Completion. There are no outstanding claims or, so far as the Warrantors are aware, threatened against any Group Company for infringement of any Intellectual Property used (or which has been used) by it and no such claims have been settled by the giving by any Group Company of any undertakings which remain in force.
- 10.7 None of the Warrantors nor any Group Company has disclosed, nor is obliged to disclose, any know how owned by any Group Company to any third party, other than those of any Group Company's officers or employees or any other third party who are bound by obligations of confidence. Save for officers or employees of a relevant Group Company or manufacturers of goods (or component parts of goods) sold by a Group Company, no third party has had access to any of such know how where it is not bound by obligations of confidence. No Group Company is restricted in their ability to use, or to disclose to any third party, any information or know how.

### 11. COMPUTER SYSTEM

- 11.1 None of the Group's records, systems, controls, data or information are recorded, stored, maintained, operated or otherwise wholly or partly dependent upon or held by any means (including any electronic, mechanical or photographic processes whether computerised or not) which (including all means of access thereto and therefrom) are not under the exclusive ownership and direct control of a Group Company.
- 11.2 So far as the Warrantors are aware, the Computer System has been satisfactorily maintained.
- 11.3 Each Group Company has duly complied with and currently complies with all requirements under the Data Protection Legislation including, without limitation:
  - (a) the data protection principles set out under the Data Protection Act 1998;
  - (b) requests from individuals for access to personal data held by it;
  - (c) the requirements relating to the registration and/or notification of processing of personal data;
  - (d) where necessary, under the Data Protection Act 1998, the consent of the data subjects to the processing of personal data relating to them has been obtained.
- 11.4 The Computer System (and each part of it) has functioned in a manner which has not caused any material disruption to the business of the Group since being installed (except for pre-planned maintenance shut downs) and so far as the Warrantors are aware, the Computer System has the capacity and performance features necessary to fulfil the requirements of the Group at Completion and is not in need of renewal or replacement in the immediate future to satisfy the requirements of the business of the Group as currently carried on.



- 11.5 No Group Company has received a notice from or been subject to enquiries by the Data Protection Registrar or Commissioner regarding non-compliance or alleged non-compliance by the relevant Group Company with any provision of the Data Protection Legislation (including, without limitation, the data protection principles set out in the Data Protection Act 1998).
- 11.6 No individual has alleged that any Group Company has failed to comply with the provisions of the Data Protection Legislation or claimed compensation from any Group Company under the Data Protection Legislation, including for unauthorised disclosure of personal data.

### 12. EMPLOYEES

- 12.1 None of the Employees has given or received notice terminating his employment or will be entitled to give notice as a result of the provisions of this agreement. No Employee is entitled to receive any payment as a consequence of the transaction to be effected by this agreement.
- 12.2 The Data Room contains a schedule of all Employees as at the Completion Date and true and accurate details of:
  - (a) the dates of commencement of employment and notice periods of, and details of all remuneration payable (including, without limitation, incentives, bonuses, profit-sharing arrangements and share option schemes) and benefits provided to all such Employees; and
  - (b) where an Employee has been continuously absent from work (for reason other than permitted holiday, compassionate, maternity or paternity leave) for more than one month, in aggregate, in any period of 12 months, the reason for such absence.

Other than the details of Employees set out in the Data Room there are no other employees employed by any Group Company.

- 12.3 There is not in existence any contract of employment with any Director or Employee (or any contract for services with any individual) which cannot be terminated by the relevant Group Company giving three months' notice or less without giving rise to a claim for damages or compensation (other than a statutory redundancy payment or statutory compensation for unfair dismissal) or which is in suspension or has been terminated but is capable of being revived or enforced or in respect of which the relevant Group Company has a continuing obligation.
- 12.4 No Group Company is involved in any disputes and, so far as the Warrantors are aware, there are no circumstances which may result in any dispute involving any of the officers or employees or former employees of any Group Company.
- 12.5 There is no agreement or arrangement between any Group Company and any of the Employees with respect to the termination of his employment which is not included in the written terms of his employment.
- 12.6 Since the Accounts Date, no change has been made in the terms of employment by any Group Company (other than those required by law) of any of the Employees and no Group Company is obliged to increase and has not made provision to increase the total annual remuneration payable to the Employees.

- 12.7 No Group Company has entered into any recognition agreement with a trade union nor has it done any act which may be construed as recognition.
- 12.8 Within the two years preceding Completion, no Group Company has been a party to a relevant transfer (as defined in the Transfer of Undertakings (Protection of Employment) Regulations 2006).
- 12.9 No amounts due to or in respect of any of the officers or employees or former employees of any Group Company (including PAYE and national insurance and pension contributions) are overdue.
- 12.10 No monies or benefits other than in respect of accrued contractual emoluments or annual bonuses are payable to any of the Employees and, so far as the Warrantors are aware, there is not at present a claim, occurrence or state of affairs which may hereafter give rise to a claim against any Group Company arising out of the employment or termination of employment of any employee or former employee for compensation for loss of office or employment or otherwise and whether under the Employment Rights Act 1996, Equality Act 2010, Race Relations Act 1976, Equal Pay Act 1970, Sex Discrimination Act 1975, Sex Discrimination Act 1986, Disability Discrimination Act 1995, Working Time Regulations 1998, National Minimum Wage Act 1998 and the regulations made under such acts or regulations or any other act or otherwise.
- 12.11 There are no persons engaged as independent contractors whose working patterns would enable them to allege that their proper status is as an employee.

#### 13. CONTRACTS

- 13.1 There are set out in the Data Room true, accurate and complete copies of:
  - (a) the Licence Agreements; and
  - (b) all agreements (or where no written agreement exists, a description of the principal payment terms of such agreement) between any Group Company and those of its customers which in the period comprising the last financial year of the Group have accounted for gross revenue in excess of 10% of the total revenue of the Group,

#### (together "Material Contracts").

- 13.2 So far as the Warrantors are aware, no Group Company is a party to or subject to:
  - (a) any agreement or arrangement which is material in the context of the business of the Group between any Group Company and any third party which the signature or performance of this agreement will contravene or under which the third party will acquire a right of termination or any option solely as a result of the signature or performance of this agreement;
  - (b) any agreement or arrangement entered into by any Group Company otherwise than by way of bargain at arm's length or which was made between the relevant Group Company and any of its suppliers, distributors, agents, customers or employees otherwise than in the ordinary course of business;
  - (c) any agreement or arrangement which is material in the context of the business of the Group and which is known to the Warrantors to be unprofitable (that is to say known to be likely to, or the Warrantors believe it will, result in a loss to the Group on completion of performance) or (other than the Material Contracts) of a long-term nature (that is to say incapable of complete performance in accordance with its terms within 12 months after the date on which it was entered into or undertaken);



- (d) any agreement or arrangement which involves delegation of any power under a power of attorney or authorisation of any person (as agent or otherwise) to bind or commit any Group Company to any obligation;
- (e) any agreement or arrangement which restricts the freedom of a Group Company to carry on its business in any part of the world or to use or exploit any of its assets, in each case in such manner as it may think fit;
- (f) any agreement or arrangement which will result in a loss to a Group Company upon completion of performance or fulfilment of a Group Company's obligations thereunder, including, without limitation, any agreement in respect of which a Group Company has not fulfilled any minimum price, fee or quantity guarantees; or
- (g) any agreement or arrangement which is on unduly or abnormally onerous terms.
- 13.3 Neither the relevant Group Company nor, so far as the Warrantors are aware, any party to a Material Contract is in default of a Material Contract, being a default which is likely to have a material and adverse effect on the financial or trading position of the Group and, so far as the Warrantors are aware, there are no circumstances likely to give rise to such a default.
- 13.4 So far as the Warrantors are aware, no breach of contract, event or omission has occurred in respect of any Material Contract which would entitle any third party to terminate, rescind, avoid or repudiate such Material Contract or to call in any money before the date on which payment thereof would normally or otherwise be due and no Group Company has given or received notice of intention to terminate, rescind, avoid or repudiate any of the Material Contracts.

#### 14. JOINT VENTURES ETC

No Group Company is:

- 14.1 a party to any joint venture, consortium, partnership or profit-sharing arrangement or agreement; or
- 14.2 a member of any partnership, trade association, society or other group whether formal or informal and whether or not having a separate legal identity and no such body has any material influence over the business of the Group as now carried on.

## 15. BORROWINGS

- 15.1 No Group Company has outstanding:
  - (a) any borrowing or indebtedness in the nature of borrowing including any bank overdrafts, liabilities under acceptances (otherwise than in respect of normal trade bills) and acceptance credits other than trade indebtedness arising in the ordinary course of business;
  - (b) any guarantee, indemnity or undertaking (whether or not legally binding) to procure the solvency of any person or any similar obligation; or
  - (c) any Encumbrance or any obligation (including a conditional obligation) to create an Encumbrance.

15.2 No Group Company is owed any sums other than trade debts incurred in the ordinary course of business.

### 16. TRADING

- 16.1 There is not outstanding any liability or claim against any Group Company nor so far as the Warrantors are aware, are there any facts, matters or circumstances which could give rise to a liability or claim against any Group Company in relation to any goods or services for which any Group Company is or may be or become liable or responsible in the course of its business.
- 16.2 No dispute or complaint exists or, so far as the Warrantors are aware, has been threatened between any Group Company and any customer, client, distributor, supplier or consumer as to defects in products provided by the relevant Group Company. Where such disputes or complaints do exist the relevant Group Company's potential losses are fully covered by a valid insurance policy which exists for the Group Company's benefit and nothing has been done or omitted to be done to invalidate such insurance policies.
- 16.3 No Group Company has given any guarantee, indemnity, surety or form of comfort in respect of goods or services supplied or contracted to be supplied by it save for any guarantee implied by law or pursuant to a customer's standard terms of purchase or any Licence Agreement and (save as aforesaid) has not accepted any material liability or obligation in respect of any goods or services that would apply after any such goods or services have been supplied by it.

# 17. LITIGATION, OFFENCES AND COMPLIANCE WITH STATUTES

- 17.1 Otherwise than as claimant in the collection of debts arising in the ordinary course of business (none of which exceed £10,000), no Group Company or, so far as the Warrantors are aware, any person for whose acts or defaults any Group Company may be vicariously liable is claimant, defendant or otherwise a party to any litigation, arbitration or administrative proceedings which are in progress or, so far as the Warrantors are aware, are threatened or pending by or against or concerning any Group Company or any of its assets and no Group Company is being prosecuted for any criminal offence and no governmental or official investigation or inquiry concerning the business or officers of any Group Company or any of its assets is in progress or, so far as the Warrantors are aware, there are no circumstances which are likely to give rise to any such proceedings, investigation or inquiry.
- 17.2 No Group Company nor, so far as the Warrantors are aware, any of its officers, agents or employees (during the course of their duties in relation to the business of the Group) has committed or omitted to do any act or thing (ignoring road traffic offences for which no custodial sentence has been imposed or, so far as the Warrantors are aware, likely to be imposed and parking or congestion charge fines) the commission or omission of which is or could be in contravention of any statutory obligation or any other law of the United Kingdom or any other country giving rise to any fine, penalty, default proceedings or other liability in relation to the business or officers of the Group or any of its assets or any judgment or decision which would materially affect the financial or trading position or prospects of the Group.
- 17.3 No Group Company has done or agreed to do anything as a result of which either any investment or other grant paid to any Group Company is or may be liable to be refunded in whole or in part or any such grant for which application has been made by it will or may not be paid or may be reduced.
- 17.4 To the best of the Warrantors' knowledge, information and belief:

- (a) none of the assets owned by the Group has been acquired with monies representing the proceeds of crime; and
- (b) no Group Company has at any time received monies representing criminal proceeds.
- 17.5 Each Group Company has in place adequate procedures in line with the guidance published by the Secretary of State under section 9 of the Bribery Act 2010 designed to prevent their Associated Persons from undertaking conduct which would constitute an offence under the Bribery Act 2010.

# **18. RESTRICTIVE AGREEMENTS AND COMPETITION**

- 18.1 No Group Company or any Employee in the course of carrying out his duties on behalf of a Group Company has been party to or involved in any agreement, understanding, arrangement, concerted practice or conduct which infringe or have infringed nor are there any circumstances of which the Warrantors are aware which may give rise to the infringement of any of:
  - (a) the Competition Act 1998;
  - (b) the Enterprise Act 2002;
  - (c) Articles 101 and 102 of the Treaty on the Functioning of the European Union; or
  - (d) any other competition or anti-trust legislation or regulations which apply or have applied in the EEA or within any jurisdiction within the EEA or any other jurisdiction in the world.
- 18.2 No Group Company or any Employee in the course of carrying out his duties on behalf of a Group Company has ever been the subject of any investigation, inquiry or complaint by, or threat of the same from, any governmental entity, including the European Commission and the Office of Fair Trading.
- 18.3 There are no agreements in force to which any Group Company is a party the terms of which restrict the freedom of any Group Company to provide and take products, goods and services or to otherwise conduct its trade and business by such means, and from and to such persons, as it may, from time to time, think fit.

#### **19. SUBSIDIARIES**

The Company has not since its incorporation had any subsidiary or subsidiary undertaking apart from the Subsidiary Undertakings and has not been the subsidiary of any other company and the Company is not the legal or beneficial owner of any shares of any other company other than the Subsidiary Undertakings.

### 20. ADMINISTRATION

- 20.1 Every document required by the Companies Act or Companies Act 1985 (whichever was in force at the relevant time) has been duly filed and compliance has been and is being made in all material respects by the Group Companies with the Companies Act or Companies Act 1985 (whichever was in force at the relevant time).
- 20.2 The copy of the memorandum and articles of association of each Group Company set out in the Data Room or annexed to the Disclosure Letter is accurate and complete in all respects, includes copies of all resolutions and documents required to be incorporated therein and fully sets out all rights attaching to each class of the share capital of such Group Company and the register of members and other statutory registers of each Group Company have been properly kept in all material respects and contain a true, accurate and complete record of all the matters which should be dealt with therein and no notice or allegation that any of the same is incorrect or should be rectified has been received by any Group Company.

- 20.3 Each Group Company is validly existing and is entitled to carry on the business now carried on by it.
- 20.4 No Group Company has given any power of attorney or any other authority (express, implied or ostensible) which is still outstanding or effective to any person to enter into any contract or commitment or do anything on its behalf (other than any authority of Directors or Employees to enter into routine trading contracts in the normal course of their duties or bank mandates).

#### 21. INSOLVENCY

- 21.1 No resolution has been passed nor meeting called to consider such resolution, no petition has been presented and no order has been made for the winding up of or for the appointment of a provisional liquidator to any Group Company (or for any equivalent procedure in any foreign jurisdiction).
- 21.2 No petition has been presented and no application has been made to court for an administration order in respect of any Group Company and no notice of an intention to appoint an administrator of any Group Company has been given or filed (or for any equivalent procedure in any foreign jurisdiction).
- 21.3 No liquidator, administrator, receiver, receiver and manager, administrative receiver or similar officer (or any equivalent officer in any foreign jurisdiction) has been appointed in relation to any Group Company or in relation to the whole or any part of its assets, rights or revenues.
- 21.4 In relation to each Group Company:
  - (a) no voluntary arrangement (or any equivalent procedure in any foreign jurisdiction) has been proposed or implemented under section 1 of the Insolvency Act 1986;
  - (b) no scheme of arrangement (or any equivalent procedure in any foreign jurisdiction) has been proposed or implemented under Part 26 of the Companies Act;
  - (c) no scheme for the benefit of creditors has been proposed or implemented, whether or not under the protection of the court and whether or not involving a reorganisation or rescheduling of debt; and
  - (d) no proceedings have been commenced under any law, regulation or procedure relating to the reconstruction or adjustment of debts.
- 21.5 No Group Company has stopped or suspended payment of its debts, and no Group Company is unable to pay its debts within the meaning of section 123(1) of the Insolvency Act 1986.
- 21.6 As far as the Warrantors are aware, none of the assets of any Group Company have been the subject of any transaction that may be challenged by a liquidator or administrator under the Insolvency Act 1986 (or any equivalent procedure in any foreign jurisdiction).



- 21.7 No distress, execution or other process has been levied on an asset of any Group Company and no unsatisfied judgment, order or award is outstanding against any Group Company.
- 21.8 No action has been or is being taken by the Registrar of Companies (or any equivalent authority in any foreign jurisdiction) to strike any Group Company off the register under section 1000 of the Companies Act and no application for the same has been made by any Group Company under section 1003 of the Companies Act (or any equivalent legislation in any foreign jurisdiction).
- 21.9 None of the Warrantors are bankrupt and as far as the Warrantors are aware no petition has been presented to make any of them bankrupt.
- 21.1 None of the Warrantors is aware of any circumstances which are likely to give rise to any of the matters set out in this paragraph 21.

### 22. PROPERTY MATTERS

- 22.1 The Property comprises:
  - (a) all the land and premises of whatever tenure owned, occupied or otherwise used by the Group whether in the United Kingdom or elsewhere; and
  - (b) all the estate, interest, right and title whatsoever (including for the avoidance of any doubt interests in the nature of options, rights of pre-emption or other contractual relationships) of the Group in respect of any land or premises.
- 22.2 Copies of all material documents relating to the title to the Property are contained in the Data Room. The title deeds to the Property are in the Group's possession and the title deeds to be delivered to the Buyer on Completion will consist of original documents or properly marked and examined abstracts.

#### 23. ENCUMBRANCES RELATING TO THE PROPERTY

- 23.1 The Property is not subject to any liability for the payment of any outgoings, other than national non-domestic rates, water and sewerage services charges and insurance premiums and rents and service charges.
- 23.2 The Group's interest in the Property is not subject to any agreement or right to acquire the same nor any option, right of pre-emption or right of first refusal and there is no intention to enter into any such agreement or arrangement.
- 23.3 The Group's interest in the Property is not subject to any mortgage, or charge created by any member of the Group whether securing the repayment of monies or otherwise.

#### 24. STATUTORY OBLIGATIONS RELATING TO THE PROPERTY

24.1 The Group has not received notice of any breach of any applicable statutory and bye-law requirements with respect to the Property.

## 25. PLANNING MATTERS

- 25.1 So far as the Warrantors are aware, no Group Company has received notice of breach of any planning permission, building regulation or bye law consents for the time being in force in relation to the Property.
- 25.2 So far as the Warrantors are aware, no Group Company has received notice of any planning contravention notices, breach of condition notices, enforcement notices or stop notices which have been issued by any local planning authority in respect of the Property.

### 26. GENERAL MATTERS RELATING TO THE PROPERTY

- 26.1 No Group Company has any continuing liability in respect of any other property formerly owned or occupied by any Group Company either as original contracting party or by virtue of any direct covenant having been given on a sale or assignment to any Group Company or as a guarantor of the obligations of any other person in relation to such property or otherwise.
- 26.2 The Property has access to and is accessible from a public highway or has rights to pass over any land between the Property and the public highway.

### 27. PENSIONS

- 27.1 There is not and no proposal has been announced to enter into or establish and no Group Company contributes, is bound to contribute either now or in the future, any retirement, disability or death benefits, for or in respect of any of the Employees.
- 27.2 Section 3 of the Pensions Act 2008 does not apply to any Group Company at the Completion Date.

#### 28. EFFECT OF COMPLETION

- 28.1 No consent, approval, authorisation or order of any court or government or local agency or body (except in relation to any competition or anti-trust legislation or regulation) is required by the Sellers for the execution or implementation of this Agreement and the Transaction Documents and compliance with the terms of this agreement and the Transaction Documents does not and is not reasonably likely to:
  - (a) conflict with, result in the breach of or constitute a default under any agreement, instrument or obligation by which any Group Company may be bound or any provision of the articles of association (or equivalent constitutional document) of any Group Company; or
  - (b) result in the creation, imposition, crystallisation or enforcement of any Encumbrance on, over or affecting any of the assets of any Group Company.

# 29. FACTUAL INFORMATION

All factual information contained in Documents 21.1.2 and 21.1.3 of the Data Room (the "**Schedules**") is accurate in all material respects and not misleading and any opinions expressed therein are the opinion of the Warrantors and in each case are honestly and reasonably held. Except in so far as the information contained in the Schedules is anonymised in the first column of each table, it directly reflects the status of customers of the Group

#### Part 2

#### **Tax Warranties**

#### 1. Accounts and Liability for Taxation

- 1.1 The Accounts make proper provision for all Taxation for which each Group Company was liable in respect of all periods up to the Accounts Date, and for deferred Taxation, in accordance with UK GAAP.
- 1.2 Each Group Company has maintained all records in relation to Taxation as are required by law.
- 1.3 All Taxation for which each Group Company has been liable to account has been duly paid to the relevant Taxation Authority and each Group Company has made all such deductions and retentions from payments made by the relevant Group Company, and accounted for such deductions and retentions, as was required by law.
- 1.4 No Group Company has within the past three years paid or become liable to pay, and so far as the Warrantors are aware there are no circumstances by reason of which any Group Company is likely to become liable to pay, any penalty, fine, surcharge or interest relating to Taxation.

#### 2. Returns, Information and Clearances

- 2.1 All returns, notices, accounts, statements, computations, information, assessments and registrations required by law to have been made or provided by each Group Company for any Taxation purpose have been made or provided within applicable time limits and on a proper basis and were at the time and remain accurate and complete in all material respects.
- 2.2 The Disclosure Letter contains details of all transactions, schemes or arrangements in respect of which any Group Company has been a party or has otherwise been involved for which a statutory clearance application was made within the last three years.

# 3. Distributions

3.1 No distribution or deemed distribution, within the meaning of section 1000 of the CTA 2010, has been made (or will be deemed to have been made) by any Group Company within the last three years, except dividends and reductions of share capital shown in their audited accounts nor is any Group Company bound to make any such distribution.

### 4. Intangible assets

- 4.1 Since the Accounts Date:
  - (a) no Group Company owns an asset which has ceased to be a chargeable intangible asset in the circumstances described in section 859 of the CTA 2009;
  - (b) no Group Company has realised or acquired an intangible fixed asset for the purposes of Part 8 of the CTA 2009; and
  - (c) no circumstances have arisen which have required, or will require, a credit to be brought into account by any Group Company on a revaluation of an intangible fixed asset.



#### 5. Close companies

- 5.1 No Group Company is a close investment-holding company as defined in section 34 of the CTA 2010.
- 5.2 No distribution within section 1064 of the CTA 2010 has been made by a Group Company within the last three years.
- 5.3 No loan or advance made by or assigned to the Company falling within the provisions of sections 455, 459, and 460 of the CTA is outstanding or has been waived since the Accounts Date.

#### 6. Groups of companies

- 6.1 No Group Company has been a member of a group of companies or a fiscal unity for any Taxation purpose other than a group comprised solely of the other Group Companies.
- 6.2 All transactions between Group Companies have been entered into on arm's length terms.

## 7. Company residence and Overseas interests

- 7.1 Each Group Company has been resident at all times since its incorporation solely in the jurisdiction of its incorporation and is not and has never been treated for any Taxation purpose as resident (or dual-resident) in any other jurisdiction(s).
- 7.2 No Group Company has at any time since incorporation had a branch, agency or permanent establishment outside the jurisdiction of its incorporation.

### 8. Anti-avoidance

No Group Company has been a party to, or been involved in, any schemes or arrangements designed wholly or partly for the purposes of avoiding or deferring any Taxation liability, or in relation to which any disclosure has been, or will be, required to be made to any Taxation Authority.

### 9. Stamp duty, stamp duty land tax and stamp duty reserve tax

- 9.1 All stampable documents to which any Group Company is a party and which are required by a Group Company to prove title of any Group Company to any material asset have been duly stamped.
- 9.2 Each Group Company has paid all stamp duty land tax chargeable and has filed all land transaction returns required in connection with any land transaction (as defined under section 43 of the Finance Act 2003) entered into by any Group Company on or before the date of this agreement.
- 9.3 Neither entering into this agreement nor Completion will result in the withdrawal of any stamp duty or stamp duty land tax relief granted on or before completion that will affect any Group Company.
- 9.4 There is no chargeable interest (as defined under section 48 of the Finance Act 2003) acquired or held by any Group Company in respect of which an additional land transaction return will be required to be filed with a Taxation Authority and/or a payment of stamp duty land tax made on or after the date of this agreement.



#### 10. Value added tax

- 10.1 Each Group Company is duly registered for VAT purposes.
- 10.2 Each Group Company has complied with all of its obligations in respect of VAT.
- 10.3 No Group Company has an interest in any land and buildings in relation to which an option to tax has been made, either by the relevant Group Company or by any other person, under the provisions of Schedule 10 to the VATA 1994.
- 10.4 The provisions of Part XV Value Added Tax Regulations 1995 (the Capital Goods Scheme) do not apply to any assets of any Group Company.

### 11. Inheritance Tax

- 11.1 No Group Company is liable and there are no circumstances in existence as a result of which any Group Company may become liable, to be assessed to inheritance tax or any other Taxation as doner or donee of any gift, or transferor or transferee of value and there are no other circumstances by reason of which any liability in respect of inheritance tax has arisen or could arise to any Group Company.
- 11.2 There are no circumstances under which any power within section 212 of the IHTA 1984 could be exercised in relation to, and there is no HM Revenue & Customs charge within the meaning of section 237 of the IHTA 1984 attaching to or over, any shares or securities in or assets of any Group Company and there are no circumstances which could lead to any such charge arising in the future.

#### **SCHEDULE 5**

#### **Tax Covenant**

#### 1. **Definitions and interpretation**

1.1 In this schedule 5 the following words and expressions shall (except where the context otherwise requires) have the following meanings:

"Event" means an event, transaction, action or omission whether alone or in conjunction with any other transaction, action or omission and includes further (without limitation) the death of any person or a Group Company becoming, being or ceasing to be a member of a group of companies for the purposes of any Tax;

"Income, Profits or Gains" means revenue profits, chargeable gains and any other similar measure by reference to which Tax is chargeable or assessed;

#### "Post Completion Relief" means:

- (a) a Relief in respect of an Event occurring, or an accounting period commencing, after Completion; and
- (b) a Relief in respect of an Event occurring in the ordinary course of business of the Group Companies between the Locked Box Date and Completion.

""**Relevant Claim**" means any claim, counterclaim, assessment, notice, letter, determination, demand, return, account or other document issued or action taken by or on behalf of any Tax Authority, and/or any return or computation from which it appears that a Tax Liability has been, or may be, imposed on a Group Company for which the Warrantors may be liable under this schedule, or in respect of which the Warrantors may be liable for a breach of the Tax Warranties;

"Relief" means any loss, allowance, credit, deduction or set-off or any right to repayment of Taxation;

"**Saving**" means the reduction or elimination of any liability of a Group Company to make an actual payment of Tax in respect of which the Warrantors would not have been liable under paragraph 2, by the use of any Relief arising wholly as a result of a Tax Liability in respect of which the Warrantors have made payment under this schedule; [DN A "saving" for a claim for breach of a Tax Warranty would to be taken into account in mitigation and quantum].

"**Tax**" or "**Taxation**" means any form of tax or any levy, impost, duty, contribution or withholding in the nature of tax imposed, collected or assessed by, or payable to, a Tax Authority and all penalties and interest included in or relating to any of the above (in all cases, regardless of whether such taxes, penalties and interest are directly or primarily chargeable against or attributable to a Group Company or any other person and regardless of whether a Group Company has, or may have, any right of reimbursement against any other person);

"Tax Authority" means any government, state or municipality or any local, state, federal or other fiscal, revenue, customs or excise authority, body or official in the United Kingdom or elsewhere;

"Tax Liability" means a liability of a Group Company to make actual payments of, or in respect of, Tax (an Actual Tax Liability) but also the setting off against any Tax Liability or against Income, Profits or Gains earned, accrued or received on or before Completion of any Post Completion Relief, in circumstances where, but for such setting off, a relevant Group Company would have had an Actual Tax Liability in respect of which the Buyer would have been able to make a claim against the Warrantors under this schedule and so that the amount of the Tax Liability shall be the amount of Tax which has been saved in consequence of such set off; and



"Tax Refund" means a right to repayment of Tax or an actual repayment of Tax to which a Group Company becomes entitled or receives in respect of a period (or part period) prior to Completion or as a result of an Event occurring prior to Completion.

- 1.2 In this schedule:
  - (a) references to any **Income**, **Profits or Gains** earned, accrued or received before a particular date or in respect of a particular period include Income, Profits or Gains which are deemed for the purposes of any Tax to have been earned, accrued or received at or before the date or in respect of that period;
  - (b) Completion shall, for the purposes of determining the amount of any Tax Liability, be deemed to be the end of an accounting period or a period of account (as the case may be) for all Taxation purposes.

### 2. Covenant

- 2.1 The Warrantors jointly and severally covenant to pay to the Buyer an amount equal to:
  - (a) any Tax Liability of a Group Company which arises:
    - (i) in consequence of any Event which occurred on or before Completion; or
    - (ii) in respect of any Income, Profits or Gains which were earned, accrued or received on or before Completion; and
  - (b) any Tax Liability of a Group Company which arises in consequence of:
    - (i) the failure by any other person ("Primary Person") to pay Tax; and
    - (ii) of that Group Company having at any time prior to Completion:
      - (A) been a member of the same group of companies as the Primary Person; or
      - (B) had control of, or been otherwise connected with, the Primary Person for any Tax purpose; or
      - (C) been under the control of the Primary Person (whether alone or together with any other person or persons) or a person connected with the Primary Person for any Tax purpose;
  - (c) any reasonable costs and expenses reasonably and properly incurred by the Buyer and/or the relevant Group Company in connection with any successful claim made by the Buyer under this schedule.

## 3. Exclusions

3.1 The covenant at paragraph 2 does not apply in respect of any Tax Liability (and the Warrantors shall not be liable for any breach of the Tax Warranties arising as a result of a Tax Liability, or where damages for such breach are calculated in respect of or by reference to a Tax Liability) to the extent that:

- (a) provision or reserve in respect of the Tax Liability is made in the Management Accounts;
- (b) payment or discharge of the Tax Liability has been made prior to Completion and such payment or discharge has been reflected in the Management Accounts;
- (c) the Tax Liability arises as a result of an Event in the ordinary course of business of a Group Company after the Locked Box Date but on or before Completion;
- (d) the Tax Liability arises or is increased as a result of:
  - (i) a change in Tax rates or in legislation (including any subordinate legislation) made after the Locked Box Date or a change in the judicial interpretation of the law; or
  - (ii) a change or withdrawal after the Locked Box Date of any published practice, concession or interpretation of any Tax Authority;

in each case with retrospective effect;

- (e) the Tax Liability would not have arisen but for a voluntary act or transaction carried out by the Buyer or a Group Company (including any act or omission by any director, officer, employee, agent, representative or adviser of or to the Buyer or a Group Company) after Completion other than: (i) pursuant to any applicable legal or regulatory obligation; (ii) pursuant to any legally binding obligation entered by the relevant Group Company into before Completion; or (iii) at the written request of the Sellers;
- (f) the Tax Liability arises or is increased in consequence of any failure by the Buyer to comply with, or a failure to procure the compliance of a Group Company with, any of their respective obligations under this schedule;
- (g) the Tax Liability results from or is increased or extended by any change on or after Completion in:
  - (i) the accounting reference date of a Group Company; or
  - (ii) any change in the accounting policies, bases or practice of a Group Company or Tax reporting practices of a Group Company other than to comply with any accounting policies, bases or practices applicable to the Group Company which were in force at Completion;
- (h) a Relief (other than a Post-Completion Relief) is available, or is for no consideration made available, to the Group Company to set against or otherwise reduce the Tax Liability,
- (i) the Tax Liability would not have arisen or would have been reduced or eliminated but for:
  - (i) the making of a claim, election, surrender or disclaimer, the giving of a notice or consent, or the doing of any other thing under the provisions of any enactment or regulation relating to Tax, in each case after Completion by the Buyer or any Group Company, other than as required by law, as provided in the Accounts or the Management Accounts or at the Warrantors' written request; or

- (ii) the failure or omission on the part of any Group Company (other than at the Warrantors' written direction) to make any such valid claim, election, surrender or disclaimer, or to give any such notice or consent or to do any other such thing, the making or doing of which was taken into account in computing the provision or reserve for Taxation in the Management Accounts or as the Warrantors may require in respect of periods or matters for which they have conduct under paragraph 7 where, in each case, the Buyer has given written notice of the requirement to make the claim, election, surrender or disclaimer;
- (j) such Tax Liability is in respect of stamp duty or stamp duty reserve tax payable on the transfer or agreement to transfer the Sale Shares;
- (k) the Tax Liability would not have arisen but for a cessation of trade by, or a change in the nature or conduct of the trade of a Group Company on or after Completion;
- (1) the Income, Profits or Gains in respect of which that Tax Liability arises were actually earned, accrued or received by a Group Company but were not reflected in the Management Accounts;
- (m) the Buyer or a Group Company makes or has made recovery in respect of that Tax Liability under any provision of the Agreement;
- (n) the Tax Liability arises or is increased as a result of any delay or default by the Buyer or a Group Company in paying over to any Tax Authority any amount received from the Warrantors under this schedule or in respect of a breach of the Tax Warranties;
- (o) an amount in respect of the Tax Liability has already been recovered by the Buyer or a Group Company from another person; or
- (p) the Tax Liability has been made good by insurers or otherwise compensated for without cost to the Buyer or any Group Company.
- 3.2 The limitations in clause 6 of this agreement shall not apply to any claim under paragraph 2.1(b) of this schedule 5 to the extent that the Primary Person is a Warrantor and in which case the liability of the Warrantors under paragraph 2.1(b) shall be several.

#### 4. Savings, over-provisions and tax refunds

- 4.1 If the Buyer or a Group Company becomes aware that:
  - (a) a Saving has arisen; or
  - (b) any provision for Tax contained in the Management Accounts is likely to be an over-provision (other than to the extent the over-provision would arise or be increased as a result of any retrospective change in the law after the Locked Box Date or the utilisation or set off of any Post Completion Relief) (an "**Over-provision**"); or
  - (c) any Tax Refund has arisen,

the Buyer shall promptly give details of such Saving, Over-provision or Tax Refund by written notice to the Warrantors and the Buyer shall provide prompt replies to reasonable enquiries by the Warrantors and their agents and provide reasonable assistance and information in relation to these matters.

- 4.2 The Warrantors may at any time instruct the relevant Group Company's auditors to determine in writing the extent of any Saving, Over-provision or Tax Refund. If the auditors determine that a Saving, Over-provision or Tax Refund has arisen, an amount equal to the value (as so determined in writing) of such Saving, Over-provision or Tax Refund less any reasonable costs incurred by the relevant Group Company in obtaining it and less any Tax payable in respect of it (or which would have been payable but for the use of any Post Completion Relief):
  - (a) shall be set off against any payment then due from the Warrantors to the Buyer under this schedule or for breach of the Tax Warranties; and
  - (b) to the extent there is any excess:
    - (i) in the case of a Tax Refund, the excess shall be promptly repaid to the Warrantors;
    - (ii) in the case of a Saving or an Over-provision, the excess shall be set against any payment(s) already made or subsequently due under this schedule or for breach of the Tax Warranties in chronological order until exhausted, provided that to the extent that such Saving or Over-provision is set-off against any payment already made by the Warrantors, it shall promptly be repaid by the Buyer to the Warrantors.
- 4.3 For the purposes of this paragraph 4, the Buyer shall procure that the auditors, acting as experts and not as arbitrators, determine in writing whether any Saving, Over-provision or Tax Refund has arisen.

## 5. Recovery from third parties

- 5.1 If a Group Company is entitled to recover from any person (including, for the avoidance of doubt, any Tax Authority) any sum in respect of any Tax Liability to which this schedule applies or which is the subject matter of a claim under the Tax Warranties or becomes entitled at some subsequent date to make such a recovery, the Buyer shall procure that the relevant Group Company shall:
  - (a) promptly notify the Warrantors of all relevant details concerning such entitlement after as soon as is reasonably practicable after such Group Company becomes aware of such entitlement;
  - (b) take all appropriate steps to enforce recovery under such entitlement (if so required by the Warrantors; and
  - (c) promptly pay to the Warrantors the sum equal to the lesser of:
    - (i) any amount so recovered (together with an amount equal to any interest payment (after Tax) or repayment supplement received by a Group Company in connection with the recovery) less any reasonable costs incurred by the Buyer or the relevant Group Company in obtaining it and less any Tax payable in respect of it (or which would have been payable but for the use of any Post Completion Relief); and
    - (ii) the amount payable by the Warrantors in respect of such Tax Liability under this schedule or for breach of the Tax Warranties together with any interest (after Tax) and repayment supplement paid by the Tax Authority or other person in respect of the sum recovered less any Tax payable in respect of it (or which would have been payable but for the use of any Post Completion Relief save to the extent such Tax has already been made good by the Warrantors under paragraph 5.1(c)(i) above).



#### 6. Buyer's covenant in respect of secondary liabilities

- 6.1 The Buyer covenants to pay to the Warrantors an amount equal to any Tax assessed on the Warrantors to the extent that such Tax is not of a type for which the Warrantors are liable under this schedule or the Tax Warranties; and:
  - (a) arises as a result of the failure of a member of the Buyer's Group to apply an amount provided for in the Management Accounts or an amount paid by the Warrantors to the Buyer under this schedule to discharge any liability to which that amount relates; or
  - (b) relates to any Tax of a Group Company or other member of the Buyer's Group which arises in respect of any Event, period or part period after Completion.
- 6.2 The Buyer covenants to pay to the Warrantors any reasonable costs and expenses properly incurred in connection with any successful claim made under this paragraph.
- 6.3 Paragraphs 8 and 9 of this schedule apply to the covenant at paragraph 6.1 as they apply to the covenants in paragraph 2 with such modifications as are necessary to give effect to those provisions.

#### 7. Conduct of pre-completion tax affairs

- 7.1 The Warrantors shall, at the Company's cost and expense to the extent provided for in the Management Accounts and at the Warrantors' cost and expense thereafter, have the responsibility for, and the conduct of, preparing, submitting, negotiating and agreeing with the relevant Tax Authority all outstanding Tax computation and returns of the Group Companies for each accounting period ending on or before Completion ("**Relevant Accounting Periods**") provided that the Warrantors shall:
  - (a) keep the Buyer fully and promptly informed of all material matters relating to such outstanding Tax computation and returns of the Group Companies and deliver to the Buyer copies of all material correspondence with a Tax Authority relating to such computation and returns;
  - (b) submit to the Buyer for comments all material correspondence and documents (including, for the avoidance of doubt, tax returns) which it intends to submit to a Tax Authority and take into account all such reasonable and timely comments as the Buyer may make;
  - (c) not submit to a Tax Authority any such correspondence or documents, or agree any matter material for the potential liability of the Warrantors under this schedule in relation to Relevant Accounting Periods without the prior written approval of the Buyer, such approval not to be unreasonably withheld or delayed;
  - (d) obtain the prior written approval of the Buyer (not to be unreasonably withheld or delayed) to the appointment of any agent to deal on their behalf with any Tax Authority.
- 7.2 The Buyer shall procure that the Group Companies shall:

67

- (a) sign and submit any Tax return and computations relating to the Relevant Accounting Periods delivered to it in accordance with this paragraph 7 without delay, provided that the Buyer shall be under no obligation to procure the signing or submission of any document delivered to it which is false or misleading (but for the avoidance of doubt shall not be under any obligation to make enquiry as to the completeness or accuracy of an document and shall be entitled to rely on the Warrantors and their agents);
- (b) make such claims (including, but not limited to, claiming or disclaiming capital allowances, depreciation allowances and claiming roll-over reliefs);
- (c) execute such documents (including but not limited to returns of profits and losses and amended returns); and
- (d) do such other things in relation to the relevant accounting periods as are reasonably considered necessary by the Warrantors;
- (e) in each case, as directed in writing by the Warrantors and shall further procure that the Group Companies shall not (unless so directed by the Warrantors):
  - (i) amend, withdraw or disclaim any election or claim previously made; nor
  - (ii) disclaim any allowance or Relief.
- 7.3 The Buyer shall be responsible for the preparation and submission of the Tax computation and returns of the Group Companies for the accounting period in which Completion takes place. The Buyer shall prepare such Tax computations and returns on a basis consistent with that adopted in previous periods and that adopted when computing the provision for Tax contained in the Management Accounts. Draft Tax computations and returns shall be submitted to the Warrantors for comment and approval (such approval not to be unreasonably withheld) before they are submitted to the relevant Tax Authority and the Warrantors will have a period of 20 Business Days to comment on such Tax computations and returns and computations to the relevant Tax Authority.
- 7.4 The Warrantors and the Buyer shall procure the provision to each other of such information and assistance as each may reasonably require of the other to prepare, submit, and agree all Tax computations, documents or correspondence for the purposes of complying with their respective obligations under this paragraph.

#### 8. Conduct of Tax Claims

- 8.1 If the Buyer or a Group Company becomes aware of any Relevant Claim, the Buyer shall give or procure that written notice of that Relevant Claim is given to the Warrantors as soon as reasonably practicable thereafter (but so that such notice shall not be a condition precedent to the liability of the Warrantors under this schedule).
- 8.2 The Buyer shall or shall procure that the relevant Group Company shall take such action as the Warrantors may reasonably request in writing to resist, avoid, dispute, appeal, compromise or defend the Relevant Claim concerned provided that the Warrantors shall first indemnify the Buyer (and, if relevant, any other member of the Buyer's Group) against all reasonable costs and expenses and liabilities (including additional Tax and any interest, fines, surcharges and penalties) which may thereby be incurred as a result of taking such action.
- 8.3 If the Warrantors fail within 15 Business Days of being given notice under paragraph 8.1 to request the Buyer to take any action (or procure that such action is taken by a Group Company) pursuant to paragraphs 8.2 in connection with the Relevant Claim concerned, then the Buyer shall (without prejudice to the rights of the Buyer under this Agreement) be free to pay or settle such Relevant Claim on such terms as it shall, acting reasonably, decide.



- 8.4 The Buyer shall, and shall procure that the relevant Group Company shall, keep the Warrantors informed of the progress of any action taken in relation to the Relevant Claim and shall, within 2 Business Days of receipt provide the Warrantors with copies of all material correspondence concerning the Relevant Claim and the action taken in respect of such Relevant Claim.
- 8.5 The Buyer shall not be obliged to take or procure the taking of any action pursuant to paragraph 8.2:
  - (a) which involves contesting any Relevant Claim before any court or other appellate body (excluding the Tax Authority where the amount of Tax in dispute exceeds £100,000 in question) unless, at the Warrantors' expense, the Warrantors have first obtained the written opinion of tax counsel of at least 7 years' call, and after disclosure to him of all relevant information and documents, that the proposed appeal has a reasonable prospect of success; or
  - (b) if, in the reasonable opinion of the Buyer, the Warrantors or (before Completion) the relevant Group Company has committed an act or an omission which constitutes fraudulent conduct in relation to the Relevant Claim in question.
- 8.6 The Warrantors shall have the right to have the conduct of any dispute arising in connection with a Relevant Claim delegated to them or their nominated professional advisors, provided that, unless the Buyer and the Warrantors specifically agree otherwise in writing, the following provisions shall apply:
  - (a) the Buyer shall be kept informed of all matters pertaining to the Relevant Claim and shall be entitled to see and keep copies of all material correspondence and notes or other written records of telephone conversations or meetings;
  - (b) the appointment of solicitors or other professional advisors to act on behalf of the relevant Group Company shall be subject to the approval of the Buyer, such approval not to be unreasonably withheld or delayed; and
  - (c) no material communication, pertaining to the Relevant Claim shall be transmitted to the relevant Tax Authority without the same having been submitted to the Buyer for its comments, and the Warrantors shall take into account all reasonable comments which are communicated in writing to the Warrantors.

#### 9. Due date for payment

- 9.1 The Warrantors are to pay to the Buyer (in cleared funds) any required sum under paragraph 2:
  - (a) in respect of an Actual Tax Liability on the later of:
    - (i) the date 5 Business Days before the date on which the amount in question is payable to the relevant Tax Authority without any interest, penalty, fine or surcharge arising in respect of it; and

- (ii) the date 5 Business Days following the date on which notice giving written details of the amount due is received by the Warrantors from the Buyer;
- (b) in respect of any Tax Liability which is not an Actual Tax Liability, on the date 5 Business Days after the date on which the Warrantors receive written details of the amount of the liability from the Buyer or, if later, the date on which the relevant Group Company would have had to pay the Tax but for the utilisation of a Post Completion Relief;
- (c) in respect of any required sum under paragraph 2 which is not a Tax Liability, on the date 5 Business Days following the date on which notice giving written details of the amount due is received by the Warrantors from the Buyer.

#### **SCHEDULE 6**

#### **Permitted Leakage**

The following payments made or to be made by or on behalf of any Group Company in the ordinary and usual course of business:

- 1. remuneration, agreed bonuses in aggregate of £519,845 as set out in the schedule in the agreed form or other entitlements properly payable to employees, directors, consultants or officers of any Group Company pursuant to their terms of employment, service or consultancy agreements (as in force as the date of this agreement) including all associated PAYE, National Insurance and VAT obligations and, in addition, any reimbursements of expenses in the ordinary course of business; and
- 2. an amount up to but not exceeding £1,000 (plus any applicable VAT) payable to the Institutional Sellers (in aggregate) representing cash expenses incurred by the Institutional Sellers or any director appointed by them.

## 71

## **SCHEDULE 7**

## Property

# 1. Freehold

F/H Unit 1 Haig Road, Parkgate Industrial Estate, Knutsford, Cheshire, T/NO CH205013

## 2. Leasehold

2.1 Conde de Penalver 38-5, 28006 Madrid, Spain

- 2.2 No 2 Kallang Pudding Road, 06-10 Machtech Building, Singapore 348307
- 2.3 21062 Bake Parkway, Suite 100,200 and 20992 Bake Parkway Suite 104, Lake Forest, 92630, USA
- 2.4 21 Wood Lands Drive, Building 2, Country Club Estate, Woodmead, 2052, South Africa

# 72

The parties have executed and delivered this agreement as a deed on the date and year first written above.

1 0	5
SIGNED by	)
STEVEN GOSLING	) /s/ Steven Gosling
in the presence of:	)
Tim Sydor	
10 Queen Street Place	
London EC4R 1BE	
EC4K IBE	
SIGNED by	)
ANDREW MCEWAN	) /s/ Andrew McEwan
in the presence of:	)
Tim Sydor	
10 Queen Street Place	
London EC4R 1BE	
EC4K IBE	
SIGNED by	)
ESTIBALIZ ASIAIN SANCHO	) /s/ Estibaliz Asiain Sancho
in the presence of:	)
Tim Sydor	
10 Queen Street Place	
London EC4R 1BE	
EC4K IDE	
EXECUTED and delivered as a deed by	)
GCP CAPITAL PARTNERS LLP	)
in its capacity as manager of	) /s/ Adam Maidment
GREENHILL CAPITAL PARTNERS	)
EUROPE L.P. acting by its duly authorised	)
attorney in the presence of:	)
/s/ Caitlin DuPuy	
Witness signature	
Name: Caitlin DuPuy	
Address: 5 Newstreet Square London EC4A 3TW	
<b>EXECUTED</b> and delivered as a deed by	)
GCP CAPITAL PARTNERS LLP	)
in its capacity as manager of	) /s/ Adam Maidment
GREENHILL CAPITAL PARTNERS	)
EUROPE (EMPLOYEES) L.P. acting by its	)
duly authorised attorney in the presence of:	)
/s/ Caitlin DuPuy	
Witness signature	
Name: Caitlin DuPuy	
Address: 5 Newstreet Square London EC4A 3TW	

EXECUTED and delivered as a deed by GCP CAPITAL PARTNERS LLP in its capacity as manager of PARALLEL PRIVATE EQUITY PARTNERS GREENHILL L.P. acting by its duly authorised attorney in the presence of: /s/ Caitlin DuPuy	) ) ) /s/ Adam Maidment ) )
Witness signature	
Name: Caitlin DuPuy	
Address: 5 Newstreet Square London EC4A 3TW	
<b>EXECUTED</b> and delivered as a deed by <b>IFES ACQUISITION CORP LIMITED</b> acting by a director in the presence of:	) ) /s/ Jay Itzkowitz )
/s/ Marco Toracca	
Witness signature	
Name: Marco Toracca	
Address: 10 Queen Street Place London EC4R 1BE	
	74

# McDermott Will & Emery

Boston Brussels Chicago Düsseldorf Frankfurt Houston London Los Angeles Miami Milan Munich New York Orange County Paris Rome Seoul Silicon Valley Washington, D.C. Strategic alliance with MWE China Law Offices (Shanghai)

October 21, 2013

Global Eagle Entertainment Inc. 4353 Park Terrace Drive Westlake Village, California 91361 Ladies and Gentlemen:

We have acted as counsel to Global Eagle Entertainment Inc., a Delaware corporation (the "Company"), in connection with the offering by the Company of 2,453,472 shares (the "Shares") of the Company's common stock, \$0.0001 par value per share, pursuant to the Registration Statement on Form S-3, as amended (Registration No. 333-188121), filed on August 13, 2013 with the United States Securities and Exchange Commission (the "Commission") pursuant to the Securities Act of 1933, as amended (the "Securities Act"), which was declared effective by the Commission on August 22, 2013 (as amended or supplemented, the "Registration Statement"), the related prospectus dated August 22, 2013 (the "Base Prospectus"), and the prospectus supplement relating to the offering of the Shares dated the date hereof (the "Prospectus Supplement" and together with the Base Prospectus, the "Prospectus") which will be filed by the Company with the Commission pursuant to Rule 424(b)(5) under the Securities Act. The offering of the Shares is being made pursuant to a Common Stock Purchase Agreement, dated October 18, 2013, between the Company and Putnam Equity Spectrum Fund (the "Purchase Agreement"). We are rendering this opinion to you at your request to enable you to fulfill the requirements of Item 601(b)(5) of Regulation S-K, 17 C.F.R. § 229.601(b)(5), in connection with the Registration Statement and Prospectus.

We have examined the originals, or duplicates or certified or conformed copies, of the Registration Statement, the Prospectus, the Purchase Agreement and such other corporate and other records, agreements, documents and other instruments and have made such other investigations as we have deemed relevant and necessary in connection with the opinion hereinafter set forth. As to questions of fact material to this opinion, we have relied upon certificates or comparable documents of public officials and of officers and representatives of the Company.

In rendering the opinion set forth below, we have assumed the genuineness of all signatures, the legal capacity of natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as duplicates or certified or conformed copies and the authenticity of the originals of such latter documents.

The opinion set forth below is limited to the Delaware General Corporation Law.

Based upon the foregoing, we are of the opinion that the Shares have been duly authorized and, when the Shares have been issued and delivered in the manner contemplated by the Registration Statement and the Prospectus and pursuant to the terms of the Purchase Agreement, and upon receipt by the Company of the consideration therefor, the Shares will be validly issued, fully paid and non-assessable.

We hereby consent to the filing of this opinion letter as Exhibit 5.1 to the Current Report on Form 8-K to be filed by the Company on October 21, 2013 and to the use of our name under the caption "Legal Matters" in the Prospectus Supplement.

Very truly yours,

/s/McDermott Will & Emery LLP

McDermott Will & Emery LLP

U.S. practice conducted through McDermott Will & Emery LLP. 340 Madison Avenue New York New York 10173-1922 Telephone: +1 212 547 5400 Facsimile: +1 212 547 5444 www.mwe.com

#### **CONVERTIBLE NOTE PURCHASE AGREEMENT**

This Convertible Note Purchase Agreement (this "<u>Agreement</u>"), is entered into on October 21, 2013, between Global Eagle Entertainment Inc., a Delaware corporation (the "<u>Company</u>"), and PAR Investment Partners, L.P., a Delaware limited partnership (the "<u>Purchaser</u>").

WHEREAS, on the date hereof, the Company is consummating the acquisition of one or more businesses (the "Acquisition");

WHEREAS, the Purchaser is a stockholder of the Company; and

WHEREAS, in order to provide financing for the Acquisition, the Company desires to sell to the Purchaser, and the Purchaser desires to purchase from the Company, a promissory note which is convertible into shares of the Company's non-voting common stock, par value \$0.0001 per share ("<u>Non-Voting Common Stock</u>"), in the form attached hereto as <u>Exhibit A</u> (the "<u>Note</u>"), on the terms and conditions set forth herein.

**NOW**, **THEREFORE**, in consideration of the premises, representations, warranties and the mutual covenants contained in this Agreement, and for other good and valuable consideration, the receipt, sufficiency and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

#### 1. Purchase of Note.

(a) <u>Purchase</u>. Subject to the terms and conditions contained in this Agreement, the Purchaser hereby agrees to purchase the Note from the Company at the Closing (as defined herein) for a purchase price equal to \$19,000,000 (the "<u>Purchase Price</u>").

(b) <u>Commitment Fee</u>. Upon the execution of this Agreement, the Company shall pay to the Purchaser an aggregate of \$950,000 in cash (the "<u>Commitment Fee</u>").

#### 2. Procedures.

(a) <u>Closing</u>. Subject to the terms and conditions contained in this Agreement, the closing of the sale and purchase of the Note (the "<u>Closing</u>") shall take place on the date hereof immediately prior to the closing of the Acquisition.

(b) <u>Payment of Purchase Price</u>. At the Closing, the Purchaser shall pay to the Company the Purchase Price for the Note in cash by wire transfer to an account or accounts that have been designated by the Company in writing prior to the date hereof.

(c) Note. At the Closing, the Company shall deliver to the Purchaser the Note, duly executed by the Company.

(d) <u>Further Assurances</u>. The parties hereto shall execute and deliver such additional documents and take such additional actions as any party reasonably may deem to be practical and necessary in order to consummate the purchase and sale of the Note as contemplated by this Agreement.

(e) Legends. The Note shall be stamped or otherwise imprinted with a legend in substantially the following form:

#### "THE SECURITIES REPRESENTED BY THIS NOTE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION, AND MAY NOT BE TRANSFERRED IN VIOLATION OF SUCH ACT AND LAWS."

(f) <u>Registration Rights</u>. At the Closing, each of the Company, the Purchaser and the holders of at least 51% of the "Registrable Securities" outstanding as of the time of the Closing, including any Registrable Securities held by the Purchaser (the "<u>Requisite Holders</u>") shall enter into an amendment (the "<u>RRA Amendment</u>") to that certain Amended and Restated Registration Rights Agreement dated as of January 31, 2013, by and among the Company, the Purchaser and the Holders identified therein (the "<u>Registration Rights Agreement</u>"), which RRA Amendment shall be in the form attached hereto as <u>Exhibit B</u>. The term "Registrable Securities" as used in this subsection (f) shall have the meaning given to such term in the Registration Rights Agreement.

#### 3. Representations and Warranties of the Purchaser. The Purchaser represents and warrants to the Company as follows:

(a) <u>Organization and Good Standing</u>. The Purchaser is a limited partnership duly organized, validly existing, and in good standing under the laws of the State of Delaware.

(b) <u>Power and Authority; Enforceability</u>. This Agreement constitutes the legal, valid, and binding obligation of the Purchaser, enforceable against the Purchaser in accordance with its terms. The Purchaser has full entity power and authority to execute and deliver this Agreement and to perform its obligations hereunder. The Purchaser has taken all actions necessary to authorize the execution and delivery of this Agreement, the performance of its obligations hereunder and the consummation of the transactions contemplated hereby. This Agreement has been duly authorized, executed and delivered by the Purchaser.

#### (c) Investment Representations.

(i) The Purchaser is an "accredited investor" as defined in Rule 501 of Regulation D under the Securities Act of 1933, as amended (the "<u>Securities Act</u>").

Agreement.

(ii) The Purchaser has received, has thoroughly read, is familiar with and understands the contents of this

8

(iii) The Purchaser hereby acknowledges that an investment in the Note involves certain significant risks. The Purchaser acknowledges that there is a substantial risk that it will lose all or a portion of its investment and that it is financially capable of bearing the risk of such investment for an indefinite period of time. The Purchaser has no need for liquidity in its investment in the Note for the foreseeable future and is able to bear the risk of that investment for an indefinite period. The Purchaser's present financial condition is such that the Purchaser is under no present or contemplated future need to dispose of any portion of the Note purchased hereby to satisfy any existing or contemplated undertaking, need or indebtedness. The Purchaser's overall commitment to investments which are not readily marketable is not disproportionate to its net worth and the investment in the Company will not cause such overall commitment to become excessive.

(iv) The Purchaser acknowledges that the Note has not been registered under the Securities Act, or any state securities act, and is being sold on the basis of exemptions from registration under the Securities Act and applicable state securities acts. Reliance on such exemptions, where applicable, is predicated in part on the accuracy of the Purchaser's representations and warranties set forth herein. The Purchaser acknowledges and hereby agrees that the Note will not be transferable under any circumstances unless the Note is registered in accordance with federal and state securities laws or the Purchaser finds and complies with an available exemption under such laws. Accordingly, the Purchaser hereby acknowledges that there can be no assurance that it will be able to liquidate its investment in the Company.

(v) There are substantial risk factors pertaining to an investment in the Company. The Purchaser acknowledges that it has read the information set forth above regarding certain of such risks and is familiar with the nature and scope of all such risks, including, without limitation, risks arising from the fact that the Company is an entity with limited operating history and financial resources; and the Purchaser is fully able to bear the economic risks of such investment for an indefinite period, and can afford a complete loss thereof.

(vi) The Purchaser has been given the opportunity to (i) ask questions of and receive answers from the Company and its designated representatives concerning the terms and conditions of the purchase of the Note, the Company and the business and financial condition of the Company and (ii) obtain any additional information that the Company possesses or can acquire without unreasonable effort or expense that is necessary to assist the Purchaser in evaluating the advisability of the purchase of the Note and an investment in the Company. The Purchaser further represents and warrants that, prior to signing this Agreement, it has asked such questions, received such answers and obtained such information as it has deemed necessary or advisable to evaluate the merits and risks of the purchase of the Note and an investment in the Company. The Purchaser is not relying on any oral representation made by any person as to the Company or its operations, financial condition or prospects.

(vii) The Purchaser understands that no federal, state or other governmental authority has made any recommendation, findings or determination relating to the merits of an investment in the Company.

(viii) The Purchaser acknowledges that neither the Company, nor any of its officers, directors, employees, agents or affiliates has made any representation or warranty, express or implied, regarding the Company, the Note or otherwise, other than the representations and warranties set forth herein.

- 3 -

(ix) The Purchaser acknowledges its obligations under the Securities Act, and the rules and regulations promulgated thereunder, with respect to the treatment of non-public information relating to the Company.

4. Representations and Warranties of the Company. The Company represents and warrants to the Purchaser as follows:

Delaware.

(a) Organization. The Company is a corporation duly organized and validly existing under the laws of the State of

(b) <u>Power and Authority; Enforceability</u>. Each of this Agreement, the Note and the RRA Amendment (collectively, the "<u>Transaction Documents</u>") constitutes the legal, valid, and binding obligation of the Company, enforceable against the Company in accordance with its terms. The Company has full power and authority to execute and deliver each of the Transaction Documents and to perform its obligations hereunder and thereunder (including, without limitation, the issuance of the shares of Non-Voting Common Stock upon the conversion of the Note in accordance with its terms, and the issuance of the shares of the Company's common stock, par value \$0.0001 per share (the "<u>Common Stock</u>") upon the conversion of such shares of Non-Voting Common Stock in accordance with the terms of the Company's Second Amended and Restated Certificate of Incorporation (collectively, the "<u>Conversion Securities</u>")). The Company has taken all actions necessary to authorize the execution and delivery of each of the Transaction Documents, the performance of its obligations hereunder and thereunder, and the consummation of the transactions contemplated hereby and thereby (including, without limitation, the issuance of the Conversion Securities in accordance with the terms of the Note or the Company's Second Amended and Restated Certificate of Incorporation and approval of each of the foregoing by a special committee of the Company's Board of Directors (the "<u>Special Committee</u>") that was duly appointed by the Board of Directors and charged with, among other things, reviewing and approving the terms and conditions of each of the Transaction Documents and the transactions contemplated hereby and thereby. Each of the Transaction Documents has been duly authorized, executed, and delivered by the Company.

(c) No Violation; Necessary Approvals. Neither the execution and delivery of any Transaction Document by the Company, nor the consummation or performance by the Company of any of the transactions contemplated hereby or thereby (including, without limitation, the issuance of the Conversion Securities in accordance with the terms of the Note or the Company's Second Amended and Restated Certificate of Incorporation, as applicable) will: (i) with or without notice or lapse of time, constitute, create or result in a breach or violation of, default under, loss of benefit or right under, termination, cancellation, suspension or modification of, or acceleration of performance of any obligation required under any (A) law (statutory, common or otherwise), constitution, ordinance, rule, regulation, executive order or other similar authority enacted, adopted, promulgated or applied by any legislature, agency, bureau, branch, department, division, commission, court, tribunal or other similar recognized organization or body of any federal, state, county, municipal, local or foreign government or other similar recognized organization or body exercising similar powers or authority (collectively, "Law"), (B) order, ruling, decision, award, judgment, injunction or other similar determination or finding by, before or under the supervision of any governmental authority or arbitrator (collectively, "Order"), (C) contract or agreement, (D) permit, license, certificate, waiver, filing, notice or authorization (collectively, "Permit") to which the Company is a party or by which it is bound or any of its assets are subject or (E) any provision of the Company's organizational documents as in effect at the Closing, (ii) result in the imposition of any lien, claim or encumbrance upon any assets owned by the Company; (iii) require any consent, approval, notification, waiver, or other similar action under any contract or agreement or organizational document to which the Company is a party or by which it is bound, (iv) require any Permit under any Law or Order other than (A) required filings, if any, with the United States Securities and Exchange Commission ("SEC") and (B) notifications or other filings with state or federal regulatory agencies after the Closing that are necessary or convenient and do not require approval of the agency as a condition to the validity of the transactions contemplated hereunder or (v) trigger any rights of first refusal, preemptive or preferential purchase or similar rights with respect to the Note or the Conversion Securities.

- 4 -

(d) <u>Authorization of the Conversion Securities</u>. The Non-Voting Common Stock issuable upon conversion of the Note is duly authorized and when issued in accordance with this Agreement and the Note, will be duly and validly issued, fully paid and non-assessable and will be free and clear of all liens, claims or encumbrances, other than (A) transfer restrictions hereunder, (B) transfer restrictions under federal and state securities laws, and (C) liens, claims or encumbrances imposed due to the actions of the Purchaser. The Common Stock issuable upon conversion of such shares of Non-Voting Common Stock is duly authorized and when issued in accordance with the Company's Second Amended and Restated Certificate of Incorporation, will be duly and validly issued, fully paid and non-assessable and will be free and clear of all liens, claims or encumbrances, other than (A) transfer restrictions hereunder, (B) transfer restrictions under federal and state securities laws, and (C) liens, claims or encumbrances imposed due to the actions of the Purchaser.

(e) SEC Filings; Financial Statements. The Company has filed with the SEC and has heretofore made available to the Purchaser true and complete copies of all forms, reports, schedules, statements and other documents required to be filed by it and its subsidiaries under the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>") (collectively, the "<u>Company SEC Documents</u>"). As of their respective dates or, if amended, as of the date of the last such amendment, the Company SEC Documents, including, without limitation, any financial statements or schedules included therein, complied in all material respects with the applicable requirements of the Securities Act and the Exchange Act, and did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Each of the balance sheets (including the related notes) included in the Company SEC Documents fairly presented in all material respects the financial position of the Company and its subsidiaries as of the respective dates thereof, and the other related financial statements (including the related notes) included therein fairly presented in all material respects the results of operations and cash flows of the Company and its subsidiaries for the respective periods or as of the respective dates set forth therein. Each of the balance sheets and statements of operations and cash flows (including the related notes) included in the Company SEC Documents has been prepared in all material respects in accordance with generally accepted accounting principles in the United States, except as otherwise noted therein and subject, in the case of unaudited interim financial statements, to normal year-end adjustments.

**5.** Closing Conditions. The obligation of the Purchaser to purchase the Note under this Agreement shall be subject to the fulfillment, at or prior to the Closing, of each of the following conditions, any of which, to the extent permitted by applicable laws, may be waived by the Purchaser:

(a) The Company shall have issued and sold (for immediately available cash proceeds) shares of its Common Stock to funds affiliated with Putnam Investments (the "<u>Putnam Sale</u>") resulting in the receipt by the Company of cash proceeds therefrom in an aggregate amount equal to or greater than the sum of (i) the Purchase Price hereunder plus (ii) the aggregate amount of interest that would accrue on the aggregate Purchase Price at the Interest Rate (as defined in the Note) over a period of sixty (60) days. For the avoidance of doubt, this condition shall be satisfied if the Putnam Sale is consummated concurrently with the Closing hereunder and the Acquisition.

(b) The Acquisition shall be consummated concurrently with the Closing.

(c) The Company shall have delivered the Note to the Purchaser.

Amendment.

(d) Each of the Company, the Purchaser and the Requisite Holders shall have duly executed and delivered the RRA

(e) The representations and warranties of the Company set forth in <u>Section 4</u> of this Agreement shall have been true and correct as of the date hereof and shall be true and correct as of the Closing with the same effect as though such representations and warranties had been made on and as of such date (other than any such representation or warranty that is made by its terms as of a specified date, which shall be true and correct as of such specified date).

(f) No order, writ, judgment, injunction, decree, determination, or award shall have been entered by or with any governmental, regulatory, or administrative authority or any court, tribunal, or judicial, or arbitral body, and no other legal restraint or prohibition shall be in effect, preventing the purchase by the Purchaser of the Note or the issuance of the Conversion Securities.

(g) Each of the Transaction Documents and each of the transactions contemplated hereby and thereby shall have been duly authorized and approved by the Special Committee, and the Company shall have delivered a certificate of an officer of the Company certifying as to the resolutions duly adopted by the Special Committee.

(h) The Company shall have paid the Commitment Fee to the Purchaser.

6. Termination. This Agreement may be terminated at any time prior to the Closing:

(a) by mutual written consent of the Company and the Purchaser; or

- 6 -

(b) automatically if either the Putnam Sale or the Acquisition is not consummated on the date of this Agreement.

In the event of any termination of this Agreement pursuant to this <u>Section 6</u>, this Agreement shall forthwith become null and void and have no effect, without any liability on the part of the Purchaser or the Company and their respective directors, officers, employees, partners, managers, members, or stockholders and all rights and obligations of each party shall cease; provided, however, that nothing contained in this <u>Section 6</u> shall relieve either party from liabilities or damages arising out of any fraud or willful breach by such party of any of its representations, warranties, covenants or agreements contained in this Agreement.

#### 7. General Provisions.

(a) <u>Survival of Representations and Warranties</u>. All of the representations and warranties contained herein shall survive the Closing.

(b) <u>Entire Agreement</u>. The Transaction Documents, together with any documents, instruments and writings that are delivered pursuant hereto or referenced herein or therein, constitutes the entire agreement and understanding of the parties hereto in respect of its subject matter and supersedes all prior understandings, agreements, or representations by or among the parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof or the transactions contemplated hereby.

(c) <u>Successors</u>. All of the terms, agreements, covenants, representations, warranties, and conditions of this Agreement are binding upon, and inure to the benefit of and are enforceable by, the parties hereto and their respective successors.

(d) <u>Assignments</u>. Except as otherwise provided herein, no party hereto may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other party. Any purported assignment in violation of this <u>Section 7(d)</u> shall be void and ineffectual and shall not operate to transfer or assign any interest or title to the purported assignee.

(e) <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which will be deemed an original but all of which together will constitute one and the same instrument.

(f) <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and will not affect in any way the meaning or interpretation of this Agreement.

(g) <u>Governing Law</u>. This Agreement and any litigation between the parties (whether grounded in contract, tort, statute, law or equity) related to this Agreement shall be governed by, construed in accordance with, and interpreted pursuant to the laws of the State of Delaware, without giving effect to its choice of laws principles.

- 7 -

(h) <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to a jury trial in connection with any litigation pursuant to this Agreement and the transactions contemplated hereby.

(i) <u>Amendments</u>. This Agreement may not be amended, modified or waived as to any particular provision, except by a written instrument executed by all parties hereto.

(j) <u>Severability</u>. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof; provided that if any provision of this Agreement, as applied to any party hereto or to any circumstance, is adjudged by a governmental authority, arbitrator, or mediator not to be enforceable in accordance with its terms, the parties hereto agree that the governmental authority, arbitrator, or mediator making such determination will have the power to modify the provision in a manner consistent with its objectives such that it is enforceable, and/or to delete specific words or phrases, and in its reduced form, such provision will then be enforceable and will be enforced.

(k) <u>Expenses</u>. The Company shall reimburse the Purchaser for the reasonable costs and expenses incurred by the Purchaser in connection with the preparation, execution and performance of the Transaction Documents and the consummation of the transactions contemplated hereby and thereby, including, without limitation, the reasonable fees and expenses of Goodwin Procter LLP, as counsel to the Purchaser. The Company shall bear all of its own costs and expenses of any kind or nature incurred by it in connection with the preparation, execution and performance of the Transaction Documents and the consummation of the transactions contemplated hereby and thereby and thereby.

(1) <u>Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties hereto and no presumption or burden of proof will arise favoring or disfavoring any party hereto because of the authorship of any provision of this Agreement. Any reference to any federal, state, local, or foreign Law will be deemed also to refer to Law as amended and all rules and regulations promulgated thereunder, unless the context requires otherwise. The words "*include*," "*includes*," and "*including*" will be deemed to be followed by "*without limitation*." Pronouns in masculine, feminine, and neuter genders will be construed to include any other gender, and words in the singular form will be construed to include the plural and vice versa, unless the context otherwise requires. The words "*this Agreement*," "*hereof*," "*hereof*," "*hereunder*," and words of similar import refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited. The parties hereto has breached any representation, warranty, or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which such party hereto has not breached will not detract from or mitigate the fact that such party hereto is in breach of the first representation, warranty, or covenant.

- 8 -

(m) <u>Waiver</u>. No waiver by any party hereto of any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, may be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising because of any prior or subsequent occurrence.

(n) <u>Confidentiality</u>. Except as may be required by law, regulation or applicable stock exchange listing requirements (including, without limitation, legal requirements that are applicable to Purchaser under Section 13(d) and Section 16 of the Exchange Act, and the rule and regulations promulgated thereunder) unless and until the transactions contemplated by this Agreement are publicly announced, the parties hereto shall keep confidential and shall not publicly disclose the existence or terms of this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the undersigned have executed this Agreement to be effective as of the date first set forth above.

COMPANY:

GLOBAL EAGLE ENTERTAINMENT INC.

By: /s/ Michael Pigott

Name: Michael Pigott Title: Vice-President, Legal Affairs

PURCHASER:

PAR INVESTMENT PARTNERS, L.P. By: PAR Group, L.P., its general partner By: PAR Capital Management, Inc., its general partner

By: /s/ Steven M. Smith

Name: Steven M. Smith Title: Chief Operating Officer and General Counsel

#### Exhibit A

THE SECURITIES REPRESENTED BY THIS NOTE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION, AND MAY NOT BE TRANSFERRED IN VIOLATION OF SUCH ACT AND LAWS.

#### **CONVERTIBLE UNSECURED PROMISSORY NOTE**

\$19,000,000

October 21, 2013 Westlake Village, California

For value received, Global Eagle Entertainment Inc., a Delaware corporation (the "<u>Company</u>"), promises to pay to PAR Investment Partners, L.P. (the "<u>Holder</u>"), the principal sum of Nineteen Million Dollars (\$19,000,000). Interest shall accrue from the date of this Note on the unpaid principal amount at a rate equal to twelve percent (12%) per annum (the "<u>Interest Rate</u>). The Interest Rate shall be computed on the basis of the actual number of days elapsed and a year of 365 days. Capitalized terms used herein but not otherwise defined herein shall have the meanings ascribed to such terms in the Convertible Note Purchase Agreement, dated as of October 21, 2013 (the "<u>Purchase Agreement</u>"), by and between the Company and the Holder. This Note is subject to the following terms and conditions.

1. <u>Maturity</u>. The principal amount of this Note, together with any accrued but unpaid interest thereon (the "<u>Total Loan Obligation</u>"), shall be due and payable in full on the earlier of (i) December 20 and (ii) the occurrence of an Event of Default (as defined below) (such date, the "<u>Maturity Date</u>"). If the Maturity Date occurs and no Event of Default has occurred, then the Company shall, at its election, on the Maturity Date, either (a) pay to the Holder cash in an aggregate amount equal to the Total Loan Obligation as of such Maturity Date or (b) issue to the Holder that number of fully paid and non-assessable shares of Non-Voting Common Stock as is equal to the quotient obtained by dividing (x) the Total Loan Obligation as of such Maturity Date by (y) a conversion price equal to \$8.5593 per share (the "<u>Conversion Price</u>") (subject to adjustment in accordance with <u>Section 6</u> below), with any fractional shares paid in cash. If the Maturity Date occurs and an Event of Default has occurred, then the Company shall, at the Holder's election (by notice delivered to the Company), on the Maturity Date, either (a) pay to the Holder cash in an aggregate amount equal to the Total Loan Obligation as of such Maturity Date or (b) issue to the Holder that number of fully paid and non-assessable shares of Non-Voting Common Stock, in an amount calculated in the manner described in the preceding sentence.

2. <u>Prepayment</u>. The Company shall have the right to prepay in cash all or any portion of the Note prior to the Maturity Date without penalty or premium.

3. <u>Transferability of Note</u>. This Note may not be transferred or assigned in whole or in part by the Holder without the prior written consent of the Company.

4. **Events of Default.** If there shall be an Event of Default, the entire unpaid principal balance of this Note, together with any accrued but unpaid interest thereon, shall become immediately due and payable regardless of any prior forbearance and without presentment, demand, protest or further notice of any kind, all of which are hereby expressly waived by the Company and the Holder may exercise any and all rights and remedies available to the Holder under applicable law, including, without limitation, the right to collect from the Company all amounts due under this Note.

The occurrence of any one or more of the following events with respect to the Company constitutes an "Event of Default" hereunder:

(a) Any representation, warranty or certification made by or on behalf of the Company in the Purchase Agreement or this Note, or in any certificate, writing or other document delivered pursuant hereto shall prove to have been incorrect when made;

(b) The Company breaches any covenant or agreement in the Purchase Agreement or this Note, and fails to cure such breach within twenty (20) days after receipt of written notice thereof from the Holder;

(c) The Company fails to pay the principal of this Note or any accrued interest thereon when due, whether at maturity or at a date fixed by acceleration or otherwise, and fails to cure such breach within five (5) days after such payment was due;

(d) The entry of any judgment or order against the Company that could reasonably be expected to have a material adverse effect on the businesses or operations of the Company, taken as a whole, which remains unsatisfied or undischarged and in effect for thirty (30) days after such entry without a stay of enforcement or execution;

(e) The Company shall (i) apply for or consent to the appointment of a receiver, trustee, liquidator or custodian of itself, (ii) make a general assignment for the benefit of its creditors, (iii) be dissolved or liquidated, (iv) commence a voluntary case or other proceeding seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar law now or hereafter in effect or consent to any such relief or to the appointment of or taking possession of its property by any official in an involuntary case or other proceeding commenced against it, (v) become unable, admit in writing its inability, or fail generally to pay its debts as they become due, or (vi) take any action for the purpose of effecting any of the foregoing;

(f) Proceedings for the appointment of a receiver, trustee, liquidator or custodian of the Company, or an involuntary case or other proceedings seeking liquidation, reorganization or other relief with respect to the Company or the debts thereof under any bankruptcy, insolvency or other similar law now or hereafter in effect shall be commenced and an order for relief entered or such proceeding shall not be dismissed or discharged within thirty (30) days of commencement, or an order or decree approving or ordering any of the foregoing shall be entered; or

(g) The Company shall default in the payment of any principal, interest or premium, or any observance or performance of any covenants or agreements, with respect to any indebtedness (excluding trade payables and other indebtedness entered into in the ordinary course of business) in excess of \$100,000 in the aggregate for borrowed money or any obligation which is the substantive equivalent thereof and such default shall continue for more than the period of grace, if any, and if any such indebtedness or obligation shall be declared due and payable prior to the stated maturity date thereof; or

(h) The Company shall assert in writing that any material provisions of this Note are void or unenforceable.

5. <u>Registration Rights</u>. In the event that the Note converts into shares of Non-Voting Common Stock, the Company shall, within thirty (30) days of the effectiveness of such conversion, prepare and file either (a) a new registration statement (a "<u>New Resale Registration Statement</u>") or (b) a post-effective amendment (a "<u>Post-Effective Amendment</u>") to that certain registration statement on Form S-3 of the Company (SEC File No. 333-188121) (the "<u>Existing Resale Registration Statement</u>"), with the Securities and Exchange Commission, which in either case, (i) provides for the resale of the Common Stock that may be issued upon conversion of the Non-Voting Common Stock in accordance with the Securities Act and (ii) is substantially similar to (to the extent applicable) and no less favorable to the Purchaser than, the Existing Resale Registration Statement. The Company shall use reasonable best efforts to cause the New Resale Registration Statement or Post-Effective Amendment, as the case may be, to be declared effective under the Securities Act as soon as possible after its filing, and in any event within thirty (30) days after the filing thereof, and to keep such New Resale Registration Statement or the Existing Resale Registration Statement (as amended by the Post-Effective Amendment), as the case may be, effective under the Securities Act at all times until the expiration of the Effectiveness Period (as such term is defined in the Registration Rights Agreement).

6. Certain Adjustments. The provisions of this Section 6 shall apply to adjust the Conversion Price as set forth in Section 1 above.

a. <u>Stock Dividends, Subdivision, Combination or Reclassification of Non-Voting Common Stock</u>. If at any time from and after the date of the issuance of this Note through the Maturity Date, the Company shall (i) pay a dividend on the Non-Voting Common Stock in shares of its capital stock, (ii) combine its outstanding shares of Non-Voting Common Stock into a smaller number of shares, (iii) subdivide its outstanding shares of Non-Voting Common Stock as the case may be, or (iv) issue by reclassification of its shares of Non-Voting Common Stock any shares of capital stock of the Company, then, on the record date for such dividend or the effective date of such subdivision or split-up, combination or reclassification, as the case may be, the number and kind of shares to be delivered upon conversion of this Note will be adjusted so that the Holder will be entitled to receive the number and kind of shares of capital stock that Holder would have owned or been entitled to receive upon or by reason of such event had this Note been converted immediately prior thereto, and the Conversion Price will be adjusted as provided below in paragraph (d).

b. Extraordinary Distributions. If at any time from and after the date of issuance of this Note through the Maturity Date, the Company shall distribute to all holders of Non-Voting Common Stock (including any such distribution made in connection with a consolidation or merger in which the Company is the continuing or surviving corporation and Non-Voting Common Stock is not changed or exchanged) cash, evidences of indebtedness, securities or other assets (excluding dividends payable in shares of capital stock for which adjustment is made under paragraph (a) above), or rights, options or warrants to subscribe for or purchase securities of the Company, then in each such case the number of shares of Non-Voting Common Stock to be delivered to Holder upon conversion of this Note shall be increased so that the Holder thereafter shall be entitled to receive the number of shares of Non-Voting Common Stock determined by multiplying the number of shares such Holder would have been entitled to receive immediately before such record date by a fraction, the denominator of which shall be the Company in good faith) of the portion of the cash, evidences of indebtedness, securities or other assets so distributed or of such rights, options or warrants applicable to one share of the Non-Voting Common Stock (provided that such denominator shall in no event be less than \$.01) and the numerator of which shall be the Conversion Price.

c. <u>Reorganization, etc</u>. If at any time from and after the date of issuance of this Note through the Maturity Date, any consolidation of the Company with or merger of the Company with or into any other person (other than a merger or consolidation in which the Company is the surviving or continuing corporation and which does not result in any reclassification of, or change (other than a change in par value or from par value to no par value or from no par value to par value, or as a result of a subdivision or combination) in, outstanding shares of Non-Voting Common Stock) or any sale, lease or other transfer of all or substantially all of the assets of the Company to any other person (each, a "Reorganization <u>Event</u>"), shall be effected in such a way that the holders of the Non-Voting Common Stock shall be entitled to receive cash, stock, other securities or assets (whether such cash, stock, other securities or assets are issued or distributed by the Company or another person) with respect to or in exchange for the Non-Voting Common Stock, then this Note shall automatically become convertible only for the kind and amount of cash, stock, other securities or assets receivable upon such Reorganization Event by a holder of the number of shares of the Non-Voting Common Stock that such holder would have been entitled to receive upon conversion of this Note had this Note been converted immediately before such Reorganization Event, subject to adjustments that shall be as nearly equivalent as may be practicable to the adjustments provided for in this <u>Section 6</u>. The Company shall not enter into any of the transactions referred to in this paragraph (c).

d. <u>Conversion Price Adjustment</u>. Whenever the number of shares of Non-Voting Common Stock issuable upon the conversion of the Note is adjusted as provided pursuant to this <u>Section 6</u>, the Conversion Price shall be adjusted by multiplying such Conversion Price immediately prior to such adjustment by a fraction, of which the numerator shall be the number of shares of Non-Voting Common Stock issuable upon the conversion of this Note immediately prior to such adjustment, and of which the denominator shall be the number of shares of Non-Voting Common Stock issuable upon the conversion of this Note immediately thereafter; provided, however, that the Conversion Price shall in no event be less than the par value of a share of such Non-Voting Common Stock.

e. <u>Notice of Adjustment</u>. Whenever the number of shares of Non-Voting Common Stock issuable upon the conversion of this Note or the Conversion Price is adjusted as herein provided, the Company shall promptly mail by first class mail, postage prepaid, to the Holder, notice of such adjustment or adjustments setting forth the number of shares of Non-Voting Common Stock issuable upon the conversion of this Note and the Conversion Price after such adjustment, setting forth a brief statement of the facts requiring such adjustment and setting forth the computation by which such adjustment was made.

f. <u>Limitation</u>. Notwithstanding anything to the contrary contained herein, in no event shall the aggregate number of shares of Non-Voting Common Stock that may be issued upon conversion of the Note exceed 19.9% of either (a) the total number of shares of Common Stock outstanding on the date of the Purchase Agreement or (b) the total voting power of the Company's securities outstanding on the date of the Purchase Agreement that are entitled to vote on matters voted on by holders of the Common Stock, unless and until the Company has obtained the approval of its stockholders as required by the applicable rules of the NASDAQ Stock Market for issuances of shares in excess of such amount.

7. <u>Reservation and Listing of Shares</u>. The Company shall reserve and keep available solely for issuance upon (i) the conversion of the Total Loan Obligation into shares of Non-Voting Common Stock and (ii) the conversion of such shares of Non-Voting Common Stock into Common Stock, such number of shares of Non-Voting Common Stock and Common Stock, as applicable, as will from time to time be sufficient to permit the conversion of all of the outstanding Total Loan Obligation, and, if applicable, shall take all action to increase the authorized number of Non-Voting Common Stock and/or Common Stock, as applicable, if at any time there shall be insufficient authorized but unissued shares of Non-Voting Common Stock and/or Common Stock, as applicable, to permit such reservation or permit the conversion of all of the outstanding Total Loan Obligation. The Company shall take such actions as may be necessary to list for trading on the NASDAQ Stock Market upon their issuance all shares of Common Stock issued upon conversion of all Non-Voting Common Stock issued upon conversion of all Non-Voting Common Stock issued upon conversion of the Total Loan Obligation.

8. <u>Governing Law</u>. This Note and all acts and transactions pursuant hereto and the rights and obligations of the parties hereto shall be governed, construed and interpreted in accordance with the laws of the State of Delaware, without giving effect to principles of conflicts of law.

9. <u>Notices</u>. All notices and other communications required or permitted hereunder shall be in writing and shall be deemed effectively given upon personal delivery, upon three (3) business days after deposit with the United States Post Office, by registered or certified mail, return receipt requested, postage prepaid, one (1) business day after deposit with a nationally recognized air courier, upon receipt of confirmation with regard to delivery by facsimile or upon twelve (12) hours after delivery by electronic mail and addressed: (i) if to the Holder, to PAR Investment Partners, L.P., 1 International Place, Boston, MA 02110, or at such other address, facsimile number or electronic mail address as the Holder shall have furnished to the Company in writing, or (ii) if to the Company, at its current address or at such other address, facsimile number or electronic mail address as the Company shall have furnished to the Holder in writing.

10. <u>Amendments and Waivers</u>. Any term of this Note may be amended and the observance of any term of this Note may be waived (either generally or in a particular instance and either retroactively or prospectively), only with the written consent of the Company and the Holder. Any amendment or waiver effected in accordance with this <u>Section 10</u> shall be binding upon the Holder and the Company.

11. <u>Stockholders, Officers and Directors Not Liable</u>. In no event shall any stockholder, officer or director of the Company be liable for any amounts due or payable pursuant to this Note.

12. Lost Documents. Upon receipt by the Company of evidence and indemnity satisfactory to it of the loss, theft, destruction or mutilation of, and upon surrender and cancellation of this Note, if mutilated, the Company will make and deliver in lieu of this Note a new note of the same series and of like tenor and unpaid principal amount and dated as of the date to which interest, if any, has been paid on the unpaid principal amount of this Note.

13. Waiver. The Company hereby waives presentment, protest and notice of dishonor.

14. **Remedies.** No remedy herein conferred upon the Holder is intended to be exclusive of any other remedy, and each and every such remedy shall be cumulative and shall be in addition to every right or other remedy now or hereafter existing at law or in equity or by statute or otherwise.

15. <u>Severability</u>. Whenever possible, each provision of this Note shall be interpreted in such manner as to be effective and valid under all applicable laws and regulations. If, however, any provision of this Note shall be prohibited by or invalid under any such law or regulation in any jurisdiction, it shall, as to such jurisdiction, be deemed modified to conform to the minimum requirements of such law or regulation, or, if for any reason it is not deemed so modified, it shall be ineffective and invalid only to the extent of such prohibition or invalidity without affecting the remaining provisions of this Note, or the validity or effectiveness of such provision in any other jurisdiction.

[Signature Page Follows]

IN WITNESS WHEREOF, the undersigned has executed this Convertible Unsecured Promissory Note as of the date first written above.

# GLOBAL EAGLE ENTERTAINMENT INC.

By:

Name: Title: Exhibit B

RRA Amendment

[see attached]

#### AGREEMENT

This AGREEMENT (the "<u>Agreement</u>") is made and entered into as of October 21, 2013, by and between Global Eagle Entertainment Inc., a Delaware corporation (the "<u>Company</u>"), and PAR Investment Partners, L.P., a Delaware limited partnership (together with its affiliates, the "<u>Holder</u>").

### RECITALS

WHEREAS, the Holder beneficially owns 19,118,233 shares (the "<u>Shares</u>") of the Company's non-voting common stock, par value \$0.0001 per share (the "<u>Non-Voting Common Stock</u>") and may in the future acquire additional shares of Non-Voting Common Stock;

WHEREAS, pursuant to Article IV, Section 4.4(i) of the Company's Second Amended and Restated Certificate of Incorporation (the "<u>Certificate</u>"), any outstanding share of Non-Voting Common Stock shall, without the payment of any additional consideration or any other action on the part of the Company or the holder thereof, convert into one fully paid and non-assessable share of the Company's common stock, par value \$0.0001 per share (the "<u>Common Stock</u>") immediately upon the delivery of written notice by the holder of such share of such holder's election to convert such share of Non-Voting Common Stock into a share of Common Stock (the "<u>Conversion Notice</u>") to the office of the transfer agent of the Company (the "<u>Transfer Agent</u>") during normal business hours at any time on or after October 31, 2013; and

WHEREAS, both the Company and the Holder are interested in prolonging the period in which the Non-Voting Common Stock cannot be converted into Common Stock and desire to enter into this Agreement in order to agree to certain matters relating to the conversion by the Holder of the Shares and any additional shares of Non-Voting Common Stock acquired by the Holder after the date of this Agreement.

#### AGREEMENT

NOW, THEREFORE, in consideration of the mutual promises, covenants and agreements contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

- <u>Conversion Forbearance</u>. Notwithstanding the rights contained in Article IV, Section 4.4(i) of the Certificate, the Holder hereby
  agrees that Holder and its affiliates will forbear from electing to convert any of the Shares and any additional shares of NonVoting Common Stock acquired by the Holder after the date of this Agreement from Non-Voting Common Stock into Common
  Stock until the earlier of (i) May 1, 2014 and (ii) 30 days after the date on which the shares of capital stock of Advanced
  Inflight Alliance AG, a German stock corporation ("AIA"), cease to be listed on the Regulated Market (General Standard) of
  the Frankfurt Stock Exchange (the earlier of such dates, the "Outside Date").
- 2. <u>Legal and Factual Restriction</u>. It is the Company's and the Holder's mutual understanding that the Holder does not have the right to demand nor the factual ability to convert its Non-Voting Common Stock into Common Stock prior to the Outside Date.
- 3. <u>No Further Restriction</u>. For the avoidance of doubt, nothing contained in Sections 1 and 2 shall restrict, preclude or otherwise impair any of the rights of the Holder or its affiliates in any way except with respect to conversion of the Shares and any additional shares of Non-Voting Common Stock acquired by the Holder after the date of this Agreement pursuant to Article IV, Section 4.4(i) of the Certificate prior to the Outside Date, and Holder shall continue to have any and all other rights and privileges under the Certificate, including without limitation, the right to cause a conversion of the Shares or any additional shares of Non-Voting Common Stock acquired by the Holder after the date of this Agreement upon any transfer of such shares in accordance with Article IV, Section 4.4(ii) of the Certificate.

- 4. <u>Transfer Agent Instructions</u>. The parties agree that the Company may direct the Transfer Agent not to process any conversion request made by the Holder or any of its affiliates pursuant to Article IV, Section 4.4(i) of the Certificate to the extent, but only to the extent, necessary to enforce the agreements contained in Section 1 of this Agreement. For the avoidance of doubt, no such instructions shall prevent any conversion of the Shares or any additional shares of Non-Voting Common Stock acquired by the Holder after the date of this Agreement by Holder or any of its affiliates following the Outside Date.
- 5. <u>Amendment; Waiver</u>. Any term hereof may be amended or waived only with the prior written consent of both the Company and the Holder. Any amendment or waiver effected in accordance with this Section 5 shall be binding upon the Company and the Holder and each of their respective successors and assigns.
- 6. <u>Severability</u>. If one or more of the provisions of this Agreement are held to be unenforceable under applicable law, then such provision shall be excluded from this Agreement, and the balance of the Agreement shall be interpreted as if such provision were so excluded and shall be enforceable in accordance with its terms.
- 7. <u>Governing Law</u>. This Agreement and all acts and transactions pursuant hereto and the rights and obligations of the parties hereto shall be governed, construed and interpreted in accordance with the laws of the State of Delaware, without giving effect to principles of conflicts of law.
- 8. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original and all of which together shall constitute one instrument.
- 9. <u>Successors and Assigns</u>. The terms and conditions of this Agreement shall inure to the benefit of and be binding upon the respective successors and assigns of the parties. Except as expressly provided herein, nothing in this Agreement is intended to confer upon any party other than the parties hereto or their respective successors and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement.
- 10. <u>Entire Agreement</u>. This Agreement constitutes the entire agreement between the parties hereto pertaining to the subject matter hereof, and any and all other written or oral agreements relating to the subject matter hereof existing between the parties hereto are hereby superseded.

[Signature pages follow]

2

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on and as of the date first written above.

# GLOBAL EAGLE ENTERTAINMENT INC.

By:/s/ Michael PigottName:Michael PigottTitle:Vice-President, Legal Affairs

# SIGNATURE PAGE TO CONVERSION AGREEMENT

# PAR INVESTMENT PARTNERS, L.P.

- By: PAR Group, L.P., its general partner
- By: PAR Capital Management, Inc., its general partner

By: /s/ Steven M. Smith

Name: Steven M. Smith

Title: Chief Operating Officer and General Counsel

# SIGNATURE PAGE TO CONVERSION AGREEMENT

# COMMON STOCK PURCHASE AGREEMENT

This Common Stock Purchase Agreement (this "<u>Agreement</u>"), is entered into on October 18, 2013, between Global Eagle Entertainment Inc., a Delaware corporation (the "<u>Company</u>"), and Putnam Equity Spectrum Fund, a series of Putnam Funds Trust, a Massachusetts business trust (the "<u>Purchaser</u>").

WHEREAS, the Company is considering the acquisition of one or more businesses (each, a "Target").

**WHEREAS**, in order to secure financing for the acquisition of a Target (the "<u>Transaction</u>"), the parties hereto desire to enter into this Agreement to set forth certain agreements with respect to the sale and purchase of shares of the Company's common stock, par value \$0.0001 per share ("<u>GE Common Stock</u>"), upon and subject to the closing of the Transaction (the "<u>Closing</u>").

**WHEREAS**, the Company has filed a shelf registration statement on Form S-3 which was declared effective by the Securities and Exchange Commission (the "<u>SEC</u>") on August 22, 2013 (as amended, the "<u>Shelf Registration Statement</u>").

**WHEREAS**, the Company intends to offer the GE Common Stock pursuant to this Agreement by means of a prospectus supplement to the prospectus included in the Shelf Registration Statement (the "<u>Prospectus Supplement</u>") which Prospectus Supplement shall be filed with the SEC pursuant to Rule 424 of the Securities Act of 1933, as amended (the "<u>Securities Act</u>).

**NOW**, **THEREFORE**, in consideration of the premises and the mutual covenants contained in this Agreement, and for other good and valuable consideration, the receipt, sufficiency and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

#### 1. Common Stock Purchase.

The Purchaser hereby agrees that concurrent with, and subject to, the Closing, it shall purchase from the Company, and the Company hereby agrees that concurrent with, and subject to, the Closing, it shall issue and sell to the Purchaser, 2,453,472 shares of GE Common Stock (the "<u>Shares</u>") at a per share purchase price equal to \$8.5593 for an aggregate purchase price of \$21,000,002 (the "<u>Purchase</u> <u>Price</u>"), payable by the Purchaser by wire transfer of immediately available funds.

#### 2. Fractional Shares.

If any fractional share would be obligated to be issued to the Purchaser pursuant to <u>Section 1</u>, the number of shares issued to the Purchaser shall be rounded up to the nearest whole share.

#### 3. Procedures.

(a) Timing of Actions. All actions taken at the Closing shall be deemed to have been taken simultaneously.

(b) <u>Purchase Price</u>. At the Closing, the Purchaser shall deliver to the Company the Purchase Price for the Shares against delivery of the Shares by the Company.

(c) <u>Further Assurances</u>. The parties hereto shall execute and deliver such additional documents and take such additional actions as either party reasonably may deem to be practical and necessary in order to consummate the purchase and sale of the GE Common Stock as contemplated by this Agreement.

#### 4. Closing Conditions.

The obligation of the Purchaser to purchase shares of GE Common Stock under this Agreement shall be subject to the fulfillment, at or prior to the Closing, of each of the following conditions, any of which, to the extent permitted by applicable laws, may be waived by the Purchaser:

(a) The Closing shall have occurred, or happen concurrently with, the purchase of the Shares.

(b) The Company shall have offered and delivered the Shares to the Purchaser pursuant to the Prospectus Supplement.

(c) The Shares shall have been duly authorized, validly issued and, when paid for in accordance with this Agreement, will be fully paid and non-assessable, and the issuance of such Shares shall not be subject to any preemptive or similar rights.

(d) The Company and affiliates of the Purchaser shall have entered into a voting rights waiver agreement in the form attached hereto as <u>Exhibit A</u>.

#### 5. Termination.

This Agreement may be terminated at any time prior to the Closing by mutual written consent of the Company and the Purchaser.

#### 6. General Provisions.

(a) <u>Entire Agreement</u>. This Agreement, together with any documents, instruments and writings that are delivered pursuant hereto or referenced herein, constitutes the entire agreement and understanding of the parties hereto in respect of its subject matter and supersedes all prior understandings, agreements, or representations by or among the parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof or the transactions contemplated hereby.

(b) <u>Successors</u>. This Agreement shall be binding upon, and inure to the benefit of and are enforceable by, the parties hereto and their respective successors.

(c) <u>Assignments</u>. Except as otherwise provided herein, no party hereto may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other party.

(d) <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which will be deemed an original but all of which together will constitute one and the same instrument.

(e) <u>Headings.</u> The section headings contained in this Agreement are inserted for convenience only and will not affect in any way the meaning or interpretation of this Agreement.

(f) <u>Governing Law</u>. This Agreement, the entire relationship of the parties hereto, and any litigation between the parties (whether grounded in contract, tort, statute, law or equity) shall be governed by, construed in accordance with, and interpreted pursuant to the laws of the State of Delaware, without giving effect to its choice of laws principles.

(g) <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to a jury trial in connection with any litigation pursuant to this Agreement and the transactions contemplated hereby.

(h) <u>Amendments</u>. This Agreement may not be amended, modified or waived as to any particular provision, except by a written instrument executed by all parties hereto.

(i) <u>Severability</u>. The provisions of this Agreement will be deemed severable and the invalidity or unenforceability of any provision will not affect the validity or enforceability of the other provisions hereof; provided that if any provision of this Agreement, as applied to any party hereto or to any circumstance, is adjudged by a governmental authority, arbitrator, or mediator not to be enforceable in accordance with its terms, the parties hereto agree that the governmental authority, arbitrator, or mediator making such determination will have the power to modify the provision in a manner consistent with its objectives such that it is enforceable, and/or to delete specific words or phrases, and in its reduced form, such provision will then be enforceable and will be enforced.

(j) <u>Expenses</u>. Each party hereto will bear its own costs and expenses incurred in connection with the preparation, execution and performance of this Agreement and the consummation of the transactions contemplated hereby, including all fees and expenses of agents, representatives, financial advisors, legal counsel and accountants.

(k) <u>Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the parties hereto and no presumption or burden of proof will arise favoring or disfavoring any party hereto because of the authorship of any provision of this Agreement. Any reference to any federal, state, local, or foreign Law will be deemed also to refer to Law as amended and all rules and regulations promulgated thereunder, unless the context requires otherwise. The words *"include," "includes,"* and *"including"* will be deemed to be followed by *"without limitation."* Pronouns in masculine, feminine, and neuter genders will be construed to include any other gender, and words in the singular form will be construed to include the plural and vice versa, unless the context otherwise requires. The words *"this Agreement," "hereof," "hereof," "hereunder,"* and words of similar import refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited. The parties hereto intend that each covenant contained herein will have independent significance. If any party hereto has breached any covenant contained herein in any respect, the fact that there exists another covenant relating to the same subject matter (regardless of the relative levels of specificity) which such party hereto has not breached will not detract from or mitigate the fact that such party hereto is in breach of the first covenant.

- 3 -

(1) <u>Waiver</u>. No waiver by any party hereto of any default or breach of covenant hereunder, whether intentional or not, may be deemed to extend to any prior or subsequent default or breach of covenant hereunder or affect in any way any rights arising because of any prior or subsequent occurrence.

(m) <u>Confidentiality</u>. Except as may be required by law, regulation or applicable stock exchange listing requirements or, unless and until the Transaction is publicly announced, the parties hereto shall keep confidential and shall not publicly disclose the existence or terms of this Agreement.

[Signature page follows]

IN WITNESS WHEREOF, the undersigned have executed this Agreement to be effective as of the date first set forth above.

COMPANY:

GLOBAL EAGLE ENTERTAINMENT INC.

By: /s/ Michael Pigott

Name: Michael Pigott Title: Vice-President, Legal Affairs

PURCHASER:

PUTNAM EQUITY SPECTRUM FUND

By: Putnam Investment Management, LLC

By: /s/ David Glancy

Name: David Glancy Title: Portfolio Manager

A copy of the Agreement and Declaration of Trust of Putnam Funds Trust (the "Trust") is on file with the Secretary of The Commonwealth of Massachusetts, and notice is hereby given that this instrument is executed on behalf of the Trustees of the Trust as Trustees and not individually and that the obligations of or arising out of this instrument are not binding on any of the Trustees, officers or shareholders individually of the Trust, but are binding only upon the trust property of the Trust. Furthermore, notice is given that the trust property of the Purchaser and each other series of Putnam Funds Trust is separate and distinct and that the obligations of or arising out of this Agreement are several and not joint or joint and several and are binding only on the trust property of the Purchaser with respect to its obligations under this Agreement.

Exhibit A

Voting Rights Waiver

See Attached

#### Putnam letterhead

October 21, 2013

Global Eagle Entertainment Inc. 4353 Park Terrace Drive Westlake Village, California 91361 To Whom It May Concern:

Putnam Investment Management, LLC ("PIM") is the investment manager to Putnam Capital Spectrum Fund ("Capital Spectrum") and Putnam Equity Spectrum Fund ("Equity Spectrum"), both series of Putnam Funds Trust, a Massachusetts business trust. Capital Spectrum and Equity Spectrum hold shares of common stock, \$0.0001 par value per share ("Shares"), of Global Eagle Entertainment Inc. ("Global Eagle"). In addition, other funds, entities or accounts sponsored or managed on a discretionary basis by PIM, Putnam Investments Limited or The Putnam Advisory Company, LLC and other officers, directors, and/or employees of PIM, Putnam Investments Limited, or The Putnam Advisory Company, LLC (collectively, "Other Putnam Investors") may own or acquire securities issued by Global Eagle.

In the event that Shares of Global Eagle, together with any other securities issued by Global Eagle that constitute "voting securities" (as defined below) ("Global Eagle Voting Securities"), confer "voting rights" (as defined below) ("Global Eagle Voting Rights"), in each case exercisable by Capital Spectrum, Equity Spectrum, and any Other Putnam Investors, exceeding, in the aggregate, 4.99% of the total voting rights exercisable by outstanding Global Eagle Voting Securities (the portion of the Global Eagle Voting Securities held by Capital Spectrum, Equity Spectrum, and Other Putnam Investors that confer Global Eagle Voting Rights exceeding 4.99% of all voting rights exercisable by outstanding Global Eagle Voting Securities being referred to as the "Excess Voting Securities"), Capital Spectrum, Equity Spectrum, and PIM (on behalf of each Other Putnam Investor) agree to forego and to waive any and all voting rights they may have in respect of the Excess Voting Securities and any additional Global Eagle Voting Securities ("Additional Voting Securities") so that Global Eagle Voting Rights exercisable by Capital Spectrum, Equity Spectrum, and Other Putnam Investors in the aggregate do not confer voting rights exceeding 4.99% of total voting rights exercisable by all outstanding Global Eagle Voting Securities after subtracting from such total voting rights, in the calculation of the total number of Global Eagle Voting Rights deemed to be outstanding, all Global Eagle Voting Rights exercisable by Excess Voting Securities, Additional Voting Securities and any other Global Eagle Voting Securities held by other securityholders of Global Eagle who have waived any or all voting rights. (In respect of PIM's agreement on behalf of Other Putnam Investors, PIM has or will obtain authority from each such other Putnam Investor to agree to the foregoing waiver on behalf of the Other Putnam Investor.) The voting rights so foregone and waived will be apportioned among Capital Spectrum, Equity Spectrum, and Other Putnam Investors on a pro rata basis based upon their relative Share holdings in Global Eagle at the time of the exercise of any vote. For purposes of this agreement, "voting rights" are rights to vote for the election or removal of Global Eagle's directors, or rights deemed to be equivalent to the right to vote for the election or removal of a director, under applicable interpretations of the term "voting security" under the Investment Company Act of 1940, as amended, by the Securities and Exchange Commission or its staff.

A-2-

This agreement to forego and waive voting rights is irrevocable with respect to Capital Spectrum, Equity Spectrum, and any Other Putnam Investor that is an investment company registered under the Investment Company Act of 1940, as amended, and is irrevocable with respect to Other Putnam Investors until such time as Capital Spectrum, Equity Spectrum, and any Other Putnam Investor that is an investment company Act of 1940, as amended, is no longer an owner of any Shares in Global Eagle, at which time this agreement will expire and the remaining Other Putnam Investors will be entitled to any and all voting rights pertaining to their Shares. It is the intention of the undersigned that this letter be interpreted broadly to effect the desire that the Excess Shares and Additional Shares be identical to that of a separate non-voting class.

A copy of the Agreement and Declaration of Trust of each of the Putnam Funds is on file with the Secretary of the Commonwealth of Massachusetts and notice is hereby given that this instrument is executed on behalf of the Trustees of each Putnam Fund as Trustees and not individually and that the obligations of this instrument are not binding on any of the Trustees or officers or shareholders individually, but are binding only on the assets or property of each Putnam Fund with respect to its obligations hereunder. Furthermore, notice is given that the assets and liabilities of each series of each Putnam Fund that is a series company are separate and distinct and that the obligations of or arising out of this instrument are several and not joint and are binding only on the assets or property of each series with respect to its obligations hereunder. In addition, although multiple Putnam Funds may be party hereto, each Putnam Fund is entering into this instrument individually (and not jointly or jointly and severally) and is not liable for any matter relating to any other Putnam Fund.

Very truly yours,

## Putnam Investment Management, LLC

For itself and as investment manager for Putnam Capital Spectrum Fund and Putnam Equity Spectrum Fund, each a series of Putnam Funds Trust

By: \_\_\_\_\_\_\_ Brian Lenhardt, Authorized Person

Accepted and Agreed to:

**Global Eagle Entertainment Inc.** 

By: [Name]

A-3-

### AMENDMENT NO. 1 TO THE AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This AMENDMENT NO. 1 TO THE AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (the "*Amendment*") is entered into on October 21, 2013, by and between Global Eagle Entertainment Inc., a Delaware corporation (the "*Company*"), and the parties named on the signature pages hereto (collectively referred to hereinafter as the "*Holders*").

### WITNESSETH:

WHEREAS, the Company and the Holders, among others, entered into an Amended and Restated Registration Rights Agreement dated as of January 31, 2013 (the "*Agreement*"); and

WHEREAS, the Company and the Holders desire to amend the Agreement as provided in this Amendment.

#### AGREEMENT

NOW, THEREFORE, in consideration of the foregoing, the mutual covenants, promises and agreements hereinafter set forth, the mutual benefits to be gained by the performance thereof, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged and accepted, the parties to this Amendment, intending to be legally bound, hereby agree as follows:

Section 1. <u>Defined Terms</u>. Capitalized terms used herein, unless otherwise defined herein, have the meanings ascribed to them in the Agreement.

Section 2. <u>New Defined Terms</u>. The parties hereby agree to amend Section 1.1 to the Agreement to add the following new defined terms, in alphabetical order, as follows:

"Convertible Note" means the convertible promissory note issued to PAR as of October 21, 2013.

"Existing Resale Registration Statement" has the meaning specified in Section 1.2(a).

"New Resale Registration Statement" has the meaning specified in Section 1.2(a).

"Original Resale Registration Statement" has the meaning specified in Section 1.2(a).

"PAR Secondary Registrable Securities" means those shares of Non-Voting Common Stock issued pursuant to the conversion of the Convertible Note.

"Post-Effective Amendment" has the meaning specified in Section 1.2(a).

Section 3. <u>Amendment to Definition of PAR Registrable Securities</u>. The definition of "PAR Registrable Securities" in Section 1.1 is deleted in its entirety and replaced with the following text:

"PAR Registrable Securities" means the Registrable Securities issued to PAR and its affiliates pursuant to the Merger Agreement, the Stock Purchase Agreement, the Equity Backstop Agreement and the shares of Common Stock issuable upon conversion of the shares of Non-Voting Common Stock issuable upon conversion of the <u>Convertible Note</u>.

Section 4. <u>Amendment to Definition of Registrable Securities</u>. The definition of "Registrable Securities" in Section 1.1 is deleted in its entirety and replaced with the following text:

"Registrable Securities" means (i) all shares of Common Stock of the Company held by any Holder, (ii) all shares of Common Stock of the Company issued or issuable upon the conversion, exercise or exchange of any other equity security of the Company held by any Holder, including without limitation, (A) any warrants to purchase shares of Common Stock and any shares of Non-Voting Common Stock of the Company, including the PAR Secondary Registrable Securities and (B) any warrants originally exercisable for shares of capital stock of Row 44, Inc., which warrants, pursuant to the terms of the Merger Agreement, are now, or may be, exercisable for shares Common Stock, (iii) all shares of Common Stock issued to any Holder in connection with any stock split, stock dividend, recapitalization or similar event (including without limitation, any shares of Common Stock issued or issuable upon conversion of any shares of Non-Voting Common Stock issued to any Holder in connection with any stock split, stock dividend, recapitalization or similar event), and (iv) any shares of Common Stock issued or issuable upon the exercise of any equity security of the Company that is issuable upon conversion of any working capital loans in an amount up to \$500,000 made to the Company by any Holder; provided, however, that as to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (a) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (b) such securities shall have been otherwise transferred, new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (c) such securities shall have ceased to be outstanding; (d) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction; or (e) such securities (other than with respect to the Founder Registrable Securities, the PAR Registrable Securities or the Registrable Securities issued to Putnam Capital Spectrum Fund and Putnam Equity Spectrum Fund and their affiliates) shall be eligible for sale pursuant to Rule 144 (or any similar rule or regulation then in force).

Section 5. <u>Amendment to Section 1.2(a)</u>. Section 1.2(a) is deleted in its entirety and replaced with the following text:

"Registration Statement Covering Resale of Registrable Securities. The Company shall prepare and file or cause to be prepared and filed with the SEC, no later than seven (7) Business Days after January 31, 2013, a Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415 of the Securities Act registering the resale from time to time by Holders of all of the Registrable Securities held by the Holders (the "Original Resale Registration Statement") and, in the event that the Convertible Note converts into shares of Non-Voting Common Stock, the Company shall prepare and file either (a) a new Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415 of the Securities Act registering the resale from time to time by Holders of all of the Registrable Securities issuable upon conversion of such shares of Non-Voting Common Stock held by the Holders (a "New Resale Registration Statement") or (b) a post-effective amendment (a "Post-Effective Amendment") to that certain registration statement on Form S-3 of the Company (SEC File No. 333-188121) (the "Existing Resale Registration Statement")(collectively with the Original Registration Statement, the "Resale Shelf Registration Statement"). The Resale Shelf Registration Statement shall be on Form S-3 or another appropriate form permitting registration of such Registrable Securities for resale by such Holders. The Company shall use reasonable best efforts to cause the Original Resale Registration Statement to be declared effective under the Securities Act on or prior to February 14, 2013, provided that if the Original Resale Registration Statement is not declared effective on or prior to February 14, 2013, then the Company shall use its reasonable best efforts to cause the Original Resale Registration Statement to be declared effective as soon as possible thereafter and in any event within seventyfive (75) days after the date of January 31, 2013, and to keep the Original Resale Shelf Registration Statement effective under the Securities Act at all times until the expiration of the Effectiveness Period. The Company shall prepare and file with the SEC any New Resale Registration Statement or Post-Effective Amendment no later than thirty (30) days after the Maturity Date (as defined in the Convertible Note) (the "Filing Date"). The Company shall use reasonable best efforts to cause any New Resale Registration Statement or Post-Effective Amendment, as the case may be, to be declared effective under the Securities Act within thirty (30) days following the Filing Date, provided that if such New Resale Registration Statement or Post-Effective Amendment, as the case may be, is not declared effective on or prior to such thirtieth day, then the Company shall use reasonable best efforts to cause such New Resale Registration Statement or Post-Effective Amendment, as the case may be, to be declared effective as soon as possible thereafter and in any event within seventy-five (75) days after the Filing Date, and to keep the New Resale Registration Statement or Post-Effective Amendment, as the case may be, effective under the Securities Act at all times until the expiration of the Effectiveness Period."

Section 6. <u>Amendment to Section 1.2(e)(iii)</u>. Section 1.2(e)(iii) is deleted in its entirety and replaced with the following text:

"If the managing underwriter in an Underwritten Takedown advises the Company and the Requesting Holder that, in its view, the number of shares of Registrable Securities requested to be included in such underwritten offering exceeds the largest number of shares that can be sold without having an adverse effect on such offering, including the price at which such shares can be sold (the "<u>Maximum Offering Size</u>"), the Company shall include in such underwritten offering, up to the Maximum Offering Size, first, as many of the Registrable Securities not including the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities, requested to be included by all Selling Holders (and allocated pro rata among such Selling Holders on the basis of the relative number of Registrable Securities, not including the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities, held by each such Selling Holder at such time (it being understood that for the purposes of calculating the relative number of Registrable Securities held by any Selling Holder, in the event such Selling Holder owns any security of the Company that may be converted, exercised or exchanged into Registrable Securities, the relative number of Registrable Securities held by such Selling Holder shall be determined as if such Selling Holder exercised such equity security on a cashless exercise basis or converted such equity security, exclusive in each such instance of the PAR Secondary Registrable Securities)) as can be included without adversely affecting such distribution by all Selling Holders; and second, as many of the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities that PAR has requested be included as can be included without adversely affecting such distribution."

Section 7 Amendment to Section 1.3. Section 1.3 is deleted in its entirety and replaced with the following text:

"Piggyback Rights. If the Company proposes to register any of its shares of Common Stock (other than any registration for the account of the Company of securities issued pursuant to any employee benefit plan or in any acquisition by the Company), the Company will include in such registration all Registrable Securities held by the Holders requested to be so included; provided, however, that if, in the case of an underwritten offering, the managing underwriter informs the Company that the number of shares held by the Holders requested to be included exceeds the amount which can be sold in such offering without adversely affecting the distribution of the shares being offered, the Company shall include, first, all of the shares the Company has proposed to register; second, as many of the Registrable Securities not including the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities (pro rata based on the number of Registrable Securities, not including the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities, that each Holder has requested be included in such underwritten offering and the aggregate number of Registrable Securities, not including the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities, that the Holders have requested to be included in such underwritten offering), as can be included without adversely affecting such distribution; third, as many of the Registrable Securities issuable upon conversion of the PAR Secondary Registrable Securities that PAR has requested be included in such underwritten offering as can be included without adversely affecting such distribution; and, fourth, any other shares of Common Stock proposed to be included in such offering; provided, however, that in no event may less than one-third of the total number of shares of Common Stock to be included in such underwritten offering be made available for Registrable Securities."

8. Effect of Amendment. Except as explicitly amended by the terms of this Amendment, the terms of the Agreement shall remain in effect and are unchanged by this Amendment.

# [Signature Page Follows]

IN WITNESS WHEREOF, each of the parties hereto has caused this Amendment to be duly executed, all as of the day and year first above written.

### **COMPANY:**

### GLOBAL EAGLE ENTERTAINMENT INC.

By:/s/ Michael PigottName:Michael PigottTitle:Vice-President, Legal Affairs

### **HOLDERS:**

### GLOBAL EAGLE ACQUISITION LLC

 By:
 /s/ James A. Graf

 Name:
 James A. Graf

 Title:
 Vice President, CFO, Treasurer and Secretary

PAR INVESTMENT PARTNERS, L.P.

By: PAR Group, L.P., its general partner

By: PAR Capital Management, Inc., its general partner

By: /s/ Steven M. Smith

Name: Steven M. Smith Title: Chief Operating Officer and General Counsel

## PUTNAM CAPITAL SPECTRUM FUND PUTNAM EQUITY SPECTRUM FUND

By: Putnam Investment Management, LLC

By: <u>/s/ David Glancy</u> Name: David Glancy Title: Portfolio Manager Putnam letterhead

October 21, 2013

Global Eagle Entertainment Inc. 4353 Park Terrace Drive Westlake Village, California 91361

### To Whom It May Concern:

Putnam Investment Management, LLC ("PIM") is the investment manager to Putnam Capital Spectrum Fund ("Capital Spectrum") and Putnam Equity Spectrum Fund ("Equity Spectrum"), both series of Putnam Funds Trust, a Massachusetts business trust. Capital Spectrum and Equity Spectrum hold shares of common stock, \$0.0001 par value per share ("Shares"), of Global Eagle Entertainment Inc. ("Global Eagle"). In addition, other funds, entities or accounts sponsored or managed on a discretionary basis by PIM, Putnam Investments Limited or The Putnam Advisory Company, LLC and other officers, directors, and/or employees of PIM, Putnam Investments Limited, or The Putnam Advisory Company, LLC (collectively, "Other Putnam Investors") may own or acquire securities issued by Global Eagle.

In the event that Shares of Global Eagle, together with any other securities issued by Global Eagle that constitute "voting securities" (as defined below) ("Global Eagle Voting Securities"), confer "voting rights" (as defined below) ("Global Eagle Voting Rights"), in each case exercisable by Capital Spectrum, Equity Spectrum, and any Other Putnam Investors, exceeding, in the aggregate, 4,99% of the total voting rights exercisable by outstanding Global Eagle Voting Securities (the portion of the Global Eagle Voting Securities held by Capital Spectrum, Equity Spectrum, and Other Putnam Investors that confer Global Eagle Voting Rights exceeding 4.99% of all voting rights exercisable by outstanding Global Eagle Voting Securities being referred to as the "Excess Voting Securities"), Capital Spectrum, Equity Spectrum, and PIM (on behalf of each Other Putnam Investor) agree to forego and to waive any and all voting rights they may have in respect of the Excess Voting Securities and any additional Global Eagle Voting Securities ("Additional Voting Securities") so that Global Eagle Voting Rights exercisable by Capital Spectrum, Equity Spectrum, and Other Putnam Investors in the aggregate do not confer voting rights exceeding 4.99% of total voting rights exercisable by all outstanding Global Eagle Voting Securities after subtracting from such total voting rights, in the calculation of the total number of Global Eagle Voting Rights deemed to be outstanding, all Global Eagle Voting Rights exercisable by Excess Voting Securities, Additional Voting Securities and any other Global Eagle Voting Securities held by other securityholders of Global Eagle who have waived any or all voting rights. (In respect of PIM's agreement on behalf of Other Putnam Investors, PIM has or will obtain authority from each such other Putnam Investor to agree to the foregoing waiver on behalf of the Other Putnam Investor.) The voting rights so foregone and waived will be apportioned among Capital Spectrum, Equity Spectrum, and Other Putnam Investors on a pro rata basis based upon their relative Share holdings in Global Eagle at the time of the exercise of any vote. For purposes of this agreement, "voting rights" are rights to vote for the election or removal of Global Eagle's directors, or rights deemed to be equivalent to the right to vote for the election or removal of a director, under applicable interpretations of the term "voting security" under the Investment Company Act of 1940, as amended, by the Securities and Exchange Commission or its staff.

This agreement to forego and waive voting rights is irrevocable with respect to Capital Spectrum, Equity Spectrum, and any Other Putnam Investor that is an investment company registered under the Investment Company Act of 1940, as amended, and is irrevocable with respect to Other Putnam Investors until such time as Capital Spectrum, Equity Spectrum, and any Other Putnam Investor that is an investment company registered under the Investment Company Act of 1940, as amended, is no longer an owner of any Shares in Global Eagle, at which time this agreement will expire and the remaining Other Putnam Investors will be entitled to any and all voting rights pertaining to their Shares. It is the intention of the undersigned that this letter be interpreted broadly to effect the desire that the Excess Shares and Additional Shares be identical to that of a separate non-voting class.

A copy of the Agreement and Declaration of Trust of each of the Putnam Funds is on file with the Secretary of the Commonwealth of Massachusetts and notice is hereby given that this instrument is executed on behalf of the Trustees of each Putnam Fund as Trustees and not individually and that the obligations of this instrument are not binding on any of the Trustees or officers or shareholders individually, but are binding only on the assets or property of each Putnam Fund with respect to its obligations hereunder. Furthermore, notice is given that the assets and liabilities of each series of each Putnam Fund that is a series company are separate and distinct and that the obligations of or arising out of this instrument are several and not joint and are binding only on the assets or property of each series with respect to its obligations hereunder. In addition, although multiple Putnam Funds may be party hereto, each Putnam Fund is entering into this instrument individually (and not jointly or jointly and severally) and is not liable for any matter relating to any other Putnam Fund.

Very truly yours,

# Putnam Investment Management, LLC

For itself and as investment manager for Putnam Capital Spectrum Fund and Putnam Equity Spectrum Fund, each a series of Putnam Funds Trust

By: /s/ Brian Lendhardt Brian Lenhardt, Authorized Person

-3-

Accepted and Agreed to:

# Global Eagle Entertainment Inc.

By: <u>/s/ Michael Pigott</u> Michael Pigott

# **Global Eagle Entertainment Acquires IFE Services Limited**

# Transaction Solidifies Position as In-Flight Media Leader

# Adds to Coverage of Key Emerging Markets

Los Angeles, CA – October 21, 2013 – Global Eagle Entertainment Inc. (NASDAQ: ENT) announced today that it has acquired Travel Entertainment Group Equity Limited, the UK-based parent company of IFE Services Limited (IFE Services) from GCP Capital Partners LLP for approximately \$36 million in cash. IFE Services is a leading provider of in-flight entertainment services to airlines and cruise lines worldwide. The acquisition expands Global Eagle's leadership in delivering content and software solutions to the international travel industry.

Global Eagle's purchase of IFE Services follows closely on the Company's acquisition of digital content leader Post Modern Group (PMG) in July 2013. With a 20-year history, IFE Services provides a broad range of content solutions, spanning movies, TV programs, games, mobile apps, publications, safety videos and technical support to a worldwide client base of over 50 airline and cruise ship operators. IFE Services has a particularly strong customer presence among quickly-growing airlines in developing markets in Africa, Asia and South America.

"This acquisition is in line with our strategic objective to grow our position as the leading provider of in-flight media, to broaden our client base and to further strengthen the combined service offerings we provide to airlines worldwide," said John LaValle, Chief Executive Officer of Global Eagle. "The addition of IFE Services will significantly enhance our presence in multiple fast-growing, emerging markets where our combined service offerings are in demand. We look forward to working together with IFE's CEO Andy McEwan and the talented team at IFE Services in bringing our innovative solutions to the worldwide travel industry."

"This is another transaction that we expect to be highly complementary to our existing business," said Dave Davis, Chief Financial Officer of Global Eagle. "In addition to adding IFE Service's strong cash flows, we believe we can unlock substantial efficiencies from our combined operations to the benefit of our shareholders. At the same time, we continue to be in a solid position to pursue additional acquisitions given our healthy balance sheet, strong management team and growing global footprint."

For the full year 2013 and on a stand-alone basis, IFE Services is expected to generate approximately \$37 million to \$40 million of annual revenue\* and approximately \$7 million to \$9 million of Adjusted EBITDA.\* In addition, we expect to achieve substantial synergies beginning in first half of 2014 as a result of the combination of IFE Services with Global Eagle.

"Having supported the business and the management team of IFE over the last five years, we are delighted that, in Global Eagle, IFE Services has a supportive new owner that will be able to capitalize on its achievements to date," said Adam Maidment, Managing Director of GCP Capital Partners LLP. "We would like to thank the CEO, Andy McEwan, and his team for their excellent work over the last few years and we wish them well in this next phase of the business' development."

In connection with the consummation of the IFE transaction, Global Eagle agreed to sell to one of its existing institutional stockholders, 2,435,472 shares of Global Eagle's common stock for an aggregate purchase price of \$21,000,000, in a registered direct offering by means of a prospectus supplement to Global Eagle's effective shelf registration statement which Global Eagle will file with the Securities and Exchange Commission ("SEC"). In addition, Global Eagle will issue to PAR Investment Partners, L.P., an existing Global Eagle stockholder, a \$19,000,000 promissory note which is convertible into shares of Global Eagle's non-voting common stock. The terms of these sales of securities will be contained in a Current Report on Form 8-K to be filed by Global Eagle with the SEC.

### About Global Eagle

Global Eagle Entertainment Inc. is the leading full service platform offering both content and connectivity for the worldwide airline industry. Through its combined content, distribution and technology platforms, Global Eagle provides airlines and the millions of travelers they serve with the industry's most complete offering of in-flight video content, e-commerce and information services. Through its Row 44 subsidiary, Global Eagle utilizes Ku-band satellite technology to provide airline passengers with Internet access, live television, shopping and travel-related information. Currently installed on more than 500 aircraft, Row 44 has the largest fleet of connected entertainment platforms operating over land and sea globally. In addition, through its AIA subsidiary, Global Eagle provides film and television content, games and applications to more than 130 airlines worldwide. Global Eagle is headquartered in greater Los Angeles, California and maintains offices and support personnel around the world. Find out more at www.globaleagleent.com.

### **About IFE Services**

IFE Services is a leading provider of in-flight entertainment solutions to the airline industry. Its focus on quality and innovation is supported by a strong commitment to customer service and investment in the latest systems and technologies.

IFE Services supplies a full range of services to enable its clients to provide a first class entertainment experience to passengers. They include movies, TV programs, audio, games, 3D map, apps, safety and destination films, portable entertainment systems, onboard publications and AVOD technical support and management.

IFE Services works with a broad client-base worldwide of over 50 airlines and cruise ship operators. The company's headquarters are in the UK with regional offices located in Madrid, Singapore, Santiago, Sao Paulo, Nairobi, Kuala Lumpur, Oberhausen, Johannesburg and Irvine, California.

## About GCP Capital Partners LLP

GCP Capital Partners LLP ("GCP") is a leading UK mid-market private equity business. It was established in 2007 as Greenhill Capital Partners Europe, the European private equity arm of Greenhill & Co. Inc.

## **Notices Regarding Sales of Securities**

This press release shall not constitute an offer to sell or the solicitation of an offer to buy the securities described herein, nor shall there be any offer or sale of these securities in any state or jurisdiction in which such offer, sale or solicitation would be unlawful. The registered direct offering is being made only by means of the prospectus supplement and accompanying base prospectus. When available, copies of the prospectus supplement and accompanying base prospectus. When available, copies of the prospectus supplement and accompanying base prospectus relating to the registered direct offering may be obtained at the SEC's website at http://www.sec.gov or directly from the company by contacting Global Eagle Entertainment Inc., Attention: Corporate Secretary, 4353 Park Terrace Drive, Westlake Village, CA 91361 (818) 706-3111. The issuance of the convertible promissory note and the underlying shares will not be registered under the Securities Act of 1933, as amended, and may not be offered or sold in the United States absent registration or an applicable exemption from registration requirements.

### \* Financial Information of IFE

IFE has historically prepared its financial statements in accordance with U.K. Generally Accepted Accounting Practice ("U.K. GAAP"). Accordingly, projected revenue and certain components of adjusted EBITDA of IFE for full year 2013 on a stand-alone basis have been prepared in accordance with U.K. GAAP. Certain differences exist between U.K. GAAP and generally accepted accounting principles in the United States of America which might be material to these projections. With respect to 2013 projected adjusted EBITDA of IFE Services, we define adjusted EBITDA as net earnings before interest expense, income taxes, depreciation and amortization, plus exceptional costs incurred at the company that will no longer be incurred after the transaction has closed. Adjusted EBITDA is a non-GAAP financial measure commonly used by financial analysts, investors, rating agencies and other interested parties in evaluating companies. However, adjusted EBITDA has limitations as an analytical tool and should not be considered in isolation from, or as an alternative to, operating income, cash flow, EPS or other income or cash flow data prepared in accordance with GAAP. A quantitative reconciliation is not available without unreasonable efforts.

### **Cautionary Note Concerning Forward-Looking Statements**

We make forward-looking statements in this release within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements relate to expectations or forecasts for future events, including without limitation our earnings, revenues, expenses or other future financial or business performance or strategies, or the impact of legal or regulatory matters on our business, results of operations or financial condition. These statements may be preceded by, followed by or include the words "may," "might," "will," "will likely result," "should," "estimate," "plan," "project," "forecast," "intend," "expect," "anticipate," "believe," "seek," "continue," "target" or similar expressions. These forward-looking statements are based on information available to us as of the date of this earnings release, and involve substantial risks and uncertainties. Accordingly, forward-looking statements should not be relied upon as representing our views as of any subsequent date, and we do not undertake any obligation to update forward-looking statements as a result of as a result of new information, future events or developments or otherwise.

Among the factors that could cause actual results to differ materially from past results and future plans and projected future results are the following: our ability to integrate the Row 44, AIA, PMG and IFE businesses, the ability of the combined business to grow, including through acquisitions which we are able to successfully integrate, and the ability of our executive officers to manage growth profitably; the possibility that any of the anticipated benefits of our acquisitions, including synergies, cost savings and positive financial impact, will not be realized as anticipated or at all; the outcome of any legal proceedings pending or that may be instituted against us or our subsidiaries; changes in laws or regulations that apply to us or our industry; our ability to recognize and timely implement future technologies in the satellite connectivity space, including Ka-band system development and deployment; our ability to deliver end-to-end network performance sufficient to meet increasing airline customer and passenger demand; our ability to obtain and maintain international authorizations to operate our service over the airspace of foreign jurisdictions our customers utilize; our ability to expand our service offerings and deliver on our service roadmap; our ability to timely and cost-effectively identify and license television and media content that passengers will purchase; general economic and technological circumstances in the satellite transponder market, including access to transponder space in capacity limited regions and successful launch of replacement transponder capacity where applicable; our ability to obtain and maintain licenses for content used on legacy installed in-flight entertainment systems; the loss of, or failure to realize benefits from, agreements with our airline partners; the loss of relationships with original equipment manufacturers or dealers; unfavorable economic conditions in the airline industry and economy as a whole; our ability to expand our domestic or international operations, including our ability to grow our business with current and potential future airline partners or successfully partner with satellite service providers, including Hughes Network Systems; our reliance on third-party satellite service providers and equipment and other suppliers, including single source providers and suppliers; the effects of service interruptions or delays, technology failures, material defects or errors in our software, damage to our equipment or geopolitical restrictions; the limited operating history of our connectivity and in-flight television and media products; costs associated with defending pending or future intellectual property infringement actions and other litigation or claims; increases in our projected capital expenditures due to, among other things, unexpected costs incurred in connection with the roll out of our technology roadmap or our international plan of expansion; fluctuation in our operating results; the demand for in-flight broadband internet access services and market acceptance for our products and services; and other risks and uncertainties set forth in this release and in our most recent Annual Report on Form 10-K and any subsequently filed Quarterly Reports on Form 10-Q.

#### **Global Eagle Entertainment Media Contact:** Karin Pellmann

kpellmann@row44.com 646-515-6933

#### **Global Eagle Entertainment Investor Contacts:** Chris Plunkett or Brad Edwards

Brainerd Communicators, Inc. (212) 986-6667 plunkett@braincomm.com edwards@braincomm.com